Technical Report TR-04-18

# SR-Can

# Data and uncertainty assessment

# Migration parameters for the bentonite buffer in the KBS-3 concept

Michael Ochs and Caterina Talerico BMG Engineering Ltd

August 2004

#### Svensk Kärnbränslehantering AB

Swedish Nuclear Fuel and Waste Management Co Box 5864 SE-102 40 Stockholm Sweden Tel 08-459 84 00 +46 8 459 84 00 Fax 08-661 57 19 +46 8 661 57 19



# SR-Can

# Data and uncertainty assessment

# Migration parameters for the bentonite buffer in the KBS-3 concept

Michael Ochs and Caterina Talerico BMG Engineering Ltd

August 2004

This report concerns a study which was conducted for SKB. The conclusions and viewpoints presented in the report are those of the authors and do not necessarily coincide with those of the client.

A pdf version of this document can be downloaded from www.skb.se

# Abstract

SKB is currently preparing license applications related to the deep repository for spent nuclear fuel and an encapsulation plant. The present report is one of several specific data reports feeding into the interim reporting for the latter application; it is concerned with the derivation and recommendation of radionuclide migration input parameters for a MX-80 bentonite buffer to PA models. Recommended values for the following parameters as well as the associated uncertainties are derived and documented for a total of 38 elements and oxidation states:

- diffusion-available porosity (ε),
- effective diffusivity (D<sub>e</sub>),
- distribution coefficient (K<sub>d</sub>).

Because of the conditional nature of these parameters, particularly of  $K_d$ , they were derived specifically for the conditions expected to be relevant for PA consequence calculations.  $K_d$ values were generally evaluated for the specific porewater composition and solid/water ratio representative for MX-80 compacted to 1,590 kg/m<sup>3</sup>. Because of the highly conditional nature of  $K_d$ , this was done for several porewater compositions which reflect possible variations in geochemical boundary conditions.  $D_e$  and  $\epsilon$  were derived as a function of density. Parameter derivation was based on systematic datasets available in the literature and/or on thermodynamic models. Associated uncertainties were assessed for a given set of PA conditions and as a function of variability in these conditions.

In a final step, apparent diffusivity  $(D_a)$  values were calculated from the recommended parameters and compared with independent experimental measurements to arrive at self-consistent sets of migration parameters.

# Sammanfattning

SKB förbereder för tillfället en ansökan för en inkapslingsanläggning för använt kärnbränsle. Den här rapporten är en i mängden av de rapporter som levererar specifika data för säkerhetsredovisningen som ingår i ansökan. Den behandlar framtagning av och urval av transportparametrar för radionuklider i en buffert av MX-80 bentonit. Rekommenderade värden för följande parametrar tillsammans med tillhörande osäkerheter har tagits fram och dokumenterats för 38 grundämnen och oxidationstal:

- porositet (tillgänglig för diffusion) (ε),
- effektiv diffusivitet (D<sub>e</sub>),
- distributionskoefficienter (K<sub>d</sub>).

På grund av det sätt som dessa parametrar styrs av omgivningsfaktorer, särskilt  $K_d$ , har de bestämts specifikt för de förhållanden som förväntas vara relevanta för konsekvensberäkningar i säkerhetsredovisningen.  $K_d$  värden har normalt bestämts från givna porvattensammansättningar och fast fas/vätska förhållanden vilka är representativa för MX-80 kompakterad till 1 590 kg/m<sup>3</sup>. För  $K_d$  har detta gjorts för ett flertal porvattensammansättningar för att reflektera variationer is de geokemiska randvillkoren.  $D_e$  och  $\varepsilon$  har tagits fram som en funktion av densitet. Parametervalet har baserats på systematiska publicerade datasammanställningar och/eller på termodynamiska modeller. Osäkerheterna i parametrarna har bedömts för givna förhållanden/randvillor från säkerhetsredovisningen och presenteras som en funktion av variabiliteten i dessa förhållanden.

Slutligen har den apparenta diffusiviteten  $(D_a)$  beräknats med de framtagna parametrarna och jämförts med oberoende experimentella resultat för att testa att parametrarna är internt konsistenta.

# **Extended Summary and road map**

#### Introduction

SKB, the Swedish Nuclear Fuel and Waste Management Co, is currently preparing license applications to locate, build and operate i) the deep repository for spent nuclear fuel and ii) an encapsulation plant in which the spent fuel will be emplaced in canisters to be deposited in the deep repository.

Two safety reports will be produced within the next five years; one for the application to build an encapsulation plant, SR-Can, and one for the application to build the repository, SR-Site. The present report is one of several specific data reports feeding into the SR-Can Interim Report. The SR-Can Interim Report will be produced after an initial phase of the SR-Can project, with the main purpose of demonstrating the adopted methodology. Specific data reports are being prepared in cases where uncertainties of input data to safety assessment (consequence) calculations will have a decisive influence on safety related output uncertainty. The parameter and uncertainty estimates given in the specific data reports will also be compiled and further assessed in a SR-Can Data Report.

#### Content and organisation of the present report

The present specific data report is concerned with the derivation and recommendation of the radionuclide migration parameters for a MX-80 bentonite buffer with a dry density of 1,590 kg/m<sup>3</sup>. Recommended values for the following parameters as well as the associated uncertainties are derived and documented for each relevant element:

- diffusion-available porosity (ε),
- effective diffusivity (D<sub>e</sub>),
- distribution coefficient (K<sub>d</sub>).

The report follows the general outline developed by SKB for the SR-Can interim reporting and is organised as follows:

- The derivation of recommended data and uncertainties is presented in the central chapter of the report, chapter 5, and are summarised in chapter 6.
- General discussions on data sources used and treatment of uncertainties are given in chapters 3 and 4.
- Detailed data derivation tables as well as background information on system characteristics and models used are given in the appendix.
- Chapters 1 and 2 provide a general introduction and explain the purpose of the present report in SR-Can.

The purpose of the present report is the derivation of input data for PA models. Accordingly, brief reviews of the literature are provided to the degree deemed necessary for supporting data selection, but no general review of the literature is provided. Similarly, arguments are provided why certain pieces of the literature were selected as source data base; but reasons for rejection are not discussed for each study that was considered unreliable or otherwise too limited.

A brief discussion on general aspects of sorption and diffusion processes are given in the SR-Can Process Report /SKB, 2004/, including brief discussions of conceptual uncertainties. It is recommended to consult the pertinent sections of the process report.

## Approach to the derivation of data and uncertainties

#### General framework and scope

The present migration parameters, in particular  $K_d$ , are conditional in nature. Therefore, they were derived specifically for the conditions expected to be relevant for PA calculations. The underlying premise for the direct use of conditional parameters in consequence calculations is that conditions in parameter derivation and application are consistent. Significant discrepancies would require the re-derivation of the migration parameters.

In principle, this situation could be resolved by directly including sorption/diffusion processes in the model used for consequence calculations, which would require a THMC model approach. However, in addition to the great difficulties still associated with fully coupling THM processes on one and C processes on the other hand, the present state of the art of quantifying radionuclide sorption and diffusion in compacted bentonite does not render this a realistic option to date. For example, reliable and sufficiently constrained thermodynamic sorption models are available for very few relevant radionuclides only, the limiting factor being the lack of sorption data as function of conditions that is needed to parameterise such models. Further, the application of a thermodynamic sorption model requires a reliable standard thermodynamic database, and this prerequisite is not given for a number of radionuclides.

#### General approach for deriving migration parameters

To arrive at consistent sets of migration parameters,  $D_e$ ,  $\varepsilon$  and  $K_d$  were first derived independently. In a second step, apparent diffusivity ( $D_a$ ) values were calculated from these parameters and compared with independent experimental data. Specifically:

- Diffusion-available porosity was viewed as a physical parameter for a given bentonite and density, except for anions.
  - For anions, this parameter was evaluated together with D<sub>e</sub> to ensure a consistent treatment of anion exclusion (see below).
  - For neutral and cationic species, ε was calculated as a function of dry and specific density of the bentonite
- Effective diffusivities were derived for non-sorbing anions, Cs, and HTO:
  - D<sub>e</sub> for non-sorbing anions was derived on the basis of i) experimental data recorded as a function of density for several bentonite-water systems and ii) electrical double layer calculations. To take anion exclusion into account, a reduction factor for  $\epsilon$  was derived from independent studies. The plausibility of the selected D<sub>e</sub> and  $\epsilon$  values was verified by comparisons with independent experimental D<sub>a</sub> values for a range of dry densities.
  - For Cs, a higher value for  $D_e$  was selected than in case of all other elements, in combination with the physical porosity, indicating enhanced cation diffusion. Evaluations based on experimental data recorded as a function of density as well as on electrical double layer calculations confirmed this effect. The plausibility of the selected  $D_e$  and  $\varepsilon$  values was again verified by comparisons with independent experimental  $D_a$  values for a range of dry densities. As Cs sorbs on bentonite, this comparison had to include K<sub>d</sub> (see below) as well.

- Because of the lack of element-specific data for most radionuclides, D<sub>e</sub> values for all other elements were represented by D<sub>e</sub> for HTO and the physical porosity. As most radionuclides exist as a mixture of positively, negatively and uncharged species under the relevant conditions, HTO was considered to represent average diffusion properties adequately. D<sub>e</sub> for HTO was derived as a function of density from experimental data. D<sub>a</sub> values were calculated from these data and the recommended K<sub>d</sub> values (see below) to evaluate the self-consistency of all selected parameters.
- Distribution coefficients (K<sub>d</sub>) were generally evaluated for the specific porewater composition and solid/water ratio representative for MX-80 compacted to 1,590 kg/m<sup>3</sup>. Because of the highly conditional nature of K<sub>d</sub>, this was done for several porewater compositions which reflect possible variations in geochemical boundary conditions (groundwater and bentonite composition, pCO<sub>2</sub>). Wherever possible, derivation of K<sub>d</sub> was based on i) systematic datasets recorded as a function of pH and other conditions and/or ii) thermodynamic sorption models:
  - In the first case, extrapolation from experimental conditions to the conditions relevant for compacted MX-80 was done with the help of conversion factors taking into account differences in surface and radionuclide solution speciation as well as in sorption site density if a substrate other than MX-80 had been used. While the latter is straightforward, the conversion factors related to surface and solution speciation had to be evaluated carefully based on speciation calculations for both experimental and compacted systems.
  - In contrast, thermodynamic sorption models could be applied directly to the conditions relevant for the compacted system to yield K<sub>d</sub>.
  - For most radionuclides, data derivation was based on experimental results and conversion procedures, due to the lack of reliable thermodynamic sorption models. It has to be further realised that many of the published sorption models had been calibrated in the absence of relevant chemical factors, such as carbonate or sulphate, and are therefore not well constrained when applied to a complex system. In some cases, the sorption data presently available in the literature would possibly allow to develop appropriate sorption models. However, such model development was not within the scope of the present project.

#### Sources and treatment of uncertainties

Uncertainties in input parameters can be related to two fundamentally different sources (see Figure 4-1 in section 4.1.2):

- 1. Uncertainties of the input parameters themselves: These include experimental uncertainties and uncertainties introduced by extrapolating conditional parameters (in particular K<sub>d</sub>) from experimental to PA conditions. This type of uncertainty was addressed by explicitly deriving uncertainty limits for each parameter and radionuclide.
- 2. Uncertainties caused by expected variability in the PA conditions. Because of the conditional nature of the present input parameters, it is obvious that any significant deviation from the expected conditions will make the parameters les reliable. Thus, any potential uncertainty in conditions needs to be addressed as well in an overall evaluation of parameter uncertainty. This was addressed by specifically deriving input parameters for the reference boundary conditions as well as several other sets of boundary conditions to trace out the resulting variations in the input parameters.

#### Uncertainties in input parameters

Diffusion parameters (effective diffusivity  $D_e$ , and diffusion-available porosity  $\varepsilon$ ) were evaluated directly as a function of bentonite density and composition, by analysing three different bentonites (including MX-80) with smectite contents ranging from 50% to > 95%:

- Physical porosity was calculated as a function of density. Based on the excellent agreement with measured data, no uncertainty was attributed to this value. In case of diffusion-available porosity for anions, the uncertainty for the reduction factor for  $\varepsilon$  followed directly from the experimental uncertainties reported in the original literature (section 5.3.4).
- A D<sub>e</sub> value for HTO was recommended based on a regression of experimental data vs bentonite density. The standard deviation of the regression prediction at the reference density was taken as the uncertainty in D<sub>e</sub>.
- In case of Cs, the uncertainty of  $D_e$  was defined by the interval between the highest value selected as reasonable for representing enhanced cation mobility and  $D_e$  for HTO.
- In case of D<sub>e</sub> for anions, upper and lower limit were evaluated separately based on the range of experimental observations and electrical double layer calculations.

In the case of  $K_d$  values, uncertainties were defined in analogy to the approach used for data derivation:

- Where extrapolation from experimental conditions to the conditions relevant for compacted MX-80 was done with the help of conversion factors, uncertainties were quantified as follows:
  - An average experimental uncertainty was assumed, based on several representative data sets.
  - For each of the conversion factors used to extrapolate to PA conditions, a corresponding uncertainty factor was defined (see section 5.1.4). These factors were evaluated for each extrapolation procedure.
- The use of thermodynamic sorption models was considered to be associated with the same uncertainty in all cases.

#### Uncertainties caused by expected variability in the PA conditions

The influence of groundwater and porewater composition on diffusion parameters was not generally included (in contrast to  $K_d$ , see below), for the following reasons:

- Diffusion-available porosity can be viewed as a physical parameter for a given bentonite and density, except for anions.
- D<sub>e</sub> refers to steady-state conditions where effects of porewater composition on sorption are not relevant. Effects of porewater composition on the actual diffusion process were implicitly included by pooling data sources referring to a fairly wide range of conditions.

Therefore, density and composition of bentonite are the main factors where variations in the PA conditions could cause significant uncertainties. Since all diffusion parameters were evaluated as a function of density, using data from different bentonites, possible variations are already taken into account.

On the other hand,  $K_d$  values were derived independently for several bentonite/porewater systems which reflected possible variations in groundwater and bentonite composition.

While such sets of  $K_d$  values for different conditions can be used to approximately trace out the effect of the considered variations on  $K_d$ , it has to be kept in mind that the  $K_d$  values are strictly valid only for the specific conditions they had been derived for. The systems considered for data derivation are described in section 5.2.

It is further pointed out that it is not possible to relate variations in groundwater composition directly to uncertainties in  $K_d$ ; rather, this needs to be assessed through specifically calculated porewater compositions resulting from bentonite-groundwater interaction.

Uncertainties in redox conditions with respect to the sorption and diffusion of redox-sensitive elements were addressed by providing parameters separately for each relevant oxidation state. Where redox conditions for PA cannot be constrained sufficiently, overall parameter uncertainty will encompass the entire range of values derived for the relevant oxidation states.

# Contents

1	Introduction	13
1.1	Background	13
	1.1.1 Planned safety assessments	13
	1.1.2 Assessing input data – need for traceable expert decision	13
1.2	Scope and Objectives	14
	1.2.1 Specific objectives	14
1.3	Procedures	14
	1.3.1 Predefined instructions	14
	1.3.2 Experts involved etc	15
2	Use in SR-Can	17
2.1	Applicable SR-Can conditions	17
2.2	Models	18
	2.2.1 Instructions – issues to be addressed	18
	2.2.2 Models used for buffer migration	18
	2.2.3 Differences in models for different conditions	18
2.3	Results of sensitivity analysis – if any?	19
	2.3.1 Instructions – issues to be addressed	19
	2.3.2 Sensitivity of buffer migration parameters	19
3	Sources of information	21
3.1	Sources of data	21
	3.1.1 Instructions – issues to be addressed	21
	3.1.2 Sources of data for buffer migration parameters	21
3.2	Use of models for data derivation	25
	3.2.1 Instructions – issues to be addressed	25
	3.2.2 Modelling used to support the data	25
4	Uncertainty assessment	27
4.1	Overview	27
	4.1.1 Instructions – issues to be addressed	27
	4.1.2 Overall uncertainties (experts view)	27
4.2	Uncertainty related to different conditions	28
	4.2.1 Instructions – issues to be addressed	28
	4.2.2 Conditions affecting buffer migration parameters	29
4.3	Conceptual uncertainties	30
	4.3.1 Instructions – issues to be addressed	30
	4.3.2 Conceptual uncertainties in buffer migration parameters	30
4.4	Data uncertainty, spatial and temporal variation	31
	4.4.1 Instructions – issues to be addressed	31
	4.4.2 Spatial, temporal and site-specific variation	31
4.5	Correlations among migration parameters	32
	4.5.1 Instructions – issues to be addressed	32
	4.5.2 Correlations among buffer migration data	32

5	Quant	ification of data and uncertainties	35
5.1	Metho	ds and approaches	35
	5.1.1	Instructions – issues to be addressed	35
	5.1.2	General approach	35
	5.1.3	Definition of conditions	36
	5.1.4	Derivation of K <sub>d</sub> values	37
5.2	Condit	ions for which parameter values are to be supplied	42
	5.2.1	Instructions – issues to be addressed	42
	5.2.2	Buffer migration parameters	43
	5.2.3	Groundwater	43
	5.2.4	Bentonite density and composition	43
	5.2.5	Porewater compositions	44
	5.2.6	Treatment of variable conditions in the quantification of data	
	D	and uncertainties	45
5.3	Data ai	nd uncertainty estimates	46
	5.3.1	Instructions – issues to be addressed	46
	5.3.2	Buffer migration parameters	47
	5.5.5	Physical porosity	4/
	5.3.4	Effective diffusivity and diffusion-available porosity	10
	5 2 5	lor anions	48
5 1	J.J.J Data C	Derivation of $\mathbf{K}_{d}$ values, comparison with diffusion data	54 07
3.4	Data S	uninary	87
6	Conclu	iding remarks	95
7	Refere	nces	97
Appe	endix A	Groundwater compositions	105
Appe	endix B	Bentonite characteristics	107
Appe	endix C	Porewater compositions	109
Appe	endix D	Thermodynamic sorption models and calculational tools	115
Appe	endix E	Data derivation tables for K <sub>d</sub> values	119
Appe	endix F	Radionuclide speciation tables	145

# 1 Introduction

This document concerns data and uncertainty assessment of buffer migration data in support of the SKB Safety Assessment SR-Can. The report follows a given outline, provided by SKB, where the authors are instructed to answer various questions as set out by this outline.

# 1.1 Background

SKB, the Swedish Nuclear Fuel and Waste Management Co, is currently preparing license applications to locate, build and operate i) the deep repository for spent nuclear fuel and ii) an encapsulation plant in which the spent fuel will be emplaced in canisters to be deposited in the deep repository. Several investigations are conducted in support of these license applications.

## 1.1.1 Planned safety assessments

SKB is currently pursuing site investigations for a deep repository in the municipalities of Östhammar and Oskarshamn. The investigations are conducted in two stages, an initial phase followed, if the expected site suitability is confirmed, by a complete site investigation phase. The aim is to build a deep repository at one of these candidate sites, provided that the bedrock and other relevant conditions are found suitable.

Two safety reports will be produced within the next five years; one for the application to build an encapsulation plant, SR-Can, and one for the application to build the repository, SR-Site. SR-Can will be based on site data from the initial site investigation phase and SR-Site on data from the complete site investigation. After an initial phase of the SR-Can project, an SR-Can Interim report will be produced, with the main purpose of demonstrating the adopted methodology, so that this can be reviewed before it is used for the applications. Also, preliminary safety evaluations /SKB, 2002/, of each site will be made as sub-tasks within the SR-Can project.

## 1.1.2 Assessing input data – need for traceable expert decision

All input data to quantitative aspects of the safety assessment will exhibit uncertainties. The quality of the results of any calculation in the assessment will, among other factors, depend on the quality of the input data and on the rigor with which input data uncertainties have been managed. A common and methodological philosophy for the determination of input data and the subsequent management of data uncertainty is therefore required.

In SR 97, a standardised procedure was employed for all input data to radionuclide transport calculations. The outcome was presented in the SR 97 Data Report /Andersson, 1999/. The uncertainty treatment in SR 97 is discussed by the SKI/SSI review /SKI and SSI, 2001/. The authorities have since conducted some investigations on Expert Judgement /e.g. Wilmot and Galson, 2000; Wilmot et al, 2000; Hora and Jensen, 2002; Hora, 2002/. Also SKB has continued development work /Hedin, 2002, 2003/.

Among other things the reviewers required quantification of uncertainties into a form suitable for probabilistic assessment and traceable records on the expert input to data selection and uncertainty assessment. A new procedure, based on the one used in SR 97 and taking into account review comments is therefore established for SR-Can.

# 1.2 Scope and Objectives

The set of input data parameters to a full safety assessment is very large. Some input data uncertainties will have a decisive influence on safety related output uncertainty whereas others will essentially not influence output uncertainty at all. An obvious example of the latter is transport properties of those radionuclides that never give a significant contribution to the total dose. It is thus appropriate to identify input data to which output is sensitive and use these insights in allocating resources to the determination of input data uncertainties.

Based on such reasoning the SR-Can project team has identified various data where there is a need for a subject specific data report. The information and judgement supplied in these reports are those of the report authors. It should be noted that there will be several different such subject specific data reports. Furthermore, all parameter and uncertainty estimates will also be compiled and further assessed in a SR-Can Data Report. Judgements made in the SR-Can Data Report are usually made by the SR-Can project team (if not explicitly noted else by proper referencing etc).

# 1.2.1 Specific objectives

This report supplies data with uncertainty to the buffer migration properties, i.e. diffusivity (De), porosity ( $\epsilon$ ) and distribution coefficients Kd, for all relevant elements. Also apparent diffusivities Da, should be supplied, although dependent on the first set, as this would simplify comparison with some laboratory data and with other data sets.

# 1.3 Procedures

## 1.3.1 Predefined instructions

The procedure for the data and uncertainty assessment presented in this report is based on:

- a standardized outline with detailed instructions supplied by the SR-Can project team and
- required input (marked by <text within brackets> in this outline) from the experts listed as author(s) to this report addressing the pre-defined instructions.

The instructions are generally marked "Instructions – issues to be addressed" and are usually the first subsection in a given report section. Subsequent subsections contain the input and judgement made by the expert authors.

The standardised outline with instructions aim at establishing:

• Context; the use of the input data in SR-Can, results of sensitivity analyses, correlation to other input data.

- Sources of information (experimental data, site investigation data, model calculations etc).
- Qualitative uncertainty assessment (a description of what types of uncertainty affect the data).
- Quantitative uncertainty assessment (the actual assigning of e.g. data values, data intervals or probability distributions).

### 1.3.2 Experts involved etc

#### Instructions – issues to be addressed

Depending on practicalities and judgements on the importance of the data different procedures for addressing these issues have been used for different sets of data. This section should record the procedures for judgements etc made as regards the data discussed in this report. The following should be addressed:

- Which experts (both SKB internal and SKB external) have been involved in the report and it content? (CV etc may be supplied in appendix).
- What has been there role in this context?
- Have there been special measures for expert elicitation of data and uncertainties (this may e.g. range from single expert judgement, evaluation with the SR-Can data review committee, workshop like procedures etc)? Please provide details (if lengthy in an appendix).

#### Experts and their role for assessing buffer migration data

The scientific content and judgements made in this report are made by the report authors, i.e. Michael Ochs and Caterina Talerico, BMG Engineering Ltd. Their work is based on various scientific and technical input as referenced in the normal scientific way.

The report has been assessed by a subset of the SR-Can Project Team consisting of Patrik Sellin (SKB), Allan Hedin (SKB), Fredrik Vahlund (SKB) and Johan Andersson (JA Streamflow AB). Their role has been to technically review the report and to make sure it serves the need as input to SR-Can and the SR Can Data report. The scientific content has also been reviewed by members of the SKB SIERG committee.

# 2 Use in SR-Can

This chapter briefly discusses how the parameter will be used in SR-Can and if there is any information as regards its importance. As the section concern the interface between safety assessment and the specific model, the content is produced jointly by SKB and the experts producing this document.

# 2.1 Applicable SR-Can conditions

This section should list the various "conditions" for which parameter and uncertainty estimates are needed. "Conditions" refer to boundary conditions, states and other circumstances, which potentially may affect the values of the parameters to be estimated. Changes of a "condition" may be due to various initial states, evolution within a scenario or conditions under different scenarios.

#### Specific for buffer migration parameters

As regards buffer migration parameters data with uncertainties are solicited for the following situations:

- A "full" range of near field groundwater flow.
- A "full" range of near-field groundwater compositions.
- Various deposition hole geometries.
- A range of temperatures.
- Various buffer densities.
- Various potential initial defects.
- Impact of gas transport.

Table 2-1 lists the conditions for which buffer migration data are actually supplied in the report, together with an assessment on the sensitivity to these various conditions. The specific ranges of condition, as well as the assessed sensitivity to these conditions are defined in subsequent chapters of the report.

Table 2-1. SR-Can conditions for burier migration parameters	Table 2-1.	SR-Can	conditions	for buffer	migration	parameters.
--	------------	--------	------------	------------	-----------	-------------

Condition	Importance
groundwater flow, geometry, defects	/
groundwater composition (pe, pCO <sub>2</sub> )	very high
temperature	probably medium
buffer density	high
gas transport	?
bentonite composition (salt inventory, exchangeable cations)	very high
porewater composition (choice of approach)	very high
buffer/porewater evolution (choice of approach)	very high

# 2.2 Models

## 2.2.1 Instructions – issues to be addressed

This section should address the following:

- In what model or other circumstances are the parameter(s) used in SR-Can?
- Will there be differences between ranges of conditions (see 0)?

The actual model used (or other use) should be referred to – not all possibilities. For the latter refer to the Process Report.

## 2.2.2 Models used for buffer migration

The buffer migration parameters are primarily used in the COMP23 (#ref) near-field migration code. It assumes constant values of (De), porosity ( $\epsilon$ ) and distribution coefficients Kd in each compartment.

Migration parameters are also used in assessments of buffer evolution. These analyses primarily concern other elements, but consistency with RN-migration parameters is still needed.

Buffer porosity is a key parameter when assessing earth-quake impacts. Consistency in data choice is needed.

## 2.2.3 Differences in models for different conditions

COMP23 is probably applicable to most foreseen calculation cases. However, there may be problems – or at least need for special considerations – in cases of channels through the buffer – or other cases with inhomogeneous buffer. In cases without buffer (should they be formulated), the buffer migration are not needed anyway!

# 2.3 Results of sensitivity analysis – if any?

## 2.3.1 Instructions – issues to be addressed

This section should discuss whether the parameter has been included in a sensitivity analysis in a Safety Assessment context (provide reference). This concern any sensitivity analysis conducted prior to the final SR-Can analyses (i.e. also analyses set up during the course of producing this report). However, the reason to discuss this here is to help limiting the scope of the data and uncertainty evaluation – not to make final judgements on the importance of different parameters. Overall conclusions on importance are to be made in the SR-Can report.

In addressing importance and sensitivity it is also necessary to consider various calculation endpoints. Endpoints are not restricted to yearly risk (although this is the ultimate endpoint), but also include isolation potential, barrier performance etc (the relevant endpoints will be further determined in coming planning documents to be provided by SKB).

In case sensitivity analyses have been performed, discuss:

- At what ranges of the parameter is the impact on Safety Assessment significant and are there ranges where the impact is negligible? (For example, an elemental solubility larger than say 0.1 mole/dm<sup>3</sup> is unlikely to imply any solubility limitation. Consequently, we need not be very precise in estimating such solubilities as long as it is established that the solubility is "high enough").
- Is the impact monotonous, i.e. higher/lower values will always provide "worse" performance is there an "optimal" value or is the impact complicatedly dependent upon the values of other input parameters?
- What precision is needed to have an impact on Safety Assessment results (this answer may be different for different parameter ranges)?
- Do the answers apply to all scenarios listed in section 0 or only to some?

In answering the above, do consider if the cited sensitivity analyses are sufficiently general to provide definitive answers.

## 2.3.2 Sensitivity of buffer migration parameters

As regards the buffer migration parameters the following may be stated:

- For some nuclides the impact is linear, e.g. for I-129 De is linearly proportional to dose. The impact from  $K_d$  depends on half-life and canister containment time, but may be at least linear for some nuclides, i.e. is isotope specific. Impact on porosity less severe (but porosity affects other buffer conditions as well).
- In general impact from Kd and De is monotonous, i.e. low Kd and high De tend to increase risk. However, if biosphere modelling does not take appropriate account of accumulation etc modelling artefact may result (e.g. U-234 released into the biosphere may not produce high doses, but left to decay in the NF/FF the daughter Ra-226 will produce high doses). Anyway, we should provide arguments in SR-Can why containment, even if restricted in time, always is a good safety feature!
- The impact of porosity is monotonous on RN-migration in the same way as for Kd.
- As regards precision, this should be related to the fact that impact on risk may be essentially linear to parameter value input (see previous).

# **3** Sources of information

This chapter should describe the sources for the estimating the parameter and the uncertainty estimates provided. In complex cases references to underlying reports would usually be needed. Still the text in this chapter should provide enough information to the reader making it possible to understand the factual basis for the values provided in the report. If the source of information is part of a Site Description /e.g. SKB, 2003/ usually only short text, with reference, is needed. All factual information in this chapter is provided by the experts listed as authors to the report

# 3.1 Sources of data

#### 3.1.1 Instructions – issues to be addressed

This section should address the following:

- Which sources of data have been used?
- Are data based on measurements made in the field, in the laboratory or more generic?
- Under which conditions (e.g. water composition, type of rock, temperature etc) are the data measured?
- Who has produced the data?

If the database essentially is obtained from modelling, still try to provide the above information for the data used as input to this modelling. (Make it short though).

#### 3.1.2 Sources of data for buffer migration parameters

Data sources to be considered as basis for selecting diffusion parameters and distribution coefficients for compacted MX-80 bentonite in contact with specific groundwaters need to fulfil two requirements:

- They have to correspond to, or allow the derivation of parameters applicable to the compacted state (1,590 kg/m<sup>3</sup> in this case).
- They have to correspond to, or allow the derivation of parameters applicable to the relevant geochemical conditions. In particular, this includes the porewater composition corresponding to the compacted state.

Both issues are discussed below. Based on the evidence presented, it is concluded that batch sorption data can be applied to compacted systems. In contrast to diffusion studies, batch sorption data are often measured as a function of several conditions, which is very important for the derivation of  $K_d$  values corresponding specifically to the present reference conditions. Further, batch sorption data are comparatively easy to interpret, since they are the direct result of relatively simple experiments that can be well controlled. In contrast, diffusion coefficients need to be extracted from the raw experimental data through models, and care must be taken in interpreting the experimental results (see /Yu and Neretnieks, 1997/, for a discussion of potential experimental problems).

Therefore, the derivation of distribution coefficients and diffusion parameters for the specified conditions was generally based on element-specific batch sorption data and diffusion parameters for HTO (see section 5.3 regarding exceptions for anions and Cs). This is considered as significantly more reliable than a possible attempt of deriving  $K_d$  values based on data for compacted bentonite. A final consistency check was done wherever possible, by calculating  $D_a$  for all reference conditions from the derived  $K_d$  values and diffusion parameters and comparing it with independent experimental  $D_a$  measurements. The good agreement observed further supports the validity of the chosen approach (see section 5.3).

Table 3-1 gives an overview of the advantages and disadvantages of batch sorption vs diffusion experiments. Table 3-2 shows the overall data hierarchy used for the derivation of the present buffer migration parameters. Details of source data selection are given in section 5.3.

Table 3-1. Overview of data sources for buffer migration parameters.

type of experiment	advantage	disadvantage
batch experiments	K <sub>d</sub> directly measured parameter variation process understanding / model	(very) low solid/water ratio
diffusion experiments	relatively close to reality (compacted clay)	somewhat of a black-box results are model-dependent

Table 3-2. Data merarchy for the derivation of burier inigration parameters.	Table 3-2. Data hierarch	/ for the derivation of buffer	migration parameters.
--	--------------------------	--------------------------------	-----------------------

	Sorption	Dif	fusion (general)
1.	large systematic data set / model for MX-80 or similar	1.	data as f(density) in pre-equilibrated bentonite
2.	representative model	2.	data as f(density) for MX-80
3.	few systematic data for MX-80	3.	data as f(density) for other bentonite
4.	few systematic data for other bentonite (1. edge, 2. isotherms)	4.	data for single condition MX-80
5.	single condition datasets for MX-80	5.	" other bentonite
6.	" for other bentonite		
7.	analogy with similar element		
8.	chemical reasoning		
		specific for D <sub>e</sub> :	
		1.	data for mobile elements / HTO are preferred over data for sorbing RN
		2.	data as f(density) are preferred over single-density data
		3.	type of bentonite is of 2 <sup>nd</sup> priority

#### Relevance of data for the compacted state

#### **Diffusion data**

In case of diffusion parameters, the situation is relatively straightforward as most diffusion studies available to date for MX-80 bentonite have been carried out at relatively high degrees of compaction (see section 5.3.4). While few data correspond directly to the present reference density, it is included in the span of densities covered by the various studies. In addition, diffusion coefficients measured as a function of density for other compacted bentonites and clays (Kunigel-V1, Kunipia-F) generally agree with data for MX-80 (see sections 5.3.4 and 5.3.5 for examples) and show the same linear dependence on density. Thus, parameters applicable the reference density can be derived by simple interpolation.

#### K<sub>d</sub> values

The situation is more difficult for the derivation of K<sub>d</sub> values for compacted bentonite. Practically all sorption measurements are carried out in dilute suspensions. Due to experimental difficulties, no reliable sorption measurements in compacted material are available. Moreover, it is extremely difficult to obtain pore solution from compacted bentonite, and the representativeness of extracted solutions is questionable. This situation requires the application of sorption data from batch experiments to compacted material; i.e. the transfer of data from low to very high solid/water ratios. In principle, K<sub>d</sub> is independent of the solid/water ratio, and no transfer is necessary given that the available specific surface area or sorption site density and solution composition remain constant. With regard to the available surface area, /Bradbury and Baeyens, 1998/ measured similar BET values for loose and compacted (2,000 kg/m<sup>3</sup>) MX-80 and Montigel bentonites. The diffusion data by /Kato et al, 1995/ also suggest that no reduction of sorption-available surface area takes place upon compaction. Based on these studies, it is concluded that batch data can be directly applied to compacted systems, provided that any differences in solution composition are accounted for. This point is critical, as further discussed in the following section.

#### Consistency with the porewater composition in compacted bentonite

#### **Diffusion data**

Because of the timeframes typically needed for diffusion experiments, and the difficulties in externally controlling conditions in compacted bentonite, diffusion data are typically only available for a few conditions that are often not well constrained. In particular,  $D_e$  values for reactive elements are very sparse, and/or do not correspond to the appropriate reference conditions. Therefore,  $D_e$  values for most elements were selected based on data for HTO, which are not sensitive to the specific chemical conditions and thus more representative than element-specific data obtained under single conditions. The electrostatic potential in bentonite pores was taken into account in selecting  $D_e$  values for anions and Cs.

#### K<sub>d</sub> values

As pointed out in section 5.1.3, the derivation of  $K_d$  values in this report is based on calculated porewater compositions that are representative for the pore solution in compacted bentonite. There are good indications that most apparent discrepancies between batch and intact systems can be accounted for by calculating the appropriate porewater composition as a function of all relevant factors, including compaction (i.e. solid/water ratio); see /Ochs et al, 2003; Bradbury and Baeyens, 2003a/, as well as the sections on diffusion and sorption in the process report /SKB, 2004/. Because of the variability of geochemical conditions that need to be considered (see sections 2.1 and 4.2), and the conditional nature of sorption data for reactive elements, it is critical that the data sources used provide enough information to allow the derivation of  $K_d$  values that are consistent with the various specified reference conditions.

In general, highest priority was given to systematic sets of high quality data where parameters are given as a function of conditions. If possible, data were used that had been obtained directly for the bentonite of interest to the present data derivation (MX-80). As indicated in Table 3-2, other bentonites as well as single clay minerals (montmorillonite) were also considered:

- A large portion of the systematic sorption data available has been obtained for simplified systems containing montmorillonite rather than whole bentonite. As montmorillonite is the main sorbing component of bentonite (see Appendix B), the respective sorption data are considered as being representative for bentonite. The transfer of sorption data to the conditions specific for the targeted system with MX-80 bentonite can be done using the procedures discussed in section 5.1.
- Most diffusion data available in the literature have been obtained with actual bentonite samples. Because of the importance of bentonite density for many parameters relevant for diffusion, high priority was assigned to datasets that had been obtained as a function of bentonite density. A significant portion of such data had been measured for Japanese bentonites, Kunigel-V1 and Kunipia-F. Their properties are also given in Appendix B). Kunigel-V1 is considered as fairly representative for MX-80, although it contains less smectite. Kunipia-F is a purified material consisting almost exclusively of smectite and may therefore be less representative, particularly at high degrees of compaction. However, the data presented in Figures 5-3, 5-4 and 5-6 show that diffusion data for these bentonites do not differ substantially from data for MX-80.

The main databases used for the derivation of the buffer migration parameters given in the present report are listed below; see also the specific discussion and data presentation for each radionuclide

- A recent Nagra report on selected K<sub>d</sub> values for MX-80 /NTB 02-18; Bradbury and Baeyens, 2003a/, and original data sources cited therein.
- SKB reports on K<sub>d</sub>, D<sub>e</sub>/D<sub>a</sub> and groundwater /SKB TR-97-12, Yu and Neretnieks, 1997; SKB TR-98-03, Laaksoharju et al, 1998/, and original data sources cited therein.
- JNC reports on diffusion in Kunigel-V1 and Kunipia-F bentonites as a function of dry density (mainly by Sato and co-workers).

# 3.2 Use of models for data derivation

# 3.2.1 Instructions – issues to be addressed

This section should address the following:

- Is the database the result of some kind of modelling?
- Specify, and also list critical assumptions made in this modelling.

For example, information obtained from Site Descriptions is usually the result of various modelling activities, with the actually measured data as input (see e.g. /Andersson, 2003/ for an overview). Another example could be the inventory, which results from various burn-up calculations.

# 3.2.2 Modelling used to support the data

Some modelling considerations are made in order to address various surrounding groundwater compositions, various temperatures, various bentonite densities, etc. Here, the use of models in different stages of data derivation needs to be distinguished:

- i) Models were used in defining the various conditions and some of the databases underlying the derivation of buffer migration parameters:
  - Groundwater composition, including pe and pCO<sub>2</sub>.
  - Bentonite composition (salt inventory, exchangeable cations, edge site density).
  - Measured diffusion coefficients (and porosity, in some cases).
- ii) Models were used in this report for calculating underlying system properties:
  - The porewater composition in the buffer cannot be directly measured and was calculated using a surface complexation/ion exchange model.
  - Radionuclide speciation in experimental solutions and bentonite porewater was calculated (using the TDB specified for each element, see Appendix F, and the Davies equation to carry out activity coefficient corrections).
- iii) Models were used directly for parameter derivation in this report:
  - For Cs, Sr/Ra, Ni, and Eu (as an analogue for Ce, Ho and Sm) thermodynamic sorption models where directly used to calculate K<sub>d</sub> values.
  - For the selection of  $D_e$  values, results of electric double layer model calculations were considered.

```
4 Uncertainty assessment
```

This chapter assesses in general terms the different sources of uncertainty affecting the parameter estimates. The actual estimates of the parameter(s) and associated uncertainties, are presented in the next chapter. All factual information in this chapter is provided by the experts listed as authors to the report.

# 4.1 Overview

#### 4.1.1 Instructions – issues to be addressed

The expert may provide an overview of the different sources of uncertainty affecting the data estimates. (i.e. this overview section is optional).

#### 4.1.2 Overall uncertainties (experts view)

Figure 4-1 provides an overview of the various contributions to overall uncertainty of the recommended data. In order to define overall uncertainties in a traceable fashion, is important to distinguish between uncertainties at the following different levels:

#### 1. Uncertainty at the source condition:

In the case of  $K_d$  values, this corresponds to the experimental error. In the case of diffusion coefficients, this corresponds to the experimental error as well as any uncertainty introduced in the required modelling for raw data reduction. Additional conceptual uncertainties are introduced in the interpretation of the diffusivity and diffusion-available porosity of anions (as well as of certain mobile cations, in particular Cs).

#### 2. Uncertainty at defined, PA-relevant application conditions :

Almost invariably, the conditions relevant for PA will not be covered exactly by matching experimental data. This necessitates the conversion of the source data to the application conditions (condition A in Figure 4-1) through models or estimation procedures. The overall uncertainty at the application condition will then include any uncertainties introduced by the applied conversion procedures in addition to the uncertainties already listed under point 1 above.

#### 3. Uncertainties in the application conditions themselves:

If there are significant uncertainties associated with the application conditions themselves, it is critical to take the conditional nature of the relevant migration parameters into account.  $K_d$  values in particular are highly conditional and need to be derived for each specified set of (expected) conditions. In the present report, variability of geochemical conditions was therefore addressed by deriving  $K_d$  values separately for several sets of possible geochemical conditions.



*Figure 4-1.* Overview of the various sources of overall uncertainty at different levels, illustrated for the derivation of  $K_d$  values.

At the same time, it is not possible to cover the expected variability in geochemical conditions by considering only an individual chemical parameter (e.g. pH). Because of the interplay of the various geochemical factors (pH, carbonate concentration...) in affecting RN behaviour, the expected variability has to be addressed by providing complete and self-consistent sets of porewater composition and bentonite characteristics corresponding to given boundary conditions.

The following sections describe in some detail the general aspects of uncertainties related to the three points above. Conceptual uncertainties regarding sorption and diffusion processes are discussed in the process report, these discussions are not repeated here. An overview of how overall uncertainties are treated in the data derivation process can be found in section 5.1.2.

# 4.2 Uncertainty related to different conditions

## 4.2.1 Instructions – issues to be addressed

This section should discuss if and how the parameter is influenced by the various conditions as specified in section 0 and how uncertainty is influenced. (For example, solubility depends on groundwater composition, which in term may depend on the scenario due to differences in future groundwater flow evolution for e.g. climatic change scenarios and stable climate scenarios.)

Also consider whether there are other conditions, but those specified in section 0 that influence the data?

In answering, the expert(s) should separate between their own opinion and what they would suggest being the range of answers provided by other acknowledged scientist/experts in the field. (If their opinion is the same as all others – state this as well).

# 4.2.2 Conditions affecting buffer migration parameters

Figure 4-2 illustrates that variations in geochemical conditions influence RN migration not directly, but through their influence on the composition of the porewater in compacted bentonite. The latter has to be derived based on model calculations, which introduces an additional (conceptual) uncertainty (see section 4.3). The overall resulting uncertainties in buffer migration parameters were assessed based on different defined sets of conditions and the corresponding porewater compositions (see section 5.2.5).

One of the main conditions influencing the calculated porewater composition is the way  $pCO_2$  in the bentonite buffer is handled: If the buffer is assumed to be closed with respect to an exchange of  $CO_2$  with the surrounding host rock (or if the kinetics of this exchange are sufficiently slow),  $pCO_2$  in the buffer is the result of bentonite-groundwater interaction and, thus, is determinated by groundwater composition and bentonite properties. The edge sites are the most important pH buffer system in this case. If the bentonite buffer is assumed to be open with respect to an exchange of  $CO_2$  with the surrounding host rock, the various carbonate equilibria make up the main pH buffer system and porewater pH in the bentonite is controlled by the external  $pCO_2$  imposed by the host formation.

In view of the developments made in the derivation of RN migration parameters in the last decade or so, it is fair to state that the present approach, fully taking into account the dependency of  $K_d$  on the porewater composition of the compacted buffer, represents the state of the art /cf Bradbury and Baeyens, 2003a,b/. To date, there is substantial consensus /cf OECD/NEA, 2004/ that the appropriate description of overall solution and geochemistry is the key to minimising the uncertainties in  $K_d$ , irrespectively of whether this is done through thermodynamic sorption models or estimation procedures (see section 5.1.4). How, in detail, geochemical conditions and bentonite properties should be taken into account for the calculation of the buffer porewater composition still is a matter of debate to some degree (see section 5.2.5). However, it seems that the different approaches lead to approximately comparable results.



**Figure 4-2.** Illustration of the influence of conditions on buffer migration parameters: The calculated composition of bentonite porewater is the central factor governing RN migration; variations in geochemical conditions are only effective through porewater composition (Figure 4-3). This model calculation also introduces some conceptual uncertainties (see section 4.3). The degree to which these can be covered based on present knowledge is indicated by the sequence 1) better than 2) better than 3).

# 4.3 Conceptual uncertainties

## 4.3.1 Instructions – issues to be addressed

With reference to the Process Report and the actual modelling in SR-CAN (see section 2.2), this section should discuss:

- Are there conceptual uncertainties related to the model where the parameter is used?
- Are there conceptual uncertainties related to models used for deriving the parameter value?
- In light of the previous point, can the conceptual (model) uncertainty be expressed/ illustrated as a parameter uncertainty in the given model (see chapter 2). (For example, the modelling uncertainty related to various sorption processes may possibly be handled by an increased range in K<sub>d</sub>).

In answering, the expert(s) should separate between their own opinion and what they would suggest being the range of answers provided by other acknowledged scientist/experts in the field. (If their opinion is the same as all others – state this as well).

# 4.3.2 Conceptual uncertainties in buffer migration parameters

For the derivation of buffer migration parameters, the conceptual uncertainties discussed below were identified. Note that only point 1 refers to conceptual uncertainties directly related to the data derivation process; points 2–4 are related to underlying databases and models.

- 1. Several related conceptual uncertainties exist regarding the interpretation of, and selfconsistency among, batch K<sub>d</sub> values and diffusivities of sorbing RN on one hand, and of diffusivities and diffusion-available porosities of anions on the other. Both issues are discussed to some detail in the SR-Can Process Report /SKB, 2004/ and in section 5.3.4. These issues are also related to the following point:
- 2. There are some open questions regarding the fundamental, underlying chemistry of radionuclides in aqueous solutions. This is also acknowledged in recent TDB compilations /e.g. Hummel et al, 2002/. For example, the importance of actinide(III)-silicate, mixed actinide(IV)–OH-CO<sub>3</sub> and Ni-CO<sub>3</sub> complexes is not established to date. There are also uncertainties regarding the solution speciation of many of the less well researched elements, such as Nb, Zr, etc. The TDB used for the present data derivation is indicated for each element in Appendix F. Use of different TDBs, or future updates in SKB's reference TDB could lead to different migration parameters (see example Am).
- 3. There are significant scientific shortcomings regarding the derivation of porewater composition in compacted bentonite and its evolution over time under repository conditions. For the present purpose, it is important to note that the porewater composition of compacted bentonite cannot be determined experimentally with any certainty, and is therefore calculated through thermodynamic surface chemical models. Several published models are available for this purpose. While they are based on the same principles, they differ in a number of details regarding e.g. the treatment of specific surface chemical equilibria. These differences are small in comparison to other uncertainties, however. Further questions regarding the effects of electrical double layers in the porespace on e.g. the amount of "free" water, water activity etc are clearly beyond the present scientific understanding.
- 4. There are uncertainties related to the interpretation of raw diffusion data (concentration profiles, fluxes) by different researchers.

# 4.4 Data uncertainty, spatial and temporal variation

### 4.4.1 Instructions – issues to be addressed

This section should address the following:

- What is known about the spatial variation (e.g. scales, variography, discrete feature statistics,..) of the parameter? Is there any information about the uncertainty in the spatial variability? (Usually all this information can be supplied by referring to the Site Descriptive Model report for the different sites). How is this considered in the parameter and uncertainty estimates?
- What is known about the temporal variability of the parameter? How is this considered in the parameter and uncertainty estimates?
- If the parameter value and its uncertainty is drawn from a database, is this site specific or "generic"? In the latter case, how would the lack of site specific data influence the uncertainty?
- Are parameter and uncertainty estimates based on analyses of field/laboratory data? Are there any measurement errors etc and how are they considered in the uncertainty estimates?
- If data for estimating the parameter have been produced using a model, what uncertainties does this introduce? Conceptual uncertainties in the model (e g EQ3/6, ORIGEN)? Uncertainties in model input (e g cross sections in ORIGEN)? In particular, the uncertainty discussion in the Site Descriptive Model reports /see e.g. SKB, 2003/ should be drawn upon.

In answering, the expert(s) should separate between their own opinion and what they would suggest being the range of answers provided by other acknowledged scientist/experts in the field. (If their opinion is the same as all others – state this as well).

## 4.4.2 Spatial, temporal and site-specific variation

On the scale of a typical buffer, bentonite can be considered homogeneous. Therefore, spatial variation is not considered relevant for a bentonite buffer.

Temporal variation becomes important when buffer/porewater evolution is being considered. As was already discussed for the definition of porewater composition in general (section 4.2.2), the evolution of porewater and buffer composition over time has to be assessed through models. The corresponding uncertainties can be viewed as an extension of the uncertainty introduced by variable groundwater chemistry. As no independent model results for the scenarios of SKB were available at the time of the present data selection, some extreme situations were included in the porewater derivation. These included interactions of groundwater with bentonite converted to the Ca-form or depleted of impurities (see Figure 4-3).

The sorption and diffusion data as well as the various models underlying the present data selection are generic and therefore not site-specific. However, the extracted buffer migration parameters will be site-sensitive to the extent that they were derived based on site-specific conditions (including the respective variability). Of particular importance are groundwater composition (including redox conditions) and pCO<sub>2</sub> imposed by the host rock formation.



**Figure 4-3.** Influence of conditions on porewaters in compacted bentonite. The reference condition (Beberg saline groundwater, BFI01  $pCO_2 = 1x10^{-2.6}$  atm) leads to the porewater RPW, all other compositions address the respective variability. The compositions RPW, RPWC and HSPW were applied in the derivation of parameters for each RN, the other compositions in selected cases only. See Appendix C for details on the porewater composition.

# 4.5 Correlations among migration parameters

## 4.5.1 Instructions – issues to be addressed

The extensive work with the FEPs databases and the Process report should imply that most functional dependencies between parameters should be identified – and the important one also implemented in the Safety Assessment models. Also the assessment of impacts from various conditions should cover most potential correlations. Still other statistical correlation may exist. This section should address the following questions:

- If the data varies in space or time is anything known about its autocorrelation structure?
- Is there any other reason (apart from already cited functional relations etc) to suspect correlation between parameters considered as input to SR-CAN?

In answering, the expert(s) should separate between their own opinion and what they would suggest being the range of answers provided by other acknowledged scientist/experts in the field. (If their opinion is the same as all others – state this as well).

# 4.5.2 Correlations among buffer migration data

The following important correlations can be identified

• Following their chemical characteristics, the RN considered can be organised into groups of elements and oxidation states whose migration behaviour will generally show a similar response to variations in porewater composition caused by e.g. variations in groundwater composition, bentonite evolution, etc. Moreover, elements handled via chemical analogies obviously correlate with the respective analogues. Overall, the following grouping is suggested, where analogies are also indicated (X/Y: both elements were treated identically in the data derivation; X(Y): X was derived based on analogy with Y)

- 1. Alkaline and alkaline earth elements: Cs, (Ra/Sr).
- 2. Other di-valent elements: Pb, Ni.
- 3. Tri-valent elements: Am, (Cm/Am), PuIII(Am), Sm/Ho/Ce(Eu).
- 4. Tetra-valent elements: Th, UIV(Th), PuIV(Th), NpIV(Th). Due to lack of reliable element-specific data, Tc(IV), Zr and Sn(IV) were also evaluated on the basis of data for Th, and the derived K<sub>d</sub> values show therefore some correlation (because Th is not an ideal analogue for these elements, additional uncertainties are associated with the resulting K<sub>d</sub> values).
- 5. Penta-valent elements: NpV, PuV(NpV).
- 6. Hexa-valent elements: UVI, PuVI(UVI).
- 7. Non-sorbing anions:  $(Cl^{-}/I^{-}/TcO_{4}^{-}/SeO_{4}^{2-}/HSe^{-}/simple organic anions)$ .
- 8. The weakly sorbing anions  $\text{SeO}_3^{2-}$  and carbonate will not correlate as it is proposed to handle the latter via isotope exchange.
- 9. Some elements are not known well enough to assess correlations: Pa, Nb, Pd, Ag.
- 10. Gases: (Rn, CH<sub>4</sub>).
- The redox-sensitive radionuclides will take on higher oxidation states if oxidising conditions are considered, generally leading to lower K<sub>d</sub> values. An exception is: Se(−II→ IV).
- A lower porewater pH (within the range presently considered) will decrease the K<sub>d</sub> of most radionuclides (actinides, lanthanides, transition elements, heavy metals) in a similar way. An exception are penta-and hexavalent actinides that form oxo-cations: U(VI), Np(V), Pu(VI/V).
- Similarly, an increase in major cation concentration will lower K<sub>d</sub> values for alkaline and alkaline earth elements.
- A lower density of the buffer will lead to higher D<sub>e</sub> and D<sub>a</sub> values.

# **5** Quantification of data and uncertainties

In this chapter the various sources of uncertainty are combined in order to provide quantitative uncertainty estimates of the parameter(s) being analysed.

In this chapter, the actual estimates of the parameters ( $K_d$ ,  $D_e$  and porosity in this case, with  $D_a$  as an additional parameter for comparisons) and of the associated uncertainties are made. Uncertainties from different sources are combined to give overall values.

# 5.1 Methods and approaches

#### 5.1.1 Instructions – issues to be addressed

The uncertainty estimates should be made subjectively, i.e. "what is the believed probability that a parameter has a value (less than) a certain number" – not actually measured ranges – although the latter could have a large impact on the subjective probabilities). (But we should also be aware of 'risk dilution').

If the parameter shows spatial (or temporal) variability:

- Consider how this variability is at the scale of description not at the measurement scale, some of the measured/simulated variability may in fact be averaged out at the scale of description other would not,
- Describe how the variability is accounted for. Is it included in the uncertainty estimate or given as stochastic process etc with different realisations? Are there alternative models for describing the variability if so are they equally likely?

Describe correlations.

#### 5.1.2 General approach

The general approach used for the derivation of buffer migration data and uncertainties is described in chapters 2.3.1 and 4. Details regarding the various conditions considered are given in the following section 5.2. As outlined in Figure 4-1, data and uncertainties are defined for three levels:

Level 1: data source

Level 2: reference condition

Level 3: further (alternative) conditions, reflecting the variability in conditions at level 2

The likelihood for any data to fall within the recommended ranges is expressed using "soft terminology". This is supported by consistency checks using independent data and a traceable and extensive documentation of data derivation in appendices. The likelihood for a given range to encompass all possible data for the given conditions is expressed as (decreasing confidence):

• extremely likely,

- very likely,
- likely,
- probable.

To cope with the necessity to convert data from experimental to PA-conditions, the data selection procedure relied strongly on sorption data obtained in batch experiments. For the reasons given in section 5.3.4, no element-specific  $D_e$  values were derived for reactive elements; instead, the selected  $D_e$  value for HTO was relied upon.  $D_e$  values for anions and Cs were selected to take into account the electrostatic potential in bentonite pores. Data derivation and assessment of uncertainties was carried out in four steps as described in the following sections:

- 1. Definition of all conditions to be considered, and calculation of the bentonite porewater composition corresponding to the reference density (see section 5.2). Uncertainties in boundary conditions (groundwater and bentonite characteristics) are addressed by carrying out the derivation of migration parameters for several self-consistent bentonite/ porewater systems. No further uncertainties are assumed for each of these systems.
- 2. Derivation of D<sub>e</sub> values and diffusion-available porosity (ε) for the specified reference density (sections 5.3.3 and 5.3.4). Uncertainties related to the influence of conditions on diffusion-relevant bentonite characteristics (poresize, etc) are assumed to be included in the uncertainties given for D<sub>e</sub> in each case, as these data are derived based on experimental measurements that already cover different conditions.
  - a) Selection of D<sub>e</sub> values and associated uncertainties for HTO, to be used together with the entire physical porosity for all elements except non-sorbing anions and Cs (see below), and for all conditions. Because D<sub>e</sub> refers to steady-state conditions, the influence of uncertainties in conditions through radionuclide speciation and average ionic charge is assumed to be contained in the uncertainty given for D<sub>e</sub> of HTO.
  - b) Selection of D<sub>e</sub> values and associated uncertainties for all non-sorbing anions (Cl<sup>-</sup>, I<sup>-</sup>, TcO<sub>4</sub><sup>-</sup>, SeO<sub>4</sub><sup>2-</sup>, HSe<sup>-</sup>, simple organic anions) to be used for all conditions together with the selected anion diffusion-available porosity. Uncertainties in radionuclide speciation are not relevant in this case.
  - c) Selection of D<sub>e</sub> value for Cs, to be used together with the physical porosity. Uncertainties in radionuclide speciation are not relevant in this case.
- 3. Derivation of  $K_d$  values for each element (section 5.3.5, see also Figure 4-1):
  - a) Selection of source data and quantification of their experimental uncertainty.
  - b) Conversion of source data to the selected reference conditions and quantification of the additional uncertainties introduced in this process.
- 4. Calculation of D<sub>a</sub> values for the specified reference density based on the results for points 2 and 3 above, and comparison with independent experimental data. Final assessment of overall consistency and uncertainty.

# 5.1.3 Definition of conditions

The conditions to be considered in the data derivation were defined through calculated porewater compositions (Appendix C), which were based on defined combinations of

• the chosen buffer material, MX-80, at the reference density of 1,590 kg/m<sup>3</sup>, including some alternative compositions;

• the relevant groundwater composition given in /Laaksoharju et al, 1998/, as well as some modified compositions (see Appendix A).

Further details are given in section 5.2.

#### 5.1.4 Derivation of K<sub>d</sub> values

The general hierarchy followed in selecting source data is laid out in Table 3-2. It shows that high priority is given to systematic data obtained on MX-80 bentonites. It is believed that the uncertainty introduced in extrapolating sorption data from source to the present target or application ("PA") conditions can be significantly higher than experimental uncertainties (see Figure 4-1) and will increase with the extend of extrapolation. It is further proposed that this holds for any extrapolation method, i.e. both thermodynamic model applications as well as semi-quantitative estimations. Therefore, systematic data obtained on MX-80 bentonite in solutions that closely match the present application conditions may be preferred over thermodynamic sorption models that had been calibrated in the absence of key chemical parameters, such as carbonate. While thermodynamic models in principle have the better scientific foundation, they are often not sufficiently constrained in terms of parameterisation. Thus, application of a model calibrated exclusively with data from simple systems to the complex MX-80/porewater system often involves an extensive extrapolation, which limits the value of the model. Recent results from the NEA Sorption Project /NEA, 2004/ also indicate that apparent shortcomings in the application of sorption models may often be traced back to limitations in parameterisation rather than to any fundamental problem with the model itself. In some cases, it would be worthwhile to consider recalibration or development of sorption models based on available experimental data; however, model development was outside of the present scope.

In their recent report on near-field sorption data, /Bradbury and Baeyens, 2003a/ report a significant amount of supplementary and systematic sorption data for many radionuclides that had been obtained with MX-80 in equilibrium with a synthetic porewater. Because of the chemical similarity of their experimental conditions with the present application conditions, their work could be used in many cases as starting point. In case of several radionuclides, the experimental data made available in their report represent the most relevant and reliable dataset for MX-80 that could be found. A specific discussion for each element is given in section 5.3.

In general, the selection of source data was aimed at obtaining the best available starting values for the further data derivation. Thus, subjective choices (e.g. consideration of "conservatism") were completely avoided, as they would lead to a decrease in transparency and traceability. These best available starting values are listed under the column "Data Source" in the data derivation tables (Appendix E).

#### **Conversion procedures**

The derivation of  $K_d$  values corresponding to the desired conditions in compacted bentonite from experimental data almost invariably involves an extrapolation step. Where possible, this extrapolation was done with the help of an appropriate thermodynamic sorption model. Where such a model is not available, the semi-quantitative conversion procedures documented below were applied. As explained above, these procedures were also preferred in case where it would allow to use data that correspond significantly better to the desired application conditions than the data underlying a model. The employed extrapolation procedures are related to the sorption capacity of bentonite and the behaviour of the various RN under the different conditions. The basic approach is taken from /Bradbury and Baeyens, 2003a/, and the conversion procedure related to sorption capacity was accepted as is. On the other hand, the conversion procedures related to surface and radionuclide speciation were interpreted differently, as detailed below. The actual conversion factors were in all cases calculated specifically for the present source data sets and application conditions.

#### Sorption capacity

The sorption capacity of a typical bentonite, such as MX-80, is determined by its content of smectite clay minerals (typically dominated by montmorillonite). While most bentonites contain a number of important impurities, these are often not very relevant for sorption reactions and present evidence suggest that smectite minerals are dominating the actual sorption processes through surface complexation and ion exchange, at least in typical bentonites. This is supported by the work of /Wanner et al, 1996; Ochs et al, 1998; Pabalan and Turner, 1997; Bradbury and Baeyens, 2002b/, as well as by the results of the recent NEA modelling exercise /NEA, 2004/. Of course, through their large effect on porewater chemistry, impurities are indirectly important for sorption.

To scale selected  $K_d$  values obtained for bentonites or clays other than MX-80 to the MX-80 application conditions, a conversion factor based on the respective CEC values was applied following /Bradbury and Baeyens, 2003a/:

CF-CEC = CEC (application) / CEC (data source) (5.1)

In this relation, CEC is used as a measure for total site density, based on the reasonable assumption that the density of surface complexation (edge) sites is proportional to the CEC.

#### pН

The pH is the most critical geochemical parameter for the sorption of most elements, as it determines both the speciation of the surface complexation sites as well as of the sorbing element itself. Note that pH does not only determine the hydrolysis of a RN, but also its interaction with other ligands, in particular those that undergo (de)protonation in the relevant pH range (e.g. carbonate). This is illustrated in Figure 5-1.

Where possible, data sources were selected that allowed the selection of  $K_d$  values at the pH values corresponding to the application conditions (in the data derivation tables in Appendix E, this is indicated by a pH-conversion factor of 1). Where this was not possible, scaling to the appropriate application pH could in many cases be made on the basis of additional data for the same RN, such as sorption edges on a montmorillonite. In such cases, scaling to the application pH was done following /Bradbury and Baeyens, 2003a/, using the conversion factor defined below:

 $CF-pH = K_{d}(pH \text{ data source}) / K_{d}(pH \text{ application})$  (5.2)

In contrast to /Bradbury and Baeyens, 2003a/, CF-pH is interpreted as a factor addressing the overall surface/solution speciation. Thus, the conversion factor for additional radionuclide speciation effects defined below (CF-spec) addresses only effects not included in CF-pH, i.e. does not take into account hydrolysis.



**Figure 5-1.** Schematic illustration of the dependency of  $K_d$  on pH. M stands for a hydrolysable RN (metal), surface speciation is indicated in blue. In the absence of other ligands, sorption is low at low and high pH because of the strong competition by  $H^+$  for the available surface sites and by dissolved OH<sup>-</sup> for the RN, respectively. Ligands that are not protonated at relevant pH (indicated in grey) can compete for the dissolved RN at any pH. On the other hand, the concentration of  $CO_3^{2-}$  is pH-dependent itself (at a given total carbonate level), and strong competition by the carbonate ion takes place only at high pH.

#### Additional effects of other ligands on radionuclide surface/solution speciation

As illustrated in Figure 5-1, the presence of dissolved ligands can have a significant influence on radionuclide sorption. Therefore, scaling of  $K_d$  to application conditions should be carried out in all cases where the source data correspond to a solution composition different from that of MX-80 porewater (which is nearly always the case). Unfortunately, only thermodynamic sorption models are capable of consistently taking into account all speciation effects, but such models are available for a few elements only. Therefore, it was attempted to account for variable solution compositions in a semi-quantitative, but fully traceable fashion.

The conversion factor CF-spec takes into account differences in the competition for the RN by those dissolved ligands present in the different solutions that are not already covered by CF-pH. Again, the formal definition of this factor is similar as in /Bradbury and Baeyens, 2003a/.

$$CF-spec = F_{sorb} (application) / F_{sorb} (data source)$$
(5.3a)

where

$$F_{sorb} = (RN_{tot} - RN_{cmp}) / RN_{tot}$$
(5.3b)

with  $RN_{tot}$  as the total dissolved concentration of a given radionuclide, and  $RN_{cmp}$  as the total concentration of all dissolved RN complexes evaluated as being competitive with regard to sorption.

However, the application of this factor is not straightforward, and the evaluation procedure used here differs considerably from that of /Bradbury and Baeyens, 2003a/. First, Figure 5-1 shows that the conversion factor CF-pH will already include one or more of the effects of speciation. The following differences in speciation are therefore not included in the factor CF-spec, as the effects would be counted twice:

- Effects of RN hydrolysis will be included in CF-pH, except at very dissimilar RN concentrations or in cases where a K<sub>d</sub> has to be based on an analogy with a somewhat dissimilar element (see the example of Zr).
- Complexation of a given RN with the carbonate ion (CO<sub>3</sub><sup>2-</sup>) can differ between two solutions because of i) different total carbonate concentrations, or ii) different pH values leading to different concentrations of CO<sub>3</sub><sup>2-</sup> even when total carbonate levels are the same. In case ii), the effect will also be included in the conversion factor CF-pH.

Second, several elements, especially IV- and higher-valent actinides are known or suspected to form mixed hydroxo-carbonato complexes. By analogy, the formation of ternary RN-carbonato surface complexes at the clay edge sites can be assumed. For some cases, the existence of such complexes has been confirmed by spectroscopy (e.g. for UVI sorption on iron oxide by /Bargar et al, 2001/). Therefore, it is highly questionable in these cases whether the presence of carbonate should be counted as a competitive factor. Therefore, scaling of  $K_d$  to account for speciation was carefully evaluated for each element. As a basis for further evaluation, CF-spec was calculated twice in many cases, taking into account or neglecting competition by complexation involving carbonate ions.

- CF-spec 2: Complexation of a radionuclide with carbonate is viewed as being fully competitive with respect to sorption.
- CF-spec 3: Complexation of a radionuclide with carbonate is viewed as being not competitive with respect to sorption. One reason to use CF-spec 3 may be the formation of ternary surface carbonato complexes, which would contribute to overall sorption. Further, CF-spec 3 should also be used where effects of carbonate complexation are already included in CF-pH; i.e. CF-spec 3 would then only include competition by e.g. chloride or sulphate.

The above points are particularly critical, as hydrolysis and formation of carbonato complexes are in many cases the most important speciation effects (i.e. CF-spec is often of minor importance).

#### Overall modification (scaling) of selected K<sub>d</sub> values

Selected Kd values were scaled to the appropriate application conditions using an overall conversion factor:

$$CF$$
-total =  $CF$ - $CEC \times CF$ - $pH \times CF$ - $spec$  (5.4)

Following the above discussion, for a number of elements CF-total had to be defined for different ways of scaling with respect to speciation:

$$CF-total 1 = CF-CEC \times CF-pH$$
(5.5a)

where speciation was considered to be completely included in CF-pH; and

 $CF-total 2 = CF-CEC \times CF-pH \times CF-spec 2$ (5.5b)

 $CF-total 3 = CF-CEC \times CF-pH \times CF-spec 3$ (5.5c)

depending on how carbonate complexation was evaluated. The final choice made is discussed for every element in section 5.3.5.

#### Uncertainties

Uncertainty estimates for  $K_d$  are based on the general procedure outlined in Figure 4-1, distinguishing between

- uncertainty at a given application condition,
- uncertainty as a function of variable conditions.

#### Uncertainty estimates of K<sub>d</sub> for a given application condition

There is no obvious reason for choosing between quantifying uncertainties as an error on a linear scale (i.e,  $K_d \pm error$ ) vs a logarithmic scale (i.e,  $\log K_d \pm \log error$ ). Because several careful sorption studies /e.g. Bradbury and Baeyens, 1997/ report errors that are symmetric on a log scale, the latter type of representation was adopted for the present purpose. On a linear scale, this translates to  $K_d$  multiplied/divided by an uncertainty factor UF:

log K <sub>d</sub> (upper, lower lin	nit) = $\log K_d \pm \log$ uncertainty, or	(5.6a)
log K <sub>d</sub> (upper, lower m	$m(t) = \log \kappa_d \pm \log uncertainty, or$	(3.00

$$K_d$$
 upper limit =  $K_d \times UF$  and  $K_d$  lower limit =  $K_d / UF$  (5.6b)

As statistical distribution functions can only be assumed, it is not realistic to assign percentiles (likelihood for a datum to be within the indicated range at a statistical confidence level) in a standard way to the upper/lower limits. Instead, uncertainties are evaluated in a way that makes it generally very likely that the indicated limits encompass all possible values. Where data are more uncertain, this is discussed specifically. For each element, this is supported with illustrations and consistency checks using diffusion data to facilitate an independent interpretation by the user of this report, where required.

The following uncertainty factors (log errors) are considered:

- 1. Uncertainty of source data (UF-starting K<sub>d</sub>)
  - Experimental data

For good quality experimental data an uncertainty of log  $K_d \pm 0.2$  log units is proposed based on the findings of /Bradbury and Baeyens, 1997/. This gives an uncertainty factor

- $\rightarrow$  UF-starting K<sub>d</sub> = 1.6
- Model data:

/Bradbury and Baeyens, 2003a/ suggest an uncertainty factor of 3. However, a recent extensive modelling exercise /NEA, 2004/ showed that good sorption models are able to reproduce experimental data very closely, and we propose to use the same uncertainty as for experimental data.

- 2. Uncertainty introduced by scaling to application conditions
  - CEC conversion (UF-CEC):

It is estimated that CEC can be measured within an uncertainty of ca 10%, and that SOH density can be scaled via CEC again within an uncertainty of ca 20%. This gives an overall uncertainty of  $\pm$  30% or about a factor of 2 between highest and lowest value. Thus, we propose

 $\rightarrow$  UF-CEC = 1.4
- pH conversion:

This is only needed where CF-pH  $\neq$  1, i.e. where scaling to application pH had to be done via additional data and equation (5.2). Because CF-pH contains uncertainties of two K<sub>d</sub> values (read off at pH-data source and pH-application, respectively), an uncertainty log starting K<sub>d</sub>  $\pm$  2×0.2 log units is used:

 $\rightarrow$  UF-pH conversion = 2.5

– Speciation (UF-speciation):

Following the evaluation by /Hummel and Berner, 2002/, who propose a factor of 2 between highest and lowest value, a UF-speciation = 1.4 is used. Note, however, that this is the uncertainty associated with the use of a given complete TDB. If certain species are missing or erroneous (see e.g. the discussion on mixed actinide OH-CO<sub>3</sub> complexes above), uncertainties could be much higher. Also, this UF does not take into account any inappropriate evaluation of CF-spec, as discussed above (this had to be evaluated with consistency examinations and "what-if" calculations as described above, rather than via formal uncertainties).

- Analogy considerations:

Where a  $K_d$  was based entirely on data and chemistry of an analogue element, an additional uncertainty factor UF-analogue was introduced; this is discussed for each element. Where a  $K_d$  was derived using sorption data for an analogue element but speciation was evaluated directly for the element under investigation (e.g. in the case of using sorption data for Th to evaluate  $K_d$  for PuIV), the additional uncertainty was taken into account by using for log starting  $K_d \pm 2 \times 0.2$  log units in case of good analogues (e.g. tetravalent actinides),

 $\rightarrow$  UF-starting K<sub>d</sub> = 2.5

or log starting  $K_d \pm 2 \times 0.3$  log units in case of more questionable analogues (e.g. use of Th for Zr),

 $\rightarrow$  UF-starting K<sub>d</sub> = 4.0

 Conversion of batch data to conditions in compacted bentonite: Arguments for the applicability of the K<sub>d</sub> values derived in this report to compacted conditions are presented in section 3.1.2. Accordingly, CF-batch→ compacted is unity and no UF is proposed for this conversion per se. However, an additional uncertainty factor is introduced to acknowledge possible uncertainties in the application conditions (i.e. in the porewater composition, resulting from model uncertainties and effects from variations in the amount of impurities present in bentonite samples; /see Ochs et al, 2004/). Based on the differences in the K<sub>d</sub> values derived for three difference reference porewaters (section 5.3.5), this uncertainty factor is set to two: → UF-batch→ compacted = 2.

# 5.2 Conditions for which parameter values are to be supplied

#### 5.2.1 Instructions – issues to be addressed

Based on the assessment of impact from the various conditions made in chapter 4, it may be needed to split up the data and uncertainty quantification into different set of conditions. This section should specify this set of conditions, with justification.

# 5.2.2 Buffer migration parameters

The impact from density may possibly be handled by providing a functional relation, i.e. a wide range of densities can be handled by a single set of data – provided this relation is used.

The impact from groundwater composition may probably require different tables of data and uncertainty for different type waters. Such a case would correspond to different conditions – one for each type water. However, it is not totally clear if there will be type waters in SR-Can. Furthermore, we are interested in the transition between different type waters, i.e. there should be a strive to describe impact from groundwater composition with functional relations, i.e. there may be a threshold value for the impact from Eh, but a more continuous impact from pH and TDS?. Possibly, the best way to handle the situation would be to supply a full set of data and uncertainties for a selection of type conditions and then add how to handle cases in-between these type situations.

# 5.2.3 Groundwater

The selected groundwater conditions are based on /Laaksoharju et al, 1998/. It is realised that all PA calculations (radionuclide transport, buffer evolution, etc) should be based on consistent groundwater compositions. For the purpose of the present SR-CAN report it is considered by SKB that the saline Beberg water (BFI01) fulfils this requirement and was therefore accepted as reference groundwater. Alternative groundwater compositions, covering what was estimated to be a reasonable range in terms of salinity and pH, included a non-saline Beberg groundwater as well as a hypothetical water where the salinity of the saline Beberg water was raised to that of typical seawater. In an exploratory evaluation, the range of hypothetical groundwaters was extended to include alkaline and hyper-alkaline groundwater. Possible effects of salt-saturated groundwater are discussed briefly in Appendix C.2.

These groundwaters are the basis for determining the final bentonite porewater composition. The groundwater compositions are given in Appendix A.

# 5.2.4 Bentonite density and composition

The effect of density was evaluated for De and  $\varepsilon$ , and quantitative relations are given in sections 5.3.3 and 5.3.4. The evaluation of sorption data was carried out only for the reference density of MX-80 bentonite (1,590 kg/m<sup>3</sup>). However, there is clear evidence that the porewater composition and, therefore, radionuclide sorption, is not significantly influenced by limited variations in buffer density. In comparisons to uncertainties regarding e.g. externally imposed pCO<sub>2</sub>, such effects are negligible.

The bentonite composition considered is given in Appendix B. The following variations were considered for some cases:

- Bentonite converted completely to the Ca-form.
- Bentonite completely depleted of soluble impurities (NaCl, KCl, gypsum).

# 5.2.5 Porewater compositions

As illustrated in Figure 4-2, the bentonite porewater corresponding to the reference density forms the central element for the evaluation of migration parameters, in particular  $K_d$ .

The pore solution in compacted bentonite is practically inaccessible for chemical analysis. Even at very high squeezing pressures it is difficult to obtain sufficient quantities of pore solution, and it is questionable whether the sampled solution is actually representative. Therefore, porewater compositions were evaluated by using thermodynamic models. The results of these model calculations are dependent on i) the composition of the bentonite and the geochemical boundary conditions, but ii) also on the chosen modelling approach.

Model calculations were done with the help of a surface chemical thermodynamic model that simultaneously treats solution/mineral equilibria as well as protolysis and ion exchange reactions at the edge and siloxane surfaces of clay minerals. Model parameters are given in Appendix D. Acid-base reactions at the clay edge surfaces were treated using the diffuse layer model, ion exchange equilibria were calculated using the equivalent-fraction (Gaines-Thomas) formalism /Stumm and Morgan, 1996; Fletcher and Sposito, 1989/ with the help of a modification of the code MINSURF /Berner, 1993; see also Appendix D/. Thermochemical data for dissolved species and solids were all taken from /Pearson and Berner, 1991/ and /Pearson et al, 1991/. Activity coefficient corrections were done using the Davies equation.

To evaluate the influence of the chosen modelling approach on porewater composition, some calculations were carried out using an alternative thermodynamic model. The model developed for montmorillonite by /Bradbury and Baeyens, 1997/ was chosen for this purpose, as it differs significantly from that of /Wanner et al, 1994/ in how protolysis reactions at clay edge sites are treated, which is the main surface chemical factor for controlling pH. The model parameters are also given in Appendix D. In contrast to /Wanner et al, 1994/, /Bradbury and Baeyens, 1997/ did not include any electrostatic correction terms, and their model has a much larger difference between pK<sub>1</sub> and pK<sub>2</sub>.

The resulting porewater compositions are given in Appendix C. Variations in the calculated porewater compositions result from variability in (see also Figure 4-3):

- the incoming groundwater,
- whether CO<sub>2</sub> gas is allowed to leave/enter the compacted buffer; i.e. whether pCO<sub>2</sub> in the bentonite buffer is the result of groundwater-bentonite reactions (closed system) or whether pCO<sub>2</sub> in the bentonite buffer is imposed by the CO<sub>2</sub> level in the host rock (open system),
- the bentonite composition,
- In addition, it is shown that the calculated porewater composition is also a function of the modelling approach chosen.

The reference conditions for  $K_d$  derivation illustrated in Figure 4-3 were extended in an exploratory evaluation to include alkaline and hyper-alkaline groundwater. The results of these calculations are also given in Appendix C. The calculated porewater compositions show that a significant increase of porewater pH is only to be expected when no exchange of  $CO_2$  between bentonite buffer and host rock formation is considered (closed system). If carbonate equilibria in the bentonite buffer are externally controlled by the  $CO_2$  level in the host formation, the calculated porewater pH does not increase significantly. Expected effects of these changes in pH are discussed qualitatively for different elements in section 5.3.5. Further possible effects by the presence of extremely saline groundwater are briefly discussed in Appendix C.

# 5.2.6 Treatment of variable conditions in the quantification of data and uncertainties

The underlying premise of the present report is the direct use of  $K_d$  values in performance assessment consequence calculations. Because  $K_d$  is a conditional lump-sum parameter, it is critical that values are derived specifically for the conditions that are to be considered in the consequence calculations. I.e. direct use of experimental data is only possible if experimental and PA-relevant conditions are matched.

Because of the conditional nature of  $K_d$ , all variations in conditions are expressed through defined and discrete bentonite porewater compositions (see Figure 4-2 and Appendix C) in the present assessment. Because of the importance of bentonite-water interaction, this is considered as the only possible method of translating changes in conditions to effects on migration parameters, in particular  $K_d$ .

The only exception to this was the treatment of variations in redox conditions. Because of the uncertainties regarding redox conditions in bentonite porewaters, and the large effect that this parameter can have on radionuclide behaviour, the influence of pe on redoxsensitive RN was quantified by deriving migration parameter sets for each oxidation state. It has to be admitted that this approach neglects possible additional effects by pe through alterations of the porewater composition, such as changes in the sulphate/sulphide ratio, etc.

The question then needs to be asked how intermediate conditions, which are not explicitly covered by the calculated groundwater-porewater combinations should be assessed. Because of the relations shown in Figure 4-2, it is not possible to directly (without prior calculation of the resulting porewater composition) assess e.g. the effect of an "intermediate" groundwater composition on migration parameters :

- the buffering action of bentonite has to be taken into account; this is only possible through the kind of calculations used for this work,
- pH, pCO<sub>2</sub>, etc are closely related and have to be treated together.

For an approximate estimation of the possible effects of variations in the porewater composition not explicitly covered in the present report, the diagrams in section 5.3.5 can be used in an interpolative fashion. Extrapolation should be avoided; instead, the respective  $K_d$  values should be derived following the procedures used in this report.

On the other hand, the influence of buffer density on porosity and  $D_e$  can be quantified through simple relations or regression equations (see section 5.3.3 and 5.3.4), based on the present knowledge. Note that the corresponding  $K_d$  values still would have to be derived following the procedures applied in this work. An example of the influence of buffer density on  $K_d$  is shown in Figure 5-2. Note that this type of direct calculation is only possible in case of a few elements, where thermodynamic sorption models are available. However, these data suggest that moderate deviations from the present reference density should not have significant effects within the overall uncertainty.



*Figure 5-2.*  $K_d$  calculated for Sr on crude (Kunigel-V1) and purified (Kunipia-F) bentonite with the thermodynamic model given in Appendix D as a function of density (all other conditions were kept constant, input solution was pure water). Data from /Ochs et al, 2001/.

# 5.3 Data and uncertainty estimates

# 5.3.1 Instructions – issues to be addressed

Based on the assessment in chapter 4, i.e. also considering conceptual uncertainty etc and the general instructions above provide motivated uncertainty estimates of the parameter(s) for each condition defined and motivated in section 5.2 above. Depending on possibilities and assessed importance (see section 2.2.1) for the Safety Assessment, the uncertainty estimates may be given as:

- A distribution function if it can be motivated. (For example, for a spatially varying function well described by a given stochastic process, like variagraphy or DFN, a potential distribution function may be to state that all realisations of this spatially varying function are equally probable).
- Subjective percentiles  $a_i$  in the distribution function:  $P(x < a_i) = p_i$ , i.e.  $a_i$  is the parameter value where subjective probability that the parameter will take a value less than  $a_i$  is  $p_i$ . In general the percentiles corresponding to the following  $p_i$ :s: 0.01, 0.1, 0.3, 0.5, 0.7, 0.9, 0.99, should be supplied. However, sensitivity analyses (see section 2.2.1) may show that only part of the range really has an impact on the Safety Assessment. In such a case, less effort may be given to parameter values outside this range.
- A range.

The uncertainty estimates should also provide information on correlations.

- For spatially/temporal varying function information provide information about autocorrelation etc.
- List other parameter to which the parameter in question may be correlated, and where this correlation is not already taken care of by functional relations in the Safety Assessment Models. For these parameters also discuss whether a correlation function set to one (1) may be a pessimistic choice.

In answering, the expert(s) should separate between their own opinion and what they would suggest being the range of answers provided by other acknowledged scientist/experts in the field. (If their opinion is the same as all others – state this as well).

# 5.3.2 Buffer migration parameters

For a given groundwater composition the uncertainty in parameters mainly originates from laboratory measurement errors. For this it may be reasonable to assume a normal distribution. We then need to sort out the meaning of ranges supplied. Do they refer to one standard deviation or to the "final" range?

# 5.3.3 Physical porosity

Porosity  $\varepsilon$  is calculated using the following simple relationship /Sato et al, 1995; Ochs et al, 2001; Schwyn, 2003/:

$$\varepsilon = 1 - (\rho/\rho_s) \tag{5.6}$$

where  $\rho$  is the dry density and  $\rho$ s the specific density of bentonite. For  $\rho$ s, a value of 2,760 kg/m<sup>3</sup> /Schwyn, 2003/ was used. The results (Figure 5-3) show good agreement with experimental data and indicate that little variation has to be expected as a function of possible uncertainties in the bentonite composition.

For a dry density of 1,590 kg/m<sup>3</sup>, a total physical porosity of 0.43 is recommended for all elements. For all elements except Cs and non-sorbing anions, the diffusion-available porosity corresponds to this value.



**Figure 5-3.** Comparison of calculated (eq. 5.6) and experimentally determined total available porosity for MX-80. Calculations for bentonites with 45-50% smectite (Kunigel-V1) and with >95% smectite (Kunipia-F) are shown for comparison.

It is considered that the calculation of porosity with equation (5.6) gives the correct value. Therefore, no uncertainty range is given for this parameter.

The derivation of the diffusion-available porosity for anions and Cs is discussed in the following sections.

# 5.3.4 Effective diffusivity and diffusion-available porosity for anions

As pointed out previously (section 5.1.2), no RN-specific effective diffusivities are derived, due primarily to the incomplete and uncertain database. Instead,  $D_e$  for HTO is considered and recommended as being representative for all elements except anions and Cs. As discussed in the following sections on anions and Cs,  $D_e$  values of diffusing species are influenced by their charge. Thus, use of the uncharged HTO as surrogate for most radionuclides does not take into account their speciation and related ionic charge. On the other hand, most radionuclides exist as a number of species, which may include cationic, neutral and negatively charged species (see Appendix F). Therefore, in the absence of reliable species-specific  $D_e$  measurements, use of HTO is considered to be more appropriate than the use of a cation (such as an alkali element) for representing the average ionic charge and effective diffusivity of all radionuclides except Cs (and anions).

In contrast to the derivation of  $K_d$  values, which is done specifically for each condition considered, the diffusion parameters are not derived for particular sets of conditions. Instead, they are considered to be generally valid within the boundary conditions used for the present data derivation. This is done for several reasons:

- On one hand, the limited number of systematic diffusion data sets simply does not allow to derive diffusion parameters as a function of variable input conditions (except bentonite density).
- On the other hand, diffusion experiments are invariably carried out in compacted bentonite, which implies that the porewater conditions are largely determined by the bentonite itself in most cases. Therefore, porewater conditions in a given compacted bentonite show comparatively little variation as a function of input solution. This clearly contrasts batch experiments, where typically a (very) low bentonite/water ratio is used and where, as a consequence, the solution composition is controlled by the input solution.
- Further, the parameters selected for the present report ( $D_e$ ,  $\epsilon$  for anions and HTO) are typically derived from through-diffusion experiments under steady-state conditions, where the influence of porewater conditions on sorption is not relevant.

# нто

 $D_e$  values obtained for HTO in different bentonites are plotted in Figure 5-4 as a function of dry density. Data for Kunigel-V1 and MX-80 are in quite close agreement and can be quantified as follows by a regression equation ( $r^2 = 0.94$ ):

$$D_{e} = 6.7785E - 9 \times e^{-2.5671E - 3 \times \rho}$$
(5.7)

For a dry density of 1,590 kg/m<sup>3</sup>, a D<sub>e</sub> for HTO of  $1.2 \times 10^{-10} \pm 7.06 \times 10^{-11}$  m<sup>2</sup>/s is recommended (25°C).



Figure 5-4. Plot of effective diffusivities for Cs, HTO and various anions as a function of dry density. All data labelled \* are for Kunigel-V1 bentonite /from Sato, 1998a/. The reference density of 1,590 kg/m<sup>3</sup> and uncertainty of  $D_e$  for HTO is indicated. For the Cl diffusion data in MX-80 /Muurinen et al, 1989/, the decrease of  $D_e$  with decreasing ionic strength of the input solution (1 $\rightarrow$ 0.01 M) is indicated. The model calculations by /Ochs et al, 2001/ correspond to pure water as input solution; note that this leads to a calculated ionic strength of ca 0.2–0.4 M at 1,590 kg/m<sup>3</sup>. The range calculated for I following /Yu and Neretnieks, 1997/ is based on their estimation of  $D_e = 3 \times 10^{-12} \text{ m}^2/\text{s}$  for 1,600–2,000 kg/m<sup>3</sup>. This  $D_e$  value at the low/high end of the indicated density was taken as minimum/maximum, and the same slope of  $D_e$  vs density as for HTO and the model calculations was assumed.

The above value is in close agreement with the value of  $1.0 \times 10^{-10}$  m<sup>2</sup>/s recommended by /Yu and Neretnieks, 1997/. The corresponding uncertainty represents the statistical standard error for the estimation of D<sub>e</sub> by equation (5.7) based on the experimental data in Figure 5-4. To take into account a temperature increase from ambient conditions to max. 50°C, /Schwyn, 2003/ doubles the selected value; i.e. in this case a value of  $2.4 \times 10^{-10}$  m<sup>2</sup>/s would result.

# Anions

D<sub>e</sub> and D<sub>a</sub> values for anions show a significant scatter, especially in the case of MX-80 (Figure 5-4 and Figure 5-6), reflecting the fact that relatively many investigators measured data, sometimes for a range of conditions, but only for a limited range of dry densities. In particular, few systematic data are available for solutions of intermediate-high salinity (ionic strength > 0.3 M). On the other hand, diffusivities reported for low ionic strengths refer in all likelihood actually to ionic strength values of 0.2–0.3 M, as it is not considered possible to have a lower ionic strength in the pore solution of compacted bentonite (see section 5.1.3). While using slightly different approaches to porewater modelling, /Bradbury and Baevens, 2003c/ and /Wersin, 2003/ also conclude that the large buffering capacity of bentonite and the dissolution of soluble impurities and accessory minerals invariably leads to pore solutions of intermediate ionic strength. /Ochs et al, 2004/ show that even in the case that the amount of soluble impurities would be significantly decreased, the pore solution in compacted bentonite still would have an ionic strength of > 0.1 M. Part of the experimental scatter is probably attributable to experimental difficulties /cf Yu and Neretnieks, 1997/, and in case of  $D_e$  possibly also to difficulties in assigning  $D_e$  and  $\varepsilon$  values for interpreting the measured raw data /see e.g. van Loon et al, 2003a,b/. Overall, it is estimated that the uncertainty in experimental diffusion coefficients is at least half an order of magnitude.

Despite the scattering of experimental data, it can clearly be seen that anions have significantly lower effective diffusivities than HTO. This is corroborated by model predictions for the diffusion of monovalent anions for MX-80 and Kunipia-F shown in Figure 5-4 (integrated sorption-diffusion model ISD, /Ochs et al, 2001/). These predictions are made by electric double layer models and do not take into account any reduction in available anion porosity. Note that these models had been calibrated based on data for Cs on Kunipia-F bentonite; they were then applied to anion diffusion and other bentonites in a purely predictive way.

The abovementioned problem of distinguishing between  $D_e$  and  $\epsilon$  (more specifically, between  $D_e$  and the capacity factor  $\alpha = \epsilon + K_d \rho$ ) makes the selection of a self-consistent set of  $D_e$  and  $\epsilon$  values somewhat ambiguous. A consistency check using corresponding  $D_a$  values is of limited use, because of the relation:

$$D_{a} = D_{e} / (\varepsilon + K_{d}\rho) \text{ or } D_{a} = D_{e} / \alpha$$
(5.8)

Therefore, a given measured value for  $D_a$  can be explained by any combination of  $D_e$  and  $\alpha$  (or  $D_e$  and  $\epsilon$  if  $K_d = 0$  is assumed). This is illustrated in Figure 5-5.

Based on this situation, the following approach was followed: First,  $D_e$  and the corresponding uncertainty was evaluated independently of any considerations regarding  $\epsilon$  or  $\alpha$ . In a second step, these values were related to corresponding values for  $\epsilon$  and  $\alpha$ , and a final selection was made.



**Figure 5-5.** Illustration of calculated  $D_a$  values resulting from different combinations of  $D_e$ and  $\varepsilon$  (or  $D_e$  and  $\alpha$ , if a  $K_d \neq 0$  is assumed). The green band indicates the range of expected  $D_a$ values (Figure 5-6).  $D_e$  HTO = 1.2E–10 m<sup>2</sup>/s (see above),  $D_e$  anions ISD = 1.0E–11 m<sup>2</sup>/s, full porosity = 0.43.

## **Effective diffusivity**

 $D_e$  was selected from Figure 5-4, in a first step. The two calculations by /Ochs et al, 2001/, representing completely independent estimates, agree with the bulk of the experimental data i) in terms of trend of  $D_e$  vs bentonite density as well as ii) in terms of magnitude. At the same time, the model calculation for MX-80 is in reasonably good agreement with the experimental data for Cl and I at intermediate to high salinity (0.22–1.0 M input solutions), which corresponds fairly closely to the upper limit of the range calculated based on the recommendation of /Yu and Neretnieks, 1997/.

Note that the model prediction for MX-80 by /Ochs et al, 2001/ corresponds to an ionic strength of ca 0.25 M in the pore solution at dry densities > ca 1,600 kg/m<sup>3</sup>, which we consider to be minimum values at this buffer density. Some data in Figure 5-4 suggest that very dilute groundwaters lead to lower diffusivities, but the database is not sufficient to allow that conclusion. In particular, it is not clear that equilibration of the input solution with the bentonite was sufficient: This process should lead to an ionic strength of ca 0.2–0.3 M, and all D<sub>e</sub> values for dilute input solutions should start to approach each other; this is actually indicated by the data obtained at 1,800 kg/m<sup>3</sup> by /Muurinen et al, 1989/. Ionic strength effects diffusion of anions via its influence on electric double layer properties. According to the findings of /Kozaki et al, 1998/ and /Ochs et al, 2004/, this is only relevant in sufficiently narrow pores; i.e. above a dry density of ca 1,200–1,400 kg/m<sup>3</sup>.

Based on the above discussion, a  $D_e$  for anions of  $1.0 \times 10^{-11}$  m<sup>2</sup>/s is recommended for a dry density of 1,590 kg/m<sup>3</sup> (25°C). This value is considered to be a realistic estimate for groundwaters of intermediate to high salinity (ca 0.2 to 1 M). Based on  $D_e$  values for dry densities > 1,800 kg/m<sup>3</sup>, it appears that dilute input groundwaters lead to  $D_e$  values that are significantly lower, but the available database does not allow to draw this conclusion with certainty.

Upper and lower limits for  $D_e$  were evaluated based on the following observations (Figure 5-4):

- As lower limit for  $D_e$ , the minimum value of  $3.0 \times 10^{-12}$  m<sup>2</sup>/s at 1,600 kg/m<sup>3</sup> estimated by /Yu and Neretnieks, 1997/ is accepted for a density of 1,590 kg/m<sup>3</sup>. This value is in close agreement with the data obtained by /Muurinen et al, 1989/ for Cl at intermediate densities, using dilute as well as pre-equilibrated input solutions.
- As upper limit, a value of  $3.0 \times 10^{-11}$  m<sup>2</sup>/s at 1,600 kg/m<sup>3</sup> is taken. This is based on measured D<sub>e</sub> values for several anions in compacted Kunigel-V1, as well as on the model prediction by /Ochs et al, 2001/ for Kunipia-F.

Figure 5-4 shows that the band corresponding to these limits encompasses the majority of the  $D_e$  values plotted.

#### Diffusion-available porosity, and comparison with D<sub>a</sub>

Diffusion-available anion porosity is selected based on the findings of /van Loon et al, 2003a,b/, who made an explicit effort to distinguish  $D_e$  from  $\epsilon$  in their data interpretation. Their data for chloride suggest a reduction of the diffusion-available porosity from HTO to anions by a factor of 1.8–3.5 (if a K<sub>d</sub> of 0 is assumed). Based on this finding, a reduction factor of 2.5 is proposed, resulting in an anion diffusion-available porosity of 0.17 for the present case. Use of the reduction factors 1.8 and 3.5 results in porosities of 0.24 and 0.12 as limiting values. Figure 5-5 and Figure 5-6 show that this reduction in porosity is consistent with independent  $D_a$  measurements when used together with the selected  $D_e$  for anions given above. It is also illustrated that use of the limiting values. On the other hand, use of the full physical porosity leads to calculated  $D_a$  values that underpredict most of the measured data.

Figure 5-6 also shows the effect of the uncertainties in  $D_e$  proposed above with respect to  $D_a$ . It can be seen that the resulting minimum and maximum  $D_a$  values actually encompass the entire range of measured data at the dry density of 1,590 kg/m<sup>3</sup>.

#### Selected values and uncertainties for anion diffusion parameters

Based on the introductory discussion to this section, it has to be made clear that the selected anion diffusion parameters have to be evaluated together. Figure 5-6 shows that the proposed values represent a self-consistent data set that is also consistent with independently measured  $D_a$  values. It is therefore evaluated as extremely likely that any combined dataset of  $D_e$  and  $\epsilon$  (and  $K_d$ ) under the specified scenario (as expressed by the resulting  $D_a$ ) would be within the indicated ranges.

For non-sorbing anions ( $K_d = 0$ ), it is recommended to use at a dry density of 1,590 kg/m<sup>3</sup>

- a diffusion-available anion porosity of 0.17, with upper and lower limits of 0.24 and 0.12, respectively,
- a  $D_e$  of  $1.0 \times 10^{-11}$  m<sup>2</sup>/s, with upper and lower limits of  $3.0 \times 10^{-11}$  m<sup>2</sup>/s and  $3.0 \times 10^{-12}$  m<sup>2</sup>/s, respectively.

The diffusion-available anion porosity can be related to dry density using equation (5.6) and a constant reduction factor of 2.5 (equation 5.9).

$$\varepsilon_{anion} = (1 - (\rho/\rho_s))/2.5$$
 (5.9)



**Figure 5-6.** Apparent diffusivities for different anions in MX-80 and Kunipia-F bentonite in comparison to model predictions:  $D_a$  was calculated using ISD-model [1] derived  $D_e$  values and the corresponding uncertainty limits in combination with different choices for anion porosity:

- 1) the entire physical porosity; this corresponds directly to the prediction for Cl<sup>-</sup> in MX-80 shown in Figure 5-4.
- 2) maximum diffusion-available anion porosity (DAP) based on the data of [2] for chloride; and HTO
- *3) minimum diffusion-available anion porosity (DAP) based on the data of [3] for chloride; and HTO*
- [1] /Ochs et al, 2001/; [2] /van Loon et al, 2003a/; [3] /van Loon et al, 2003b/

The De for anions as a function of density is given by the line in Figure 5-4 representing the model prediction by /Ochs et al, 2001/. To avoid the use of the respective underlying electrical double layer model, a  $3^{rd}$  degree polynomial function can be used which reproduces the model predictions as a function of density for a density range of ca 1,000–2,000 kg/m<sup>3</sup> exactly (i.e. follows the slight curve of the model predictions resulting from changes in the porewater chemistry as a function of density;  $r^2 = 0.998$ ):

$$D_{e} = -1.1844 \times 10^{-20} \times \rho^{3} + 8.130 \times 10^{-17} \times \rho^{2} - 1.9473 \times 10^{-13} \times \rho + 1.6167 \times 10^{-10}$$
(5.10)

An approximate representation of the model predictions can also be given by a logarithmic relation similar to eq. (5,7):  $D_e = 5.30087E - 10 \times e^{-2.561E - 3 \times \rho}$  ( $r^2 = 0.918$ ).

# Cs

Following the same arguments used to explain anion exclusion (i.e. interaction of negatively charged diffusant with electrical double layer extending from negatively charged pore walls), /Kato et al, 1995; Sato et al, 1995; Ochs et al, 2001/ used an electric double layer model to explain the apparently enhanced diffusion of certain cations (see section on diffusion in the SR-Can Process Report /SKB, 2004/). This model is able to explain D<sub>a</sub> values for Cs obtained for different bentonites and densities, without having to rely on a surface diffusion coefficient. Some model results are reproduced in Figure 5-11.

Based on these studies, and on consistency of  $D_e$  and  $D_a$  values determined under comparable conditions, a  $D_e$  value of  $3 \times 10^{-10}$  m<sup>2</sup>/s is recommended for Cs.

This value also serves as upper limit; the lower limit is given by the lower limit for HTO.  $D_e$  for Cs is considered to show the same dependency on density as  $D_e$  for HTO, differing by a constant factor of 3 (at most, see above).

Note that the above considerations are in all likelihood not relevant for any other of the "cationic" elements, as they exist largely as a mix of positively, negatively and uncharged species (see also /Ochs et al, 2003/).

# 5.3.5 Derivation of $K_d$ values, comparison with diffusion data

In the following sections, the derivation of  $K_d$  is discussed for each element, and the consistency of the selected data is evaluated through the calculation of  $D_a$  values and comparisons with independent experimental data. The selected source data as well as the results obtained after conversion to application conditions are tabulated in detail in Appendix E, including all conditions, conversion factors, uncertainties, resulting recommended values and upper/lower limits. The  $D_a$  values used for comparisons were preferably taken from studies where diffusion had been studied as a function of density (see section 2.3.1).

Based on a large number of calculated porewater compositions (Appendix C), three representative porewaters were selected that spanned a wide range of the possible conditions, including pH, ligand concentrations (in particular carbonate) and salinity. This allowed to assess uncertainty as a result of the variability in conditions using these three porewaters in most cases. The selected porewaters are (Appendix C).

- the reference porewater (RPW), based on the saline reference Beberg groundwater and a pCO<sub>2</sub> imposed by the host formation according to /Laaksoharju et al, 1998/,
- a water defined as RPW above, but treating bentonite as a closed system with respect toCO<sub>2</sub> (RPWC),
- a porewater based on highly saline groundwater (HSPW).
- Qualitatively, an increase of pH outside the reference range is also addressed where relevant; this is restricted to the effect of pH per se and does not include further potential concomitant changes in dissolved carbonate levels.

#### Americium

 $K_{d}$  values were derived based on the study of /Gorgeon, 1994/, as it is the only available study, to our knowledge, where Am sorption on smectite was studied systematically as a function of pH: /Gorgeon, 1994/ determined two adsorption edges from pH 3-11 in 0.1 and 1 M NaClO<sub>4</sub> solutions on smectite extracted from Wyoming bentonite. Based on speciation calculations (Appendix F), Am was below the solubility limit in all experiments. A shortcoming of this study is that  $pCO_2$  had not been controlled rigorously, but at least an estimate of total dissolved carbonate is given in /Gorgeon, 1994; Ochs et al, 1998/ also had used the data of /Gorgeon, 1994/ as the most systematic dataset available for the development of a thermodynamic sorption model. While their proposed model is probably too simple<sup>1</sup>, it clearly shows that Am sorption on smectite in the pH range 3-10 can be described by surface complexation and ion exchange (the latter becoming important only at near-neutral to low pH values in solutions of low ionic strength). This is also consistent with the model developed by /Bradbury and Baeyens, 2002a/ for the sorption of Eu on montmorillonite. /Shibutani et al, 1994/ also measured Am sorption as a function of pH on Kunigel-V1 bentonite. However, all their data points correspond to nearly 100% sorption, which can cause large errors and renders them less reliable.

Based on the above discussion, K<sub>d</sub> values for Am were derived from the data of /Gorgeon, 1994/ using the CF-approach. It is believed that the model by /Ochs et al, 1998/ would not be superior to the CF-approach, because it is probably too simple (see above) and because it is based on the same dataset and does therefore not contain any additional chemical information The resulting  $K_d$  values and uncertainties are plotted as a function of conditions in Figure 5-7. The predicted increase of K<sub>d</sub> with pH is consistent with a surface complexation mechanism (i.e. corresponds to the rising slope in Figure 5-1). Except for RPWC, the results obtained with CF-total 1-3 agree very well. The dominant dissolved Am species is  $AmCO_3^+$  (Appendix F), and the effect of full corrections for solution speciation (CF-total 3) is therefore directly related to the differences in carbonate concentration between data source and application conditions. As no systematic sorption data under variable carbonate concentration are available. the effect of carbonate on Am sorption cannot be predicted with certainty. For example, it cannot be excluded that AmCO<sub>3</sub><sup>+</sup> is involved in ion exchange, as postulated by /Shibutani et al, 1994/, which would contribute to sorption especially in the RPWC system. Therefore, the selected data are based on CF-total 3.

Overall, the uncertainties of the selected data (using CF-total 3) also encompass the values that would result from treating the scaling of speciation in different ways (i.e. using CF-total 1 or 2) even in case of RPWC. It is, therefore, extremely likely that that any  $K_d$  for Am under the specified scenario would be within the indicated ranges.

Calculated  $D_a$  values based on  $D_e$  for HTO and the selected  $K_d$  values are compared with experimental data in Figure 5-7. It is shown that the calculated  $D_a$  values are very consistent with measured data, and that the uncertainties for  $K_d$  cover well the scatter observed for  $D_a$ , corroborating the robustness of the indicated data and ranges. Am speciation was calculated using the TDB of /Silva et al, 1995/. The influence of updated TDB was evaluated using the data given in /Guillaumont et al, 2003/. Values calculated with these two TDBs are identical and agree within a factor 2 for RPW, RPWC, and HSPW.

<sup>&</sup>lt;sup>1</sup> At the time of publication, the aim was to use a model consistent with the basic bentonite model by /Wanner et al, 1994/, which treats surface complexation by a one-site diffuse layer model. Subsequent preliminary work (unpublished) showed that a two-site model would allow significantly better fits at near-neutral pH.



*Figure 5-7.*  $K_d$  values for Am derived as a function of conditions. The selected values are indicated by solid symbols (CF-total 3); the error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4. Am speciation is calculated using the data by /Silva et al, 1995/.



**Figure 5-8.** Calculated  $D_a$  values for Am based on  $D_e$  for HTO and selected  $K_d$  derived as a function of conditions. The data by Sato and co-workers were obtained with pure water as input solution, SR97 refers to the  $D_a$  value selected by /Yu and Neretnieks, 1997/.

The data by /Gorgeon, 1994/ suggest that  $K_d$  may increase up to a porewater pH of ca 11. However, at more elevated carbonate concentrations,  $K_d$  may start to decrease above a pH of about 8–9, due to increased competition through complexation with dissolved carbonate (formation of negatively charged Am-carbonato complexes). For even higher pH values, no reliable data are available.

#### Caesium

A large number of sorption studies on bentonite is available for Cs (not discussed in detail here; see e.g. the compilation of /Yu and Neretnieks, 1997/, as well as several ion exchange studies on pure clay minerals. While the data obtained on bentonite are more directly applicable to the present purpose, most studies involving bentonite are not systematic enough to allow data transfer to different conditions. It is well established that Cs sorbs via cation exchange /e.g. Wanner et al, 1996/, whereas the contribution of surface complexation is negligible. This renders sorption of Cs very sensitive with respect to changes in the major cation composition of the aqueous phase.

The study of /Wanner et al, 1996/ and the measurements reported by /Bradbury and Baeyens, 2003a/ were considered the most systematic data sources directly applicable to MX-80 and were selected for the present data derivation. Both studies used untreated MX-80 bentonite. /Wanner et al, 1996/ report a number of K<sub>d</sub> values measured in various 1:1 and 2:1 electrolyte solutions, as well as an isotherm obtained in 0.1 M NaCl. Most of their data correspond to solid/water ratios  $\leq 0.1$  g/L. /Bradbury and Baeyens, 2003a/ report a detailed sorption isotherm on MX-80 in synthetic bentonite porewater with an ionic strength of 0.7 M, using a solid/water ratio of 60 g/L. /Hurel et al, 2002/ also give more limited isotherms for MX-80 (4 g/L) corresponding to pH 7.4 and 8.2 in a dilute synthetic groundwater (I = 0,05 M). The isotherms from these three studies are nearly parallel over 5–6 orders of magnitude in terms of Cs concentration, but cover about one order of magnitude in terms of K<sub>d</sub>. This is consistent with sorption edges by /Hurel et al, 2002/ using purified and un-purified MX-80 and can be explained with the difference in ionic strength (i.e. the concentration of competing cations) between the two studies.

Using the data discussed above, /Wanner et al, 1996/ developed a thermodynamic sorption model based on a reversible one-site ion exchange of Cs. They also applied this model successfully to some literature data. For the present purpose, their ion exchange constant for Cs was incorporated in the model used to calculate bentonite-groundwater interaction; see Appendix D for details.  $K_d$  values calculated for the present reference systems are shown in Figure 5-9 as a function of pH and ionic strength. The model calculations are compared with values derived using the CF approach based on the experimental data by /Bradbury and Baeyens, 2003a/. Because of the predominance of ion exchange for Cs sorption, the pH-conversion factor was replaced in this case by a CF-cation conversion (see Appendix E), with a UF-cation conversion of 2.5. Because of the sensitivity of Cs sorption to major cation concentrations, the model calculations were extended to cover additional application conditions (Figure 5-10, see also Appendix C). It can be seen that uncertainties introduced by considering further variability in the conditions (loss of impurities, conversion to Ca-bentonite, different groundwater composition) leads to calculated  $K_d$  values that fall well within the uncertainty limits indicated in Figure 5-9.

Both Figures show that the model calculations and the values derived from the data of /Bradbury and Baeyens, 2003a/ agree very well, although the values derived with the CF-approach show a less pronounced dependency on ionic strength. The reason for this may be that the CF-approach does not include effects of cation selectivity. However, it also should be noted that the data underlying the model of /Wanner et al, 1996/ correspond to experiments in low-ionic strength solutions, whereas the data of /Bradbury and Baeyens, 2003a/ were determined at high ionic strength. Thus, the slightly different trend of  $K_d$  vs ionic strength observed with the tow approaches may also reflect the different starting conditions for model development and data derivation.

In summary, it is considered to be extremely likely that any  $K_d$  for Cs under the specified scenarios would be within the indicated ranges, based on the relatively good consistency among the calculated and derived  $K_d$  values .



**Figure 5-9.**  $K_d$  values for Cs calculated as a function of conditions with the Cs sorption model of /Wanner et al, 1996/. Values derived based on the experimental data by /Bradbury and Baeyens, 2003a/ are shown for comparison. The selected values are indicated by open circles (model); the error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.

In Figure 5-11, this assessment is corroborated by comparing  $D_a$  values calculated using the K<sub>d</sub> values derived based on the data by /Bradbury and Baeyens, 2003a/ as well as the values calculated with the sorption model of /Wanner et al, 1996/ together with the diffusion parameters selected for Cs. The agreement of the calculated  $D_a$  values with experimental data and diffusion model predictions is excellent, and the uncertainties derived for K<sub>d</sub> accommodate well the variation of the experimental diffusion data.

For Cs, a  $D_e$  value of  $3E-10 \text{ m}^2/\text{s}$  is recommended (see section 5.3.4). Figure 5-11 shows that this choice leads to an overall consistent set of migration parameters for Cs.



**Figure 5-10.**  $K_d$  values for Cs calculated using the Cs sorption model of /Wanner et al, 1996/. The values derived using the CF approach (CF-total 2) based on the experimental data by /Bradbury and Baeyens, 2003a/ are shown for comparison.



**Figure 5-11.** Calculated  $D_a$  values for Cs based on  $D_e$  for Cs and selected  $K_d$  calculated using the model of /Wanner et al, 1996/ or derived as a function of conditions with the CF approach. Experimental data obtained on Kunigel-V1 are shown as circles, data obtained on MX-80 as crosses. Sorption-diffusion model calculations /Ochs et al, 2001/ and the result of a regression analysis are also shown. The error bar spans the entire uncertainty in  $K_d$  for RPW, RPWC and HSPW.

Based on the major cation concentration in the calculated porewater compositions (Appendix C), the influence of (highly) alkaline groundwater on  $K_d$  for Cs is expected to be within the limits given by the three reference porewaters.

# Carbon

Carbon can exist in various forms, depending on the origin of the carbon and the relevant redox conditions: Considered here are carbonate species, simple organic acids, or methane.

 $D_e$  and  $\epsilon$  for carbonate species are treated analogously to the values selected for chloride, even though the carbonate ion has a charge of –II and may also sorb weakly. The data in Figure 5-4 do not point to a significant difference between carbonate species and other anions, however. Further, considering the inventory of stable carbonate in the bentonite (as calcite), it is most likely that removal of <sup>14</sup>C from the solution takes place through isotopic exchange with calcite. K<sub>d</sub> values can then be readily calculated based on the respective solution chemistry and the amount of accessible calcite in MX-80.

In case of methane, a  $K_d$  of zero and the  $D_e$  as well as porosity used for HTO (see Figure 5-4) are proposed.

In case of simple organic acids, no reliable  $K_d$  values were found in the literature. Model calculations for acetic acid, using the approach by /Ochs and Talerico, 2003/, indicated that  $K_d$  on MX-80 is < 1E-7 m<sup>3</sup>/kg. The reason for this is mainly that silicate competes effectively for the same type of sites (edge sites). Therefore, a  $K_d$  of zero and the  $D_e$  as well as porosity derived for anions are proposed. It is realised that slightly more complex organic molecules, featuring e.g. 2 carboxyl groups instead of just one (as in case of acetic acid), may exhibit significantly stronger sorption on clay edges, based on analogy with Al-oxide surfaces /e.g. Kummert and Stumm, 1980/. As the nature of the organic molecules is not known, however, this possibility is not considered. As the use of a simple organic acid is considered to be a conservative approach, no additional UF is introduced.

# Cerium, holmium, and samarium

No reliable and relevant sorption data could be found for any of these elements. Therefore,  $K_d$  values were derived using Eu as analogue element. Eu sorption on smectite has been investigated in detail by Bradbury and Baeyens. /Bradbury and Baeyens, 2002a/ report sorption edge as well as isotherm data obtained on Na- as well as Ca-montmorillonite (SWy-1) in NaClO<sub>4</sub> and Ca(NO<sub>3</sub>)<sub>2</sub> background electrolytes. These data form the basis for a thermodynamic sorption model (two-site surface complexation without EDL term, plus ion exchange) developed by the same authors. In addition, /Bradbury and Baeyens, 2003a/ report an isotherm for pH 7,6 on MX-80 in artificial porewater (I = 07 M).

K<sub>d</sub> values derived by the CF-approach using MX-80 source data are given in Figure 5-12, together with values calculated using the thermodynamic sorption model of /Bradbury and Baeyens, 2002a/. Data derived on the basis of SWy-1 by the CF-total 2 and CF-total 3 methods are also shown and bracket the model data within about an order of magnitude, as should be expected (Appendix E). While the sorption model has the better scientific foundation, it has to be realised that robustness in terms of extrapolation to different conditions depends to a large degree on model parameterisation. Unfortunately, the effects of practically all major anions (including carbonate) are not included in the dataset for SWy-1 (and therefore in the corresponding model). In that sense, it is encouraging that the difference between the model based on the simplified SWy-1/electrolyte system and the CF-approach based on the more complex MX-80/porewater system is less than an order of magnitude.



**Figure 5-12.**  $K_d$  values for Eu derived as a function of conditions based on the experimental data by /Bradbury and Baeyens, 2003a/ for MX-80. Values calculated using the Eu sorption model of /Bradbury and Baeyens, 2002a/ are also shown. The selected values are indicated by solid symbols (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.

Based on a comparison with the available  $D_a$  measurements (Figure 5-13), it is recommended to use the values derived from the MX-80 source data; i.e. to rely on source data that are comparatively close to the application conditions, rather than on the sorption model. However, it is recommended to use the values obtained in the model calculations as the upper limit of the recommended data, while the lower limit should be based on the MX-80 source data. These limits are very likely to encompass any K<sub>d</sub> for trivalent lanthanides under the specified scenarios. No additional uncertainty factors are proposed to account for the use of Eu as analogue element.

No reliable data for (highly) alkaline conditions are available, but the experimental data by /Bradbury and Baeyens, 2002a/ clearly show an increasing trend of  $K_d$  for Eu up to pH 9, which is consistent with the data for Am.

#### Chloride

Both  $D_e$  and  $\epsilon$  selected for anions are proposed, together with a  $K_d$  of zero. No additional uncertainties are considered.

#### Curium

No reliable, systematic sorption data for Cm on relevant minerals could be found. The chemistry of Cm appears to be nearly identical to that of Am, but no TDB of comparable quality is available. Therefore, Am is used directly as analogue element; no additional UF-analogue is considered for this case.



**Figure 5-13.** Calculated  $D_a$  values for Eu(III) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-12 in comparison to experimental data for Sm obtained on Kunigel-V1. For comparison, the regression line representing Am (from Figure 5-8) is also shown.

## Holmium

See section on cerium, holmium and samarium.

## lodine

It is considered that iodine will exist exclusively as iodide anion under relevant conditions. There is circumstantial evidence that iodide may exhibit very low, but notable sorption; see e.g. the compilation of /Liu and von Gunten, 1988; Yu and Neretnieks, 1997/. However, no unambiguous data could be found. Further, it may be that iodide sorption is more related to illite than smectite minerals /Kaplan et al, 2000/. Therefore, iodide is treated analogously to chloride: A  $K_d$  of zero is proposed together with  $D_e$  and  $\varepsilon$  selected for anions.

# Lead

Several studies on Pb sorption on different montmorillonites are available, but no reliable diffusion data were found. An overview of Pb sorption data is e.g. given in /Ochs et al, 1998/, who also showed that the different studies are approximately consistent with each other, using a simple sorption model (not considered for the present purpose because of its preliminary nature). /Ulrich and Degueldre, 1993/ worked with trace concentrations of Pb throughout, and their study is selected as most representative. K<sub>d</sub> was derived using the CF-approach, and the resulting values under application conditions are shown in Figure 5-14. This figure shows that scaling to application conditions is not very sensitive with respect to CF-speciation. It is considered to be very likely that any K<sub>d</sub> for Pb under the specified scenarios would be within the indicated ranges, despite the lack of diffusion data for comparison.

No reliable data for (highly) alkaline conditions are available. Based on the expected solution speciation of Pb, it cannot be excluded that sorption may decrease at pH > 9 or so.



**Figure 5-14.**  $K_d$  values for Pb derived as a function of conditions based on the experimental data by /Ulrich and Degueldre, 1993/. The selected values are indicated by solid symbols (CF-total 2); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.

#### Neptunium

In relevant aqueous solutions, stable oxidation states of Np can be IV or V, depending on redox conditions. Because of the uncertainty regarding redox conditions, both oxidation states are treated separately below, following the approach for uranium. The overall uncertainty for the  $K_d$  of Np encompasses the entire combined data range if variable redox conditions need to be taken into account.

For Np(IV), no relevant sorption data were found in the open literature, and  $K_d$  was derived based on analogy considerations and the source data set used for Th. The results are shown in Figure 5-15, the comparison with Da values is shown in Figure 5-16. As in the case of U, the data show that inclusion of carbonate species in the speciation-conversion leads to results that are completely inconsistent with diffusion data for Th as well as Np. Therefore, the  $K_d$  values based on CF-total 3 are recommended. It can be seen from Figure 5-16 that the selected sorption data are not inconsistent with the available  $D_a$  values for Th and those for Np obtained under the most reducing conditions.

As for U(IV), it is considered likely that most of the diffusion experiments may not have been controlled well enough to avoid the presence of Np(V), and the overall diffusion may represent a mix of reduced and oxidised Np fractions. It is considered to be likely that any  $K_d$  for Np(IV) under the specified scenarios would be within the indicated ranges.

Since the hydrolysis behaviour of Np(IV) appears to be identical to that of Th at high pH /Lemire et al, 2001/, no significant influence of a pH increase on  $K_d$  is expected.



**Figure 5-15.**  $K_d$  values for Np(IV) derived as a function of conditions based on the experimental data for Th by /Bradbury and Baeyens, 2003a/. The selected values were derived based on CF-total 3; error bars are given for these values and for the values based on CF-total 2. For an explanation of CF-total 1/2/3, see section 5.1.4.



**Figure 5-16.** Calculated  $D_a$  values for Np(IV) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-15 in comparison to experimental data (experiments where a reducing agent had been used are depicted by open symbols). For comparison, experimental  $D_a$  values for Th are also shown.

The most systematic and complete data set for the sorption of Np(V) on smectite available in the open literature has been published by /Turner et al, 1998; Bertetti et al, 1998/. They determined Np(V) sorption on SAz-1 Na-montmorillonite as a function of pH and pCO<sub>2</sub> in 0.1 M NaNO<sub>3</sub>, using both adsorption and desorption experiments. These data are also consistent with the work of /Gorgeon, 1994/, who determined two adsorption edges from pH 3-11 in 0.1 and 1 M NaClO<sub>4</sub> solutions on smectite extracted from Wyoming bentonite. These datasets had also been chosen for the recent NEA Sorption Project modelling exercise /NEA, 2004/, but no reliable and well-tested sorption models are available as yet. However, this exercise showed that the inclusion of ternary Np-carbonato surface complexes improves a model's ability to describe the experimental data at elevated pH and in the presence of carbonate. This supports the choice of CF-total 3 (which does not treat dissolved carbonate complexes as competitive with respect to surface complexation) as conversion procedure.

The  $K_d$  values derived based on the data by /Turner et al, 1998; Bertetti et al, 1998/ using the CF-approach are shown in Figure 5-17. Because of the similarity of conditions, the speciation-conversion factor has only little influence; i.e. CF-total 1–3 give nearly identical results. The comparison with independent  $D_a$  values is given in Figure 5-18. The calculated data using CF-total 3 are not inconsistent with the experimental diffusion data, and it is considered to be very likely that any  $K_d$  for Np(V) under the specified conditions would be within the indicated ranges.

The data by /Turner et al, 1998/ indicate that the sorption of Np(V) on montmorillonite increases up to pH  $\approx$  10.5 in the absence of significant amounts of dissolved carbonate. At atmospheric pCO<sub>2</sub>, sorption starts to decrease at pH  $\approx$  8.5.



**Figure 5-17.**  $K_d$  values for Np(V) derived as a function of conditions based on the experimental data by /Turner et al, 1998/. The selected values are based on CF-total 3; error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.



*Figure 5-18.* Calculated  $D_a$  values for Np(V) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-17 in comparison to experimental data (experiments where a reducing agent had been used are depicted by open symbols).

## Nickel

The most extensive and systematic sorption data set available for Ni is given by /Baeyens and Bradbury, 1997/, who determined Ni sorption on Na-montmorillonite (SWy-1) in NaClO<sub>4</sub> solutions as a function of pH, ionic strength and Ni concentration. /Bradbury and Baeyens, 1997/ modelled these data using a surface complexation/ion exchange model In a follow-up study /Bradbury and Baeyens, 1999/, the sorption of Ni on Camontmorillonite (SWy-1) was also investigated and modelled. It was concluded that the model parameters for the Na- and Ca-form are identical within uncertainties. In addition, /Bradbury and Baeyens, 2003a/ determined a sorption isotherm for Ni on MX-80 in artificial bentonite porewater at pH 7.6.

For the present purpose,  $K_d$  values were calculated using the thermodynamic sorption model by /Bradbury and Baeyens, 1997/. The results were compared with values derived with the CF-approach using the sorption data on Na-SWy-1 and MX-80. The resulting  $K_d$  values are summarised in Figure 5-19. It can be seen that the choice of data source appears to be important. Both the model application and the CF-approach using the SWy-1 source data result in higher predictions of  $K_d$  in comparison to the CF-approach using the MX-80 source data. The exact reason is not clear at the moment, but the results shown in Figure 5-19 suggest very strongly that the observed discrepancy is indeed related to the data source used for upscaling to application conditions rather than the actual method employed (sorption model or CF-approach). The sorption model had been calibrated based on the SWy-1 data, and application of this model vs application of the CF-approach to the SWy-1 data



**Figure 5-19.**  $K_d$  values for Ni derived as a function of conditions based on the experimental data for SWy-1 montmorillonite and MX-80 bentonite by /Bradbury and Baeyens, 1997, 2003a/ and calculated using the Ni sorption model of /Bradbury and Baeyens, 1997/. The selected values are indicated by solid symbols. Error bars are given for the values derived from data on SWy-1(brown) and MX-80 (black) using CF-total 2. For an explanation of CF-total 1/2/3, see section 5.1.4.

give nearly identical results. In comparison, upscaling based on MX-80 source data gives  $K_d$  values that are about an order of magnitude lower. It should be further noted that the speciation-conversion has very little influence on upscaling from any given data source (CF-total 2 and CF-total 3 give practically identical results, because under the specified conditions Ni-carbonato complexes are not important in comparison to complexation of Ni with sulphate and chloride according to the thermodynamic data given in /Hummel et al, 2002/).

Based on this evidence, the  $K_d$  values derived from the MX-80 source data using the CF-approach are preferred over values derived from SWy-1 source data either directly (CF-approach) or implicitly (sorption model). The main reason behind this decision is the closer proximity of the present application conditions and the chemical conditions corresponding to the MX-80 source data. This means that upscaling from the MX-80 source conditions to the application conditions requires a smaller degree of extrapolation than upscaling from the SWy-1 source data.

A comparison with the few available  $D_a$  measurements (Figure 5-20) is not conclusive due to the lack of more systematic diffusion data. However, if it is assumed that  $D_a$  of Ni shows the same trend vs dry density as  $D_a$  of Co, Cs and Am, the derived  $K_d$  values based on MX-80 source data are more consistent with apparent diffusivities than values based on SWy-1 source data. However, all calculated  $D_a$  values fall within the range given by experimental data for Ni and Co.



**Figure 5-20.** Calculated  $D_a$  values for Ni based on  $D_e$  for HTO and selected  $K_d$  values derived as a function of conditions using the CF-approach or calculated using the model of /Bradbury and Baeyens, 1997/. Experimental data obtained on Kunigel-V1 and MX-80 are given for comparison. The high pH indicated by /Christiansen and Torstenfelt, 1988/ may hold only for the input solution, but probably not for the actual porewater. To put the  $D_a$  values for Ni into perspective, experimental data for the similar element cobalt are also shown, and trends of  $D_a$  vs density are indicated for Cs (model calculation for MX-80 by /Ochs et al, 2001/; from Figure 5-11) and for Am (regression line for Kunigel-V1 by /Sato, 1998a/; from Figure 5-8).

In summary, it is recommended to use the values derived from the MX-80 source data as best estimate and the respective lower limit, but to base the upper limit of the recommended data on the uncertainties for the SWy-1 source data. Thus, the upper limit coincides approximately with the thermodynamic model calculations. These limits are very likely to encompass any  $K_d$  for Ni under the specified scenarios.

The data by /Bradbury and Baeyens, 1997/ indicate that  $K_d$  for Ni is likely to decrease above pH 10.

## Niobium

Only few sorption data for bentonite were found, they are reproduced in Figure 5-21. It is likely that the data by /Erdal et al, 1977; Taki and Hata, 1991/ have been obtained in oversatured systems, reflecting solubility limits rather than sorption. Based on the poor database, a  $K_d$  value of 3 m<sup>3</sup>/kg is proposed together with an UF-overall of 15. As Nb is expected to exist nearly exclusively as uncharged penta-hydroxo complex in the pH range considered, this value is proposed for all conditions. No assessments for higher pH can be made.



Figure 5-21. Sorption data for Nb on different bentonites.

# Palladium

Under reducing conditions, Pd is virtually insoluble /Hummel et al, 2002/, and for the present purpose only Pd(II) is considered as potentially relevant. No relevant and reliable sorption data, and no diffusion data for Pd(II) were found. /Tachi et al, 1999/ studied Pd sorption on Kunigel-V1 bentonite as a function of pH and ionic strength, but it cannot be excluded that their data reflect solubility limitation rather than sorption. Assuming sorption to be the relevant process would lead to a mean  $K_d$  of ca 8 m<sup>3</sup>/kg at pH 8 and I=0.1 M. This is relatively close to the value of 5 m<sup>3</sup>/kg deduced by /Bradbury and Baeyens, 2003a/ on the basis of analogies with Ni and Pb. For the present purpose, their estimate is accepted, including the UF-overall of 15. It is considered probable that any  $K_d$  for Pd(II) under the specified scenarios would be within the indicated ranges.

## Plutonium

Depending on redox conditions, Pu can take on the oxidation states III-VI. Because of the uncertainty regarding redox conditions, all oxidation states are treated separately below. If variable redox conditions need to be taken into account, the overall uncertainty for the  $K_d$  of Pu will encompass the entire data range for all oxidation states that will be present.

For Pu(III)), no relevant sorption data were found, and  $K_d$  was derived based on analogy considerations and the source data set used for Am. The results are shown in Figure 5-22, the comparison with  $D_a$  values is shown in Figure 5-23. Inclusion of carbonate species in the speciation-conversion leads to results that are inconsistent with the trend of  $K_d$  vs pH observed for Am. Therefore, the  $K_d$  values based on CF-total 3 are recommended. It can be seen from Figure 5-23 that the selected sorption data are consistent with available  $D_a$  values for III-valent elements and Pu. Therefore, it is considered to be very likely that any  $K_d$  for Pu(III) under the specified scenarios would be within the indicated ranges.

Based on Figure 5-22, a decrease of  $K_d$  for Pu(III) at pH > 10 cannot be excluded.



**Figure 5-22.**  $K_d$  values for Pu(III) derived as a function of conditions based on the experimental data for Am by /Gorgeon, 1994/. The selected values are indicated by open circles (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.



**Figure 5-23.** Calculated  $D_a$  values for Pu(III) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-22 in comparison to experimental data (the value by /Albinsson et al, 1991/, is a maximum value). For comparison, experimental  $D_a$  values for Sm and the regression line representing Am (from Figure 5-8) are also shown.

For Pu(IV), no relevant and reliable sorption data could be found.  $K_d$  was derived based on analogy considerations and the source data set used for Th. The results are shown in Figure 5-24, the comparison with  $D_a$  values is shown in Figure 5-25. The data show that inclusion of carbonate species in the speciation-conversion leads to results for RPWC that are inconsistent with diffusion data for Th as well as Pu. Therefore, the  $K_d$  values based on CF-total 3 are recommended. It can be seen from Figure 5-25 that the selected sorption data are consistent with available  $D_a$  values for Th and Pu. Therefore, it is considered to be likely that any  $K_d$  for Pu(IV) under the specified scenarios would be within the indicated ranges.

Since the hydrolysis behaviour of Pu(IV) appears to be identical to that of Th at high pH /Lemire et al, 2001; Hummel et al, 2002/, no significant influence of a pH increase is expected.

For Pu(V), no relevant and reliable sorption data could be found.  $K_d$  was derived based on analogy considerations and the source data set used for Np(V). The results are shown in Figure 5-26, the comparison with  $D_a$  values is shown in Figure 5-25. The data show that the speciation-conversion appears to have little influence in this case. To be consistent with the approach chosen for other actinides, the  $K_d$  values based on CF-total 3 are recommended. It can be seen from Figure 5-25 that the selected sorption data are fairly consistent with available  $D_a$  values for Np(V). Good agreement can also be observed with the higher diffusivity reported by /Torstenfelt and Allard, 1986/, which reportedly corresponds to a small, mobile fraction of Pu (presumably the oxidised fraction). Therefore, it is considered to be likely that any  $K_d$  for Pu(V) under the specified scenarios would be within the indicated ranges.

Based on the analogy with Np(V) it is expected that the sorption of Pu(V) on montmorillonite increases up to  $pH \approx 10.5$  in the absence of significant amounts of dissolved carbonate. At atmospheric pCO<sub>2</sub>, sorption may start to decrease at  $pH \approx 8.5$ .



*Figure 5-24.*  $K_d$  values for Pu(IV) derived as a function of conditions based on the experimental data for Th by /Bradbury and Baeyens, 2003a/. The selected values are indicated by solid symbols (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.



**Figure 5-25.** Calculated  $D_a$  values for Pu(IV) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-24 in comparison to experimental data (the datum by /Albinsson et al, 1991/, is a maximum value). For comparison, experimental  $D_a$  values for Th are also shown.



**Figure 5-26.**  $K_d$  values for Pu(V) derived as a function of conditions based on the experimental data for Np(V) by /Turner et al, 1998; Bertetti et al, 1998/. The selected values are indicated by open circles (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.



4 MX-80 Na-bentonite, 1% FeO Albinsson et al. (1991)

Figure 5-27. Calculated  $D_a$  values for Pu(V) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-26 in comparison to experimental data (the datum by /Albinsson et al, 1991/, is a maximum value). For comparison, some experimental  $D_a$  values for Np(V) from Figure 5-18 are also shown.

Also for Pu(VI), no relevant and reliable sorption data could be found.  $K_d$  was derived based on analogy considerations and the source data set used for U(VI). The results are shown in Figure 5-28, the comparison with  $D_a$  values is shown in Figure 5-29. The data show that inclusion of carbonate species in the speciation-conversion leads to deviating results for RPWC. To be consistent with the approach chosen for other actinides, the  $K_d$ values based on CF-total 3 are recommended. It can be seen from Figure 5-25 that the selected sorption data are fairly consistent with available  $D_a$  values for U(VI). It is considered to be likely that any  $K_d$  for Pu(VI) under the specified scenarios would be within the indicated ranges.

Based on analogy with U(VI), a significant decrease of Pu(VI) is expected for pH > 8.

## Protactinium

The aqueous chemistry of Pa is not sufficiently known, but it seems to be fairly accepted that it is different enough from that of other actinides to make the use of analogue data highly questionable /Hummel et al, 2002; Baes and Mesmer, 1976/. To our knowledge, no significant new information has become available since /Yu and Neretnieks, 1997/. The  $K_d$  values selected by these authors are fairly consistent with the two  $D_a$  values given in their report. For the present purpose, a  $K_d$  value of 3 m<sup>3</sup>/kg is selected for all conditions, together with a UF-overall of 15. This gives a range that is also approximately compatible with data for IV- and V-valent actinides. It is considered probable that any  $K_d$  for Pa under the specified scenarios would be within this range. No assessment for elevated pH values is made.



**Figure 5-28.**  $K_d$  values for Pu(VI) derived as a function of conditions based on the experimental data for U(VI) by /Pabalan and Turner, 1997/. The selected values are indicated by closed symbols (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.



2MX-80 Na-montorilloniteTorstenfelt et al. (1983)3MX-80 Na-montomorilloniteTorstenfelt et Allard (1986)

4 MX-80 Na-bentonite, 1% FeO Albinsson et al. (1991)

**Figure 5-29.** Calculated  $D_a$  values for Pu(VI) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-28 in comparison to experimental data (the datum by /Albinsson et al, 1991/, is a maximum value). For comparison, some experimental  $D_a$  values for U(VI) from Figure 5-41 are also shown.

#### Radium and strontium

These two elements were treated identically, based on their chemical similarity. No systematic and reliable sorption data for these elements were found in the literature. Therefore, a thermodynamic sorption model was used to calculate  $K_d$ , where for Sr and Ra the same selectivity coefficient as for Ca is used /see Fletcher and Sposito, 1989/. This approach had already been used by /Ochs et al, 1998/ to re-calculate  $K_d$  values for Sr and Ra for a variety of bentonite-water systems. /Ochs et al, 2001, 2003/ also successfully used this approach to model  $K_d$  and  $D_a$  values for compacted Kunigel-V1 bentonite. The results are shown in Figure 5-30, the comparison with  $D_a$  values is given in Figure 5-31. The uncertainty ranges of the calculated  $D_a$  values based on the selected data shown in Figure 5-30 accommodate the scatter of the experimental data, with very few exceptions. Therefore, it is considered to be likely that any  $K_d$  for Sr and Ra under the specified scenarios would be within the indicated ranges.

Based on the corresponding calculated porewater compositions (Appendix C), the influence of (highly) alkaline groundwater on  $K_d$  for Sr/Ra is expected to be within the limits given by the three reference porewaters.

## Radon

Radon is a noble gas. In analogy to methane, a  $K_d$  of zero, together with the diffusion parameters for HTO are proposed.

# Samarium

See section on cerium, holmium and samarium.



*Figure 5-30.*  $K_d$  values for Sr and Ra calculated with the thermodynamic sorption model used by */Ochs et al, 1998/.* 



*Figure 5-31.* Calculated  $D_a$  values for Sr/Ra based on  $D_e$  for HTO and  $K_d$  calculated with the model used by /Ochs et al, 1998/.

#### Selenium

Relevant oxidation states in aqueous solutions include –II, IV and VI, with HSe<sup>-</sup>, SeO<sub>3</sub><sup>2-</sup> and SeO<sub>4</sub><sup>2-</sup> as the most important species in typical waters. As in the case of other redox-sensitive elements, if variable redox conditions need to be taken into account the overall uncertainty for the  $K_d$  of Se would encompass the combined data range for all oxidation states predicted to be present.

Sorption of selenate and selenite on metal oxides has been studied, confirming innersphere surface complexation/ligand exchange as predominant sorption mechanism for selenite /Hayes, 1987; Davis and Kent, 1990; Séby et al, 1998/. However, only two reliable studies have been found for the sorption of selenite on bentonite or smectite under relevant conditions /Bar-Yosef and Meek, 1987; Bradbury and Baevens, 2003a/. /Bar-Yosef and Meek, 1987/ determined isotherm as well as pH-edge sorption data for selenite on Ca-Wyoming montmorillonite in 5 mM CaCl<sub>2</sub> electrolyte. /Bradbury and Baeyens, 2003a/ report a sorption isotherm on MX-80 bentonite in a porewater solution at pH 7.8 and a sorption edge on Na-SWy-1 at trace concentrations of selenite. At approximately comparable Se concentrations and pH, these two studies give nearly identical results (the data by /Bar-Yosef and Meek, 1987/, fall within the uncertainty limits indicated by /Bradbury and Baeyens, 2003a/). /Saha et al, 2004/ very recently also investigated selenite adsorption on montmorillonite, but unfortunately limited their experiments to pH 4.5. /Boult et al, 1998/ also investigated Se(IV) sorption on bentonite, but used only high initial Se concentrations. As they do not report details of the solution composition, oversaturation with respect to  $CaSeO_3(s)$  cannot be excluded with certainty. Following the above discussion, the data given for MX-80 in /Bradbury and Baeyens, 2003a/ were selected as the most reliable and directly applicable sorption data. The resulting derived  $K_d$  values as a function of conditions are shown in Figure 5-32.  $K_d$  values were additionally derived from SWy-1, they are largely contained in the range of uncertainty given for the data derived based on MX-80, and use of the latter is recommended. Note that these values are valid for Se concentrations < 1e–6 M only.

The comparison with  $D_a$  measurements is given in Figure 5-33. Because it is not clear which Se concentration was used in the diffusion experiment, it must be assumed that the concentrations were high enough to lower the  $K_d$  of Se(IV) to near zero (i.e. around 0.001 M, according to the isotherm data of /Bradbury and Baeyens, 2003a/). The experiments by /Sato et al, 1995/ were conducted under absence of oxygen, and



*Figure 5-32.*  $K_d$  values for Se(IV) derived as a function of conditions based on the experimental data by /Bradbury and Baeyens, 2003a/. The selected values (solid symbols) are based on CF-total 2); error bars refer to these values. For an explanation of CF-total 1/2, see section 5.1.4.
oxidation of Se(IV) to Se(VI) can probably be excluded. On the other hand, Figure 5-33 corroborates the choice of diffusion parameters for non-sorbing anions.

Based on the batch sorption data, it is considered to be likely that any  $K_d$  for Se(IV) under the specified scenarios would be within the indicated ranges.

The data by /Bradbury and Baeyens, 2003a/ and by /Bar-Yosef and Meek, 1987/ suggest that sorption of Se(IV) decreases sharply above pH 8–9, which would be consistent with a ligand exchange mechanism.

No reliable and relevant sorption data have been found in case of selenate and selenide. Therefore, a  $K_d$  of zero is assigned to HSe<sup>-</sup> and SeO<sub>4</sub><sup>2-</sup>, together with the diffusion parameters selected for anions (section 5.3.4). No different diffusion parameters are proposed for SeO<sub>4</sub><sup>2-</sup>, even though anion exclusion effects could be more prominent due to its double negative charge.

Note that assuming zero sorption is probably conservative in the case of selenate. /Bar-Yosef and Meek, 1987/ determined isotherm as well as pH-edge sorption data for selenate on Ca-kaolinite in 5 mM CaCl<sub>2</sub> electrolyte. At pH 6.7, a K<sub>d</sub> of ca 0.02 L/g can be deduced from their data. However, in their experiments the dissolved selenate concentration is close to the initial concentration at this pH, and indistinguishable from it at pH 8, which makes reliable measurements difficult. Further, /Bar-Yosef and Meek, 1987/ used a very dilute background electrolyte, and the data by /Hayes, 1987/ show that selenate sorption on goethite is highly dependent on the concentration of the background electrolyte (NaNO<sub>3</sub> in their case) and suggest that dissolved anions, in particular sulphate, may strongly compete with selenate sorption on bentonite. This is in line with spectroscopic data that show that selenate sorbs via inner- and outer-sphere complexation /Peak and Sparks, 2002/ and modelling work by /Hiemstra and van Riemsdijk, 2004/, which indicates that the relative importance of inner-sphere binding decreases with increasing pH.



1 Kunigel-V1 Sato et al. 1995

**Figure 5-33.** Calculated  $D_a$  values for Se(IV) and Se(VI) based on diffusion parameters for anions and  $K_d$  values from Figure 5-32 for Se(IV) in comparison to experimental data reported for Se(IV).  $D_a$  for Se(VI) was calculated with  $K_d$  zero.

### Silver

No reliable sorption data and no diffusion data were found. /Legoux et al, 1992/ determined four  $K_d$  values for Ag on four sediment samples containing ca 10% illite and smectite. These values range from 1.2–17 m<sup>3</sup>/kg. On the other hand, /Bradbury and Baeyens, 2003a/ carried out calculations with a thermodynamic sorption model, considering only ion exchange (based on a selectivity coefficient taken from /Pleysier and Cremers, 1975/). These calculations gave  $K_d$  values < 1E–7 m<sup>3</sup>/kg. This low value may possibly be attributable to the neglect of surface complexation in their model.

Based on the above, an overall  $K_d$  range of 0–15 m<sup>3</sup>/kg is proposed for Ag, with no mean or median value, for all conditions considered. It is considered probable that any  $K_d$  for Ag under the specified scenarios would fall in this range. No assessment for the effects of elevated pH is made.

## Technetium

Relevant oxidation states of Tc in aqueous solutions are VII and IV. If variable redox conditions need to be taken into account the overall uncertainty for the  $K_d$  of Tc would encompass the combined data range for all oxidation states predicted to be present.

Tc(VII) exist as the pertechnetate anion,  $TcO_4^-$ . For this oxidation state, a  $K_d$  of zero is proposed together with the diffusion parameters for anions.

For Tc(IV), no relevant sorption data were found. Because Tc(IV) and Th are both tetravalent and exists at circumneutral pH primarily as the uncharged species  $(TcO(OH)_2^0, Th(OH_4^0) K_d$  values for Tc(IV) were derived based on analogy considerations and the source data set used for Th. However, Th and Tc(IV) do not form identical species in solution and may, therefore, also show different sorption behaviour. To account for this additional uncertainty, UF-starting K<sub>d</sub> was increased to 4.0 (see section 5.1.4).

The results are shown in Figure 5-34, and the corresponding calculated  $D_a$  values are compared in Figure 5-35 with experimental data for Th. Diffusivities of Th were used for comparison because no diffusion data were found for Tc where it was clearly demonstrated that no oxidation to Tc(VII) took place. The data show that inclusion of carbonate species in the speciation-conversion leads to very high  $K_d$  values that are inconsistent with sorption data for Th; and would further lead to equally unrealistic  $D_a$  values of ca  $10^{-18}$  m<sup>2</sup>/s. Therefore, the K<sub>d</sub> values based on CF-total 3 are recommended. It can be seen from Figure 5-39 that the sorption data selected on this basis are not inconsistent with available  $D_a$  values for Th. It is considered to be likely that any K<sub>d</sub> for Tc(IV) under the specified scenarios would be within the indicated ranges.

Based on the solubility/hydrolysis behaviour of Tc(IV) /Hummel et al, 2002/, it is expected that pH has no significant influence on sorption up to pH  $\approx 10$ .

## Thorium

The data obtained by /Bradbury and Baeyens, 2003a/, a sorption isotherm on MX-80 in synthetic porewater as well as a sorption edge on SWy-1 montmorillonite in 0.1 M NaClO<sub>4</sub>, were considered to be the most reliable and directly applicable data source available to date. Because of the significant uncertainty in evaluating i) the formation of mixed hydroxocarbonato complexes and ii) the formation of ternary surface carbonato complexes, the MX-80 source data are preferable because of the relative similarity of this system to the application conditions (see appendices C and F). The derived  $K_d$  values are shown in Figure 5-36, a comparison with the few available  $D_a$  values is given in Figure 5-37.



**Figure 5-34.**  $K_d$  values for Tc(IV) derived as a function of conditions based on the experimental data for Th by /Bradbury and Baeyens, 2003a/. The selected values are indicated by solid symbols (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.



**Figure 5-35.** Calculated  $D_a$  values for Tc(IV) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-36 in comparison to experimental data for Th (from Figure 5-37).



**Figure 5-36.**  $K_d$  values for Th derived as a function of conditions based on the isotherm data determined by /Bradbury and Baeyens, 2003a/ on MX-80 at pH 7.2–7.7. The selected values are indicated by solid symbols (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.



MX-80 Na-montmorillonite Torstenfelt et al., (1982)
 MX-80 Na-montmorillonite Torstenfelt and Allard (1986)

*Figure 5-37.* Calculated  $D_a$  values for Th based on  $D_e$  for HTO and  $K_d$  values from Figure 5-36 in comparison to experimental data.

Considering all evidence, the inclusion of Th-carbonate species in CF-speciation is considered inappropriate, and the data based on CF-total 3 are recommended. The resulting  $D_a$  values are not inconsistent with independent experimental data, and it is considered to be very likely that any  $K_d$  for Th under the specified scenarios would be within the indicated ranges.

The sorption data obtained by /Bradbury and Baeyens, 2003a/ as a function of pH on SWy-1 montmorillonite indicate no decrease of Th sorption up to pH 11–11.5.

### Tin

No reliable and relevant sorption or diffusion data for Sn(IV) were found. Sn(II) is not considered for the present purpose, as it will form only under extremely reducing conditions. /Bradbury and Baeyens, 2003a/ present a sorption edge for Sn which suggests that  $K_d$  values may be as high as > 800 m<sup>3</sup>/kg. Their data are compatible with the findings of /Tachi et al, 1999/. However, initial Sn concentrations in the latter study were near or above the solubility limit as determined by /Amaya et al, 1997/. /Bradbury and Baeyens, 2003a/ do not report the initial Sn concentration, and their data are also not considered, therefore.

Based on the fact that both Sn(IV) and Th exist as uncharged tetra-hydroxo species in the pH range of about 6.0–7.5 /Hummel et al, 2002/, it is proposed to use Th directly as analogue for Sn(IV). Following the discussion in section 5.1.4, a UF-starting K<sub>d</sub> of 4.0 (instead of 1.6 for Th) is proposed to account for differences in the chemical behaviour of Sn(IV) and Th in the relevant pH range 6.6–7.4, but no further uncertainties are proposed. Note that this assessment holds only for the pH range under consideration, as the hydrolysis behaviour of Th and Sn(IV) starts to differ more significantly outside of this pH range.

Based on the hydrolysis behaviour of Sn(IV) /Amaya et al, 1997/, it seems very likely that the K<sub>d</sub> will decrease at pH values > 8 (in contrast to the sorption behaviour of Th).

### Uranium

In relevant aqueous solutions, stable oxidation states of U can be IV or VI, depending on redox conditions. Because of the uncertainty regarding redox conditions, both oxidation states are treated separately below. The overall uncertainty for the  $K_d$  of U encompasses the entire combined data range if variable redox conditions need to be taken into account.

For U(IV), no relevant sorption data were found, and  $K_d$  was derived based on analogy considerations and the source data set used for Th. The results are shown in Figure 5-38, the comparison with  $D_a$  values is shown in Figure 5-39. The data show that inclusion of carbonate species in the speciation-conversion leads to very high  $K_d$  values that are inconsistent with sorption data for Th; and would further lead to equally unrealistic  $D_a$  values of ca  $10^{-18}$  m<sup>2</sup>/s. Therefore, the  $K_d$  values based on CF-total 3 are recommended. It can be seen from Figure 5-39 that the selected sorption data are not consistent with available  $D_a$  values for U. Comparison with the  $D_a$  values for Th indicates that the redox state in the U diffusion experiments may not have been controlled well enough to completely exclude the presence of oxidised uranium species. The overall diffusion may thus represent a mix of reduced and oxidised U fractions /see also Torstenfelt and Allard, 1986/. Therefore, more weight is placed on the derived  $K_d$  values in the present assessment, and it is considered to be likely that any  $K_d$  for U(IV) under the specified scenarios would be within the indicated ranges.



*Figure 5-38.*  $K_d$  values for U(IV) derived as a function of conditions based on the isotherm data determined by /Bradbury and Baeyens, 2003a/ for Th on MX-80 at pH 7.2–7.7. The selected values are indicated by solid symbols (CF-total 3); error bars are given for CF-total 2, 3. For an explanation of CF-total 1/2/3, see section 5.1.4.



**Figure 5-39.** Calculated  $D_a$  values for U(IV) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-38 in comparison to experimental data (experiments where a reducing agent had been used are depicted by open symbols). For comparison, experimental  $D_a$  values for Th are also shown.

Based on the solubility/hydrolysis behaviour of U(IV) /Hummel et al, 2002/, no decrease of  $K_d$  is expected up to pH 11 or higher, following the analogy with Th.

In case of U(VI), the literature on sorption is relatively extensive, but the majority of studies is concerned with Fe(III)- and other metal oxides as well as with kaolinite, because of their importance for the substrates found at the Koongarra uranium deposit analogue site /cf Waite et al, 2000; Davis, 2001, and references therein/. Less work has been done on bentonite or smectite minerals in comparison. There are several spectroscopic studies that confirm that uranyl ions sorb to montmorillonite by inner-sphere surface complexation at edge sites and outer-sphere complexation at exchange sites, with the latter being predominant only at low pH /Sylvester et al, 2000; Chisholm-Brause et al, 2001/.

Only a limited number of studies addresses the sorption behaviour of U(VI) on montmorillonite or bentonite as a function of conditions. In particular, few systematic data are available regarding the influence of carbonate, which appears to be a critical factor for uranyl sorption, both in terms of competition by forming dissolved uranium carbonato complexes and in terms of forming ternary surface complexes. /Davis, 2001/. There are several studies where U(VI) sorption on smectite or bentonite is investigated in the presence of low but ill-defined carbonate concentrations /see e.g. Boult et al. 1998/: these are generally not considered for the present work and not discussed further. Zachara and co-workers /Turner et al, 1996; McKinley et al, 1995/ studied uranyl adsorption on SWy-1 montmorillonite and a natural smectite as a function of pH and ionic strength, but in the absence of carbonate. Moreover, the sample of natural smectite was reportedly a ferrogenous beidellite, which may have different properties than montmorillonite, the principal smectite mineral in bentonites. /Pabalan et al, 1998/ and /Pabalan and Turner, 1997/ published uranyl sorption data on Na-montmorillonite (SAz-1) as a function of pH, solid/water ratio and uranyl concentration at ambient pCO<sub>2</sub>. As the only systematic study carried out in the presence of carbonate, their experiments were considered to have the closest chemical proximity to the present reference conditions and were selected as representative source data set. In comparison to other studies /e.g. Turner et al, 1996/ the appropriate choice and variation of solid/water ratio renders their data also more trustworthy and easier to interpret. Both of the above datasets had been used for the development of thermodynamic sorption models /Pabalan and Turner, 1997; Turner et al, 1996/, but both models have significant shortcomings as they neglect the formation of ternary U-carbonato surface complexes and were therefore not used.

 $K_d$  values derived with the CF-approach are shown in Figure 5-40. It can be seen that the highest sorption is reached at the lowest pH value according to these data. However, this trend is consistent with data for U(VI) sorption on montmorillonite, clinoptilite, and quartz /Pabalan et al, 1998/. It is therefore considered that this trend is relevant. As the formation of ternary U-carbonato surface complexes enhances sorption,  $K_d$  is underpredicted when complexation of uranyl with carbonate is taken as competitive with respect to sorption (CF-total 2). On the other hand, CF-total 1 and 3 give identical results, indicating that carbonate species and the hydroxo ion are the only relevant ligands effecting surface complexation. Ion exchange is not important at the relevant ionic strengths /Pabalan and Turner, 1997/.

The comparison with  $D_a$  values can be seen in Figure 5-41. As in the case of U(IV), the comparison is not conclusive. The discrepancy between  $D_a$  values for RPW and HSPW calculated from  $K_d$  values derived with either CF-total 3 or CF-total 2 is within the scatter of experimental data, represented by the measurements of /Idemitsu et al, 1996/ in Kunigel-V1 on one hand vs /Idemitsu et al, 1996/ in Kunipia –F and /Muurinen et al, 1989/ in MX-80 on the other. The latter data might be the closest representation of the relevant system and agree better with  $D_a$  values calculated from  $K_d$  values derived with CF-total 3.



**Figure 5-40.**  $K_d$  values for U(VI) derived as a function of conditions based on the experimental data by /Pabalan and Turner, 1997/. The selected values are indicated by solid symbols (CF-total 3); error bars are given for these values and the data derived using CF-total 2. For an explanation of CF-total 1/2/3, see section 5.1.4.



**Figure 5-41.** Calculated  $D_a$  values for U(VI) based on  $D_e$  for HTO and  $K_d$  values from Figure 5-40 in comparison to experimental data (experiments where a reducing agent had been used are depicted by open symbols).

For the same reasons as for U(IV), more weight is placed on the batch source data in the present assessment, and it is considered to be likely that any  $K_d$  for U(VI) under the specified scenarios would be within the indicated ranges.

Based on the data by /Pabalan and Turner, 1997/, a significant decrease of  $K_d$  above pH 8–9 is expected.

### Zirconium

No sorption data measured in systematic and well-controlled conditions could be found.  $K_d$  for Zr was derived based on analogy considerations and the source data set used for Th. The results are shown in Figure 5-42. Because the hydrolysis behaviour of Zr(IV) starts to differ from that of Th(IV) at circumneutral pH values (in contrast to Th, Zr forms Zr(OH)<sub>5</sub><sup>-</sup> species, see /Hummel et al, 2002/), dissolved hydroxo –complexes were explicitly included in the speciation conversion (see appendices E and F). As in the case of Tc(IV), the additional uncertainty due to different hydrolysis behaviour is taken into account by increasing UF-starting K<sub>d</sub> to 4.0 (instead of to 2.6 as for well-matched analogues, see section 5.1.4). The recommended data are based on CF-total 3, as in the case of Th.

The resulting  $D_a$  values are compared with experimental data in Figure 5-43. Based on the good agreement, it is estimated as very likely that any  $K_d$  for Zr under the specified scenarios would be within the indicated ranges



**Figure 5-42.**  $K_d$  values for Zr(IV) derived as a function of conditions based on the isotherm data determined by /Bradbury and Baeyens, 2003a/ for Th on MX-80 at pH 7.2–7.7. The selected values are indicated by solid symbols (CF-total 3); error bars refer to these values. For an explanation of CF-total 1/2/3, see section 5.1.4.





*Figure 5-43.* Calculated  $D_a$  values for Zr based on  $D_e$  for HTO and the  $K_d$  value and uncertainty proposed above in comparison to experimental data.  $D_a$  values for Th are given for comparison.

## 5.4 Data Summary

This section summarises the migration parameters and associated uncertainties derived in section 5.3 for a bentonite buffer corresponding to the KBS-3 concept (see also section 5.2.6). The derived migration parameters are:

- effective diffusivity (D<sub>e</sub>),
- diffusion-available porosity (ε),
- distribution coefficient (K<sub>d</sub>).

In addition, apparent diffusivities ( $D_a$ ) were selected from the literature to evaluate the self-consistency of the selected migration parameters. These parameters as well as the associated uncertainties were derived for 26 radionuclides. In case of redox-sensitive elements, each relevant oxidation state was evaluated separately. The validity range of the derived data in terms of bentonite properties and density, as well as in terms of geochemical conditions is summarised below; a detailed discussion is provided in chapters 4 and 5.

### Bentonite composition and density

### K<sub>d</sub>

The derived migration parameters were derived for MX-80 bentonite of a given composition (Appendix B) at the specified reference density of 1,590 kg/m<sup>3</sup>. Because of the strong dependency of  $K_d$  on bentonite porewater composition, these data are strictly valid for these conditions only. However, it is not likely that a limited variation of the buffer density (e.g. within a range of about 1,400–1,700 kg/m<sup>3</sup>) will have a significant influence on porewater composition or  $K_d$ , in particular when pCO<sub>2</sub> is imposed by the host rock formation (see below). Similarly, the bentonite composition was varied for the present evaluation to include a pure Ca-form of MX-80 as well as a MX-80 depleted of soluble impurities. The resulting variability of porewater composition was found to be within the variability due to uncertainties in the groundwater composition and pCO<sub>2</sub>. It follows from the above, that under comparable conditions, the derived values will also have a certain validity for other bentonites, as long as the CEC is similar to that of MX-80.

### Porosity

For all elements except non-sorbing anions, the same value is proposed for diffusionavailable and total porosity. As shown in Figure 5-3, total porosity depends on buffer density, but is not significantly influenced by composition. According to eq. (5.6), total porosity is readily calculated as  $\varepsilon = 1 - (\rho/\rho_s)$ . In the same manner, diffusion-available porosity for non-sorbing anions is given by  $\varepsilon_{anion} = (1 - (\rho/\rho_s))/2.5$ ; see eq. (5.8).

## De

For all elements except non-sorbing anions and Cs,  $D_e$  for HTO is used. The  $D_e$  values plotted in Figure 5-4 for four different bentonites as a function of density show a clear dependency of  $D_e$  on density. Using eq. (5.7), this dependency can be expressed as:

 $D_e = 6.7785E - 9 \times e^{-2.5671E - 3 \times \rho}$ 

Similarly, using eq. (5.9), the dependency of  $D_e$  for non-sorbing anions can be expressed as a function of density; this relation is valid for a density range of ca 1,000–2,000 kg/m<sup>3</sup>:

 $D_e = -1.1844 \times 10^{-20} \times \rho^3 + 8.130 \times 10^{-17} \times \rho^2 - 1.9473 \times 10^{-13} \times \rho + 1.6167 \times 10^{-10}$ 

For Cs, no explicit relation of De vs density is given, but it is indicated in section 5.3.4 that the same dependency on density as for HTO may be used, with  $D_e$  for Cs being larger by a constant factor of 3.

Very little quantitative information is available regarding the dependency of  $D_e$  on bentonite composition. Based on Figure 5-4, it can be estimated, however, that the variation of  $D_e$  among significantly dissimilar bentonites (assuming all other experimental conditions being the same) is within a factor of ca 5; i.e. within experimental error.

### Geochemical conditions (groundwater composition, pCO<sub>2</sub>, redox conditions)

As pointed out in Figure 4-2 and Figure 4-3, migration parameters were selected for specific sets of bentonite porewater compositions. These porewater compositions are the result of bentonite-groundwater interaction and correspond to particular combinations of i) buffer composition and density (see above) and ii) geochemical conditions (groundwater composition, pCO<sub>2</sub>). It is pointed out that changes in groundwater composition do not translate directly to corresponding changes in porewater composition, due to the buffering effect of bentonite. An approximate estimation of the possible effects of variations in the groundwater composition may be possible based on the results given in Appendix C. The effect of redox conditions was addressed by separately evaluating each relevant oxidation state in case of redox-sensitive elements.

As discussed for each element in scetion 5.3.5,  $K_d$  values can be very sensitive with regard to porewater composition. Therefore, the validity of the selected data is bounded by the ranges of conditions covered by the porewater compositions considered: (pH 6.6 / pCO<sub>2</sub>  $1 \times 10^{-1}$  atm) to (pH 7.4 / pCO<sub>2</sub>  $1 \times 10^{-2.6}$  atm), reflecting closed and open conditions with respect to CO<sub>2</sub>. It is not advisable to extrapolate K<sub>d</sub> values to conditions outside these ranges. Salinity is somewhat of an exception to this, because of the comparative "inertness"

of Na and Cl; it is not expected that an increase of salinity to ca 1 M will have a significant effect on  $K_d$  (except for Cs, Sr, Ra).

For an approximate estimation of the possible effects of variations in the porewater composition, the diagrams in section 5.3.5 can be used. Interpolation between the three reference porewater conditions is possible for salinity, but should to be done with great care for parameters that are directly linked to others (pH,  $pCO_2$ ).

In general, not enough diffusion data are available as a function of conditions to allow a sound assessment of the effects of variable conditions. The effect of high salinity (up to 1M) on  $D_e$  of non-sorbing anions can be estimated from Figure 5-4. No significant effects are expected for such salinities. Whether even higher salinities will lead to significantly increased  $D_e$  values for anions cannot be answered conclusively to date. For sorbing elements, it is expected that the uncertainty of  $D_e$  as a function of conditions will not be relevant in comparison to the corresponding uncertainty in  $K_d$ .

#### Temperature

The selected data are valid for 25°C. An increase to 50°C is expected to lead to a twofold increase of  $D_e$  (see 5.3.4). Based on the data available to date, it is not possible to assess the influence of temperature on  $K_d$  with any certainty. However, no significant effects are expected for an interval of ca 10–50°C. No effect on porosity is expected for this temperature range.

Summary	of D <sub>e</sub>	and $\epsilon$	values
---------	-------------------	----------------	--------

Radionuclide	D <sub>e</sub>	Upper D <sub>e</sub> limit	Lower D <sub>e</sub> limit	ε	Upper $\epsilon$ limit	Lower $\epsilon$ limit
(Redox State)	(m²/s)	(m²/s)	(m²/s)	(-)	(-)	(-)
Ag(I)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Am(III)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
C, carbonate species	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
C, methane	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.17	0.12	0.24
C, organic acids	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
Ce(III)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
CI(-I)	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
Cm(III)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Cs(I)	3.0 x 10 <sup>-10</sup>	3.0 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Eu(III)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Ho(III)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
l(-l)	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
Nb(V)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Ni(II)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Np(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Np(V)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pa(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pa(V)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pb(II)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pd(II)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pu(III)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pu(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pu(V)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Pu(VI)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Ra(II)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Rn(-)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Se(-II)	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
Se(IV)	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
Se(VI)	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
Sm(III)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Sn(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Sr(II)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Tc(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Tc(VII)	1.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-11</sup>	3.0 x 10 <sup>-12</sup>	0.17	0.12	0.24
Th(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
U(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
U(VI)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-
Zr(IV)	1.2 x 10 <sup>-10</sup>	1.91 x 10 <sup>-10</sup>	4.94 x 10 <sup>-11</sup>	0.43	-	-

Reference porewater (	RPW)		
Radionuclide	K <sub>d</sub>	Upper K <sub>d</sub> limit	Lower K <sub>d</sub> limit
(Redox State)	(m <sup>3</sup> /kg)	(m <sup>3</sup> /kg)	(m <sup>3</sup> /kg)
Ag(I)	-	15	0
Am(III)	61	378	10
C, carbonate species	isotope exchange	-	-
C, methane	0	-	-
C, organic acids	0	-	-
Ce(III)	8	93	0.8
CI(-I)	0	-	-
Cm(III)	61	378	10
Cs(I)	0.11	0.6	0.018
Eu(III)	8	93	0.8
Ho(III)	8	93	0.8
l(-l)	0	-	-
Nb(V)	3	45	0.2
Ni(II)	0.30	3.3	0.03
Np(IV)	63	1113	4
Np(V)	0.02	0.2	0.004
Pa(IV)	3	45	0.2
Pa(V)	3	45	0.2
Pb(II)	74	457	12
Pd(II)	5	75	0.3
Pu(III)	100	984	10
Pu(IV)	63	1111	4
Pu(V)	0.02	0.2	0.002
Pu(VI)	3	28	0.3
Ra(II)	0.005	0.03	0.001
Rn(-)	0	-	-
Se(-II)	0	-	-
Se(IV)	0.04	0.4	0.003
Se(VI)	0	-	-
Sm(III)	8	93	0.8
Sn(IV)	63	1764	2.3
Sr(II)	0.005	0.031	0.0009
Tc(IV)	63	1764	2.3
Tc(VII)	0	-	-
Th(IV)	63	700	6
U(IV)	63	1113	3.6
U(VI)	3	18	0.5
Zr(IV)	4	103	0.1

Reference porewater for closed system with respect to CO <sub>2</sub> (RPWC)					
Radionuclide	K <sub>d</sub>	Upper K <sub>d</sub> limit	Lower K <sub>d</sub> limit		
(Redox State)	(m <sup>3</sup> /kg)	(m <sup>3</sup> /kg)	(m <sup>3</sup> /kg)		
Ag(I)	-	15	0		
Am(III)	11	68	2		
C, carbonate species	isotope exchange	-	-		
C, methane	0	-	-		
C, organic acids	0	-	-		
Ce(III)	1	14	0.1		
CI(-I)	0	-	-		
Cm(III)	11	68	2		
Cs(I)	0.10	0.6	0.017		
Eu(III)	1	14	0.1		
Ho(III)	1	14	0.1		
l(-l)	0	-	-		
Nb(V)	3	45	0.2		
Ni(II)	0.06	0.7	0.01		
Np(IV)	40	703	2		
Np(V)	0.01	0.1	0.002		
Pa(IV)	3	45	0.2		
Pa(V)	3	45	0.2		
Pb(II)	35	219	6		
Pd(II)	5	75	0.3		
Pu(III)	30	300	3		
Pu(IV)	40	703	2		
Pu(V)	0.01	0.1	0.001		
Pu(VI)	14	139	1.4		
Ra(II)	0.005	0.03	0.001		
Rn(-)	0	-	-		
Se(-II)	0	-	-		
Se(IV)	0.09	1.0	0.008		
Se(VI)	0	-	-		
Sm(III)	1	14	0.1		
Sn(IV)	40	1113	1.4		
Sr(II)	0.005	0.03	0.0009		
Tc(IV)	40	1113	1.4		
Tc(VII)	0	-	-		
Th(IV)	40	442	4		
U(IV)	40	703	2.3		
U(VI)	14	88	2		
Zr(IV)	11	305	0.4		

Highly saline porewate	er (HSPW)		
Radionuclide	K <sub>d</sub>	Upper K <sub>d</sub> limit	Lower K <sub>d</sub> limit
(Redox State)	(m <sup>3</sup> /kg)	(m <sup>3</sup> /kg)	(m <sup>3</sup> /kg)
Ag(I)	-	15	0
Am(III)	24	152	4
C, carbonate species	isotope exchange	-	-
C, methane	0	-	-
C, organic acids	0	-	-
Ce(III)	5	57	0.5
CI(-I)	0		
Cm(III)	24	152	4
Cs(I)	0.03	0.2	0.006
Eu(III)	5	57	0.5
Ho(III)	5	57	0.5
l(-l)	0	-	-
Nb(V)	3	45	0.2
Ni(II)	0.07	0.8	0.01
Np(IV)	40	702	2
Np(V)	0.02	0.1	0.004
Pa(IV)	3	45	0.2
Pa(V)	3	45	0.2
Pb(II)	46	287	7
Pd(II)	5	75	0.3
Pu(III)	43	421	4
Pu(IV)	40	700	2
Pu(V)	0.02	0.2	0.002
Pu(VI)	3	28	0.3
Ra(II)	0.001	0.01	0.0002
Rn(-)	0	-	-
Se(-II)	0	-	-
Se(IV)	0.05	0.6	0.005
Se(VI)	0	-	-
Sm(III)	5	57	0.5
Sn(IV)	40	1113	1.4
Sr(II)	0.001	0.008	0.0002
Tc(IV)	40	1113	1.4
Tc(VII)	0	-	_
Th(IV)	40	442	4
	40	703	23
	3	18	0.5
Zr(IV)	5	134	0.2

# 6 Concluding remarks

For the present selection of radionuclide migration parameters for the bentonite buffer, significant efforts were made to select the scientifically best and most defendable values, and to clearly define the associated uncertainties. To this end, different contributions to the overall uncertainties were quantified separately. Relevant uncertainties can stem from

- uncertainties in the source data (experimental error),
- uncertainties in applying the source data to the expected in-situ conditions for PA (errors in parameter transfer),
- uncertainties in defining the expected in-situ conditions for PA (variability of conditions).

It follows directly form this list that the critical points for minimising uncertainties in selected radionuclide migration parameters for the bentonite buffer are

- to obtain the best possible description of the solution chemistry under the in-situ conditions of interest; and
- to be able to rely on data sources that closely match these solution conditions.

The goal of the present data evaluation was the selection of parameters applicable to the conditions in compacted MX-80 bentonite. While diffusion parameters could be derived directly for compacted conditions,  $K_d$  values from batch experiments needed to be converted to compact conditions. This was done by fully taking into account the dependency of  $K_d$  on the calculated, expected porewater composition of the compacted bentonite buffer.

The distribution of uncertainties among the relevant sources also allowed to document all data derivation procedures with a high degree of transparency and traceability. This is obviously important to increase confidence, but will also facilitate any subsequent updates or changes (due e.g. to modifications in the expected groundwater composition, etc).

# 7 References

Albinsson Y, Forsyth R, Skanemark G, Skålberg M, Torstenfelt B, Werme L, 1990. Leaching/migration of UO2-fuel in compacted bentonite, In: Overby V.M. and Brown, P.W. (eds) Material Research Society Symp. Proc., 176, 559–565.

Albinsson Y, Christiansen-Sätmark B, Engkvist I, Johansson W, 1991. Transport of actinides and Tc through a bentonite backfill containing small quantities of iron or copper, Radiochim. Acta, 52/53, 283–286.

Albinsson Y, Andersson K, Börjesson S, Allard B, 1993. Diffusion of radionuclides in concrete/bentonite systems, SKB TR 93-29, Svensk Kärnbränslehantering AB.

**Amaya T, Chiba T, Suzuki K, Oda C, Yoshikawa H, Yui M, 1997.** Solubility of Sn(IV) oxide in dilute NaClO<sub>4</sub> solution at ambient temperature. Mat. Res. Soc. Symp. Proc. 465, 751–758.

Andersson J, 1999. Data and data uncertainties. Compilation of data and data uncertainties for radionuclide transport calculations, SKB TR-99-09, Svensk Kärnbränslehantering AB.

**Andersson J, 2003.** Site descriptive modelling - strategy for integrated evaluation, R-03-05, Swedish Nuclear Fuel and Waste Management Co., Stockholm.

**Baes C F, Mesmer R E, 1976.** The Hydrolysis of cations, Krieger Publishing, Malabar, USA, pp. 499.

**Baeyens B, Bradbury M H, 1997.** A mechanistic description of Ni and Zn sorption on Na-montmorillonite Part I: titration and sorption measurements, J. Contam. Hydrol. 27, 199–222.

**Bargar J, Reitmeyer R, Davis J A, 2001.** Spectroscopic characterization of U(VI)-carbonato surface complexes on hematite and ferrihydrite. In Surface complexation modelling of uranium (VI) adsorption on natural mineral assemblages. NUREG/CR-6708, 19-37.

**Bar-Yosef B, Meek D, 1987.** Selenium sorption by kaolinite and montmorillonite. Soil Science 144, 11–19.

**Berner U, 1993.** MIN\_SURF: PSI version of MINEQL code, revised February 1993, PSI, Villigen, Switzerland, personal communication.

**Bertetti F P, Pabalan R T, Almendarez M G, 1998.** Studies of neptuniumV sorption on quartz, clinoptilite, montmorillonite and  $\alpha$ -alumina. In: Adsorption of metals by geomedia. (ed. E. Jenne), Academic Press, New York, pp. 131–148.

**Boult K A, Cowper M M, Heath T G, Sato H, Shibutani T, Yui M, 1998.** Towards an understanding of the sorption of U(VI) and Se(IV) on sodium bentonite. J. Contam. Hydrol. 35, 141–150.

**Bradbury M H, Baeyens B, 1997.** A mechanistic description of Ni and Zn sorption on Na-montmorillonite. Part II: modelling. J. Contaminant Hydrology, 27, 223–248.

**Bradbury M H, Baeyens B, 1998.** N2-BET surface area measurements on crushed and intact minerals and rocks: A proposal for estimating sorption transfer factors. Nucl. Technol. 122, 325–336.

**Bradbury M H, Baeyens B, 1999.** Modelling the sorption of Zn and Ni on Ca-montmorillonite. Geochim. Cosmochim. Acta, 63, 3/4, 325–336.

**Bradbury M H, Baeyens B, 2002a.** Sorption of Eu on Na- and Ca-montmorillonites: Experimental investigations and modelling with cation exchange and surface complexation. Geochim. Cosmochim. Acta, 66, 13, 2325–2334.

**Bradbury M H, Baeyens B, 2002b.** A comparison of apparent diffusion coefficients measured in compacted Kunigel V1 bentonite with those calculated from batch sorption measurements and De (HTO) data: A case study for Cs(I), Ni(II), Sm(III), Am(III), Zr(IV) and Np(V). PSI-Report 03-02.

**Bradbury M H, Baeyens B, 2003a.** Far-field sorption data bases for performance assessment of high-level radioactive waste repository in an undisturbed Opalinus Clay host rock. Technical Report 02-18. Nagra, Wettingen, Switzerland.

**Bradbury M H, Baeyens B, 2003b.** Near-field sorption data bases for compacted MX-80 bentonite for performance assessment of high-level radioactive waste repository in Opalinus Clay host rock. Technical Report 02-19. Nagra, Wettingen, Switzerland.

**Bradbury M H, Baeyens B, 2003c.** Porewater chemistry in compacted re-saturated MX-80 bentonite. J. Contam. Hydrol. 61, 329–332.

**Chisholm-Brause C J, Berg J M, Matzner R A, Morris D E, 2001.** Uranium (VI) sorption complexes on montmorillonite as a function of solution chemistry. J. Colloid Interface Sci. 233, 38–49.

Choi J-W, Jung C H, Chun C H, Park H S, Whang J H, Lee B H, 1992. Diffusion of radionuclides in compacted bentonite. Proc. Conf. on high level radioactive waste management. CONF-920430. La Grand Park, IL; Am Nucl. Soc., 2278-2283.

Choi J-W, Oscarson D W, 1996. Diffusive transport through compacted Na- and ca-bentonite, J. Contam. Hydrol. 22, 189–202.

Christiansen B, Torstenfelt B, 1988. Diffusion of nickel, strontium, iodine, cesium, and americium in loosely compacted bentonite at high pH. Radiochim. Acta, 44/45, 219–223.

**Davis J A, 2001.** Surface complexation modelling of uranium(VI) adsorption on natural mineral assemblages. NUREG/CR-6708, Office of Nuclear Regulatory Research, U.S. Nuclear Regulatory Commission, Washington, DC

**Davis J A, Kent D B, 1990.** Surface complexation modelling in aqueous geochemistry. In: Mineral-water interface geochemistry, Reviews in mineralogy, Vol. 23 (eds.: M.F. Hochella and A.F. White), Mineralogical Society of America, Washington, DC, pp. 177–260.

**Erdal B R, Wolfsberg K, Vidale R, Duffy C, Hoffman D C, 1977.** Laboratory Measurements of Radionuclide Distribution between Selected Ground Waters and Geologic Media, Pacific Northwest Laboratory Report PNL -SA-6957, 503–516

**Eriksen T E, Jacobsson A, 1981.** Ion diffusion in compacted Na and ca bentonites. KBS Technical Report 91-12, Svensk Kärnbränslehantering AB.

Eriksen T E, Jacobsson A, Pusch R, 1981. Ion diffusion through highly compacted bentonite. KBS Technical Report 81-06, Svensk Kärnbränslehantering AB.

**Eriksen T E, 1982.** Diffusion of hydrogen, hydrogen sulphide and large molecular weight anions in bentonite. KBS Technical Report 82-17, Swedish Nuclear Fuel and Waste Management Co, Stockholm

**Eriksen, T E, Jansson M, 1996.** Diffusion of I<sup>-</sup>, Cs<sup>+</sup> and Sr<sup>+</sup> in compacted bentonite - Anion exclusion and surface diffusion. SKB Technical Report 96-16, Svensk Kärnbränslehantering AB.

Fletcher P, Sposito G, 1989. The chemical modelling of clay/electrolyte interactions for montmorillonite, Clays Clay Miner., 24, 375-391.

**Gorgeon L, 1994.** Contribution à la modélisation physico-chimique de la retention de radioéléments à vie longue par des matériaux argileux. Unpublished PhD Thesis. Université Paris 6.

**Guillaumont R, Fanghänel T, Fuger J, Grenthe I, Neck V, Palmer D A, Rand M H, 2003.** Update on the chemical thermodynamics of uranium, neptunium, plutonium, americium and technetium. Chemical Thermodynamics, Vol. 4. (ed. OECD/NEA). Elsevier, Amsterdam.

Hayes K F, 1987. Equilibrium, spectroscopic, and kinetic studies of ion adsorption at the oxide/aqueous interface. Ph.D. thesis, Stanford University.

**Hedin, 2002.** Safety Assessment of a Spent Nuclear Fuel Repository: Sensitivity Analyses for Prioritisation of Research. Proceedings of the 6th International Conference on Probabilistic Safety Assessment and Management, PSAM6. Elsevier Science Ltd. (2002).

Hedin, 2003. Probabilistic dose calculations and sensitivity analyses using analytic models. *Reliability Engineering and System Safety* **79** (2003) 195-204.

**Hiemstra T, van Riemsdijk W H, 2004.** Relationship between surface structuire and ion complexation. SOPRO 2004. International workshop on sorption processes at oxide and carbonate mineral interfaces. Wissenschaftliche Berichte FZKA 6986, Forschungszentrum Karlsruhe.

**Hora S, 2002.** Expert Opinion in SR 97 and the SKI/SSI Joint Review of SR 97, SSI report 2002:20, Swedish Radiation Protection Authority.

Hora S, Jensen M, 2002. "Expert Judgement Elicitation" SSI report 2002:19, Swedish Radiation Protection Authority. SR-CAN process report

**Hummel W, Berner U, 2002.** Application of the Nagra/PSI TDB 01/01: solubility of Th, U, Np and Pu. Nagra technical Report NTB 02-12. Nagra Wettingen, Switzerland.

Hummel W, Berner U, Curti E, Pearson F J, Thoenen T, 2002. Chemical thermodynamic data base (01/01). Nagra/PSI Chemical Thermodynamic Data Base 01/01, pp 565.

Hurel C, Marmier N, Séby F, Giffaut E, Bourg A C M, Fromage F, 2002. Sorption behaviour of caesium on a bentonite sample, Radiochim. Acta 90, 695–698.

Idemitsu K, Tachi Y, Furuya H, Inagaki Y, Arima T, 1996. Diffusion of uranium in compacted bentonites in the reducing condition with corrosion products of iron. In: Murphy, W.M. and Knecht, D.A. (eds) Scientic basis for Nuclear Waste Management XIX, materials Research Society Symp. Proc., 412, 683-690.

Idemitsu K, Tachi Y, Furuya H, Inagaki Y, Arima T, 1998. Diffusion of Cs and Sr in compacted bentonites under reducing conditions and in the presence of corrosion products of iron. Mat. Res. Soc. Symp. Proc., 506, 351–358.

**Ikeda T, Amaya T, 1998.** Model Development of Chemical Evolution in Repository vol.II, Acquisition of Nuclide Migration Data in Near-Field, PNC ZJ 1281 98-003.

Kaplan D I, Serne R J, Parker K E, Kutnyakov I V, 2000. Iodide sorption to surface sediments and illitic minerals. Environmental Science and Technology 34, 399–405.

**Kato H, Muroi M, Yamada N, Ishida H, Sato H, 1995.** Estimation of effective diffusivity in compacted bentonite, XVIII International Symposium on the Scientific Basis for Nuclear Waste Management Proc., 277–284.

Kim H T, Suk T W, Parks S H, 1993. Diffusivities for ions through compacted Na-bentonite with varying dry bulk density. Waste Management, 13, 303–308.

Kozaki T, Saito N, Fujishima A, Sato S, Ohashi H, 1998. Activation energy for diffusion of chloride ions in compacted sodium montmorillonite. J. Contam. Hydrol. 35, 67–75.

**Kummert R, Stumm W, 1980.** The surface complexation of organic acids on hydrous γ-Al<sub>2</sub>O<sub>3</sub>. J. Colloid Interface Science, 75, 2, 373–385.

Laaksoharju M, Gurban I, Skårman C, 1998. Summary of hydrochemical conditions at Aberg, beberg and Ceberg. SKB TR 98-03, Svensk Kärnbränslehantering AB.

**Legoux Y, Blain G, Guillaumont R, Ouzounian G, Brillard L, Hussonnoids M, 1992.** K<sub>d</sub> measurements of activation, fission and heavy elements in water/solid phase systems. Radiochim. Acta, 58/59, 211–218.

Lemire R J, Fuger J, Nitsche H, Potter P, Rand M H, Rydberg J, Spahiu K, Sullivan J C, Ullman W J, Vitorge P, Wanner H, 2001. Chemical thermodynamics of neptunium and plutonium. Chemical Thermodynamics, Vol. 4. (ed. OECD/NEA). Elsevier, Amsterdam.

Liu Y, von Gunten H R, 1988. Migration chemistry and behaviour of ions relevant to geological disposal of radioactive wastes. A literature review with a compilation of sorption data. PSI Report 19, PSI Villigen, Switzerland.

McKinley J P, Zachara J M, Smith S C, Turner G D, 1995. The influence of uranyl hydrolysis and multiple site-binding reactions on adsorption of U(VI) to montmorillonite. Clays and Clay Min. 43, 586–596.

**Molera M, Eriksen T, 2002.** Diffusion of <sup>22</sup>Na<sup>+</sup>, <sup>85</sup>Sr<sup>2+</sup>, <sup>134</sup>Cs<sup>+</sup> and <sup>57</sup>Co<sup>2+</sup> in bentonite clay compacted to different densities: experiments and modeling. Radiochim. Acta 90, 753–760.

**Muurinen A, Penttilä-Hiltunen P, Rantanen J, 1987.** Diffusion mechanisms of strontium and cesium in compacted sodium bentonite. In Bates, J.K. and Seefeldt, W.B. (eds) Scientific basis for Nuclear Waste Management IX, MRS Symp. Proc., 50, 617–624.

**Muurinen A, Penttilä-Hiltunen P, Uusheimo K, 1989.** Diffusion of chloride and uranium in compacted sodium bentonite, In: Lutze W. and Ewing, R.C. (eds) Scientific basis for Nuclear Waste Management XII, MRS Symp. Proc., 127, 743–748.

**NEA (OECD/Nuclear Energy Agency), 2004.** Final report of Phase II of the NEA Sorption Project, by Davis, J., Ochs, M., Olin, M., Payne, T., Tweed, C., NEA, Paris (in press).

**Ochs M, Lothenbach B, Yui M, 1998.** Application of thermodynamic sorption models to distribution coefficients of radionuclides in bentonite. Mat. Res. Soc. Symp. Proc. Vol. 506, 765–772.

Ochs M, Lothenbach B, Shibata M, Yui M, 1999. Bentonite porewater chemistry, JNC TN844 99-075, JNC, Tokai-mura.

Ochs M, Lothenbach B, Wanner H, Sato H, Yui M, 2001. An integrated sorptiondiffusion model for the calculation of consistent distribution and diffusion coefficients in compacted bentonite. J. Contam. Hydrol. 47, 283–296.

**Ochs M, Lothenbach B, Shibata M, Sato H, Yui M, 2003.** Sensitivity analysis of radionuclide migration in compacted bentonite: a mechanistic model approach. J. Contam. Hydrol., 61, 313–328.

**Ochs M, Talerico C, 2003.** Prediction of K<sub>d</sub> values for anion migration. Technical Report for Mitsubishi Materials Corporation, Tokyo.

Ochs M, Lothenbach B, Shibata M, Yui M, 2004. Thermodynamic modelling and sensitivity analysis of porewater chemistry in compacted bentonite. Physics and chemistry of the Earth in press.

**Pabalan R T, Turner D R, 1997.** Uranium(6+) sorption on montmorillonite: Experimental and surface complexation modeling study. Aquatic Geochemistry 2, 203–226.

**Pabalan T, Turner D R, Bertetti F P, Prikryl J D, 1998.** UraniumVI sorption onto selected mineral surfaces. In: Adsorption of metals by geomedia. (ed. E. Jenne), Academic Press, New York, pp. 99–130.

**Peak D, Sparks D L, 2002.** Mechanisms of selenate adsorption on iron oxides and hydroxides. Environ. Sci. Technol. 36, 1460–1466.

**Pearson F J Jr, Berner U, 1991.** NAGRA thermochemical data base: I. Core data, Nagra Technial Report 91-17, Wettingen, Switzerland.

**Pearson F J Jr, Berner U, Hummel W, 1991.** NAGRA thermochemical data base: II. Supplemental data. Nagra Technial Report 91-18, Wettingen, Switzerland.

**Pleysier J, Cremers A, 1975.** Stability of silver-thiourea complexes in montmorillonite clay. J. Chem. Soc. Faraday I 71, 256–264.

Sato H, Assida T, Kohara Y, Yui M, Sasaki N, 1992. Effect of dry density on diffusion of some radionuclides in compacted sodium bentonite, J. Nuclear Sci. Technol., 29(9), 873–882.

Saha U K, Liu C, Kozak L M, Huang P M, 2004. Kinetics of selenite adsorption on hydroxyaluminum- and hydroxyaluminosilicate-montmorillonite complexes. Soil Sci. Soc. Am. J. 68, 1197–1209.

**Sato H, Ashida T, Kohara Y, Yui M, 1993.** Study on retardation mechanism of <sup>3</sup>H, <sup>99</sup>Tc, <sup>137</sup>Cs, <sup>237</sup>Np, and <sup>241</sup>Am in compacted sodium bentonite. Mater. Res. Soc. Symp. Proc. 294, 403–408.

**Sato H, Yui M, Hideki Y, 1995.** Diffusion behaviour for Se and Zr in sodium-bentonite, XVIII International Symposium on the Scientific basis for Nuclear Waste Management proc., 269–276.

Sato H, Yui M, 1997. Diffusion of Ni in compacted sodium bentonite. J. Nuclear Sci. Technol., 34, 3, 334–336.

**Sato H, 1998a.** Data setting for effective diffusion coefficients (De) of nuclides in the buffer for reference case in performance assessments of the geological disposal of high-level radioactive waste (I). PNC Technical Report TN8410 98-097.

Sato H, 1998b. Diffusion behaviour of Se(-II) and Sm(III) in compacted sodium bentonite, Radiochim. Acta 82, 173–178.

**Sato H, 1998c.** Measurements of apparent diffusion coefficients ( $D_a$ ) for Cs(I), Ni(II) and Se(IV) in bentonite with silica sand. JNC Technical Report TN8400 99-060.

**Schwyn B, 2003.** Sorption values (Kd), effective diffusion coefficients (De) and accessible porosities (e) for bentonite used in safety assessment calculations within the Project Opalinus Clay (Entsorgungsnachweis). In Near-field sorption data bases for compacted MX-80 bentonite for performance assessment of high-level radioactive waste repository in Opalinus Clay host rock. NAGRA technical report 02-18, 125–130.

Séby F, Potin-Gautier M, Giffaut E, Donard O F X, 1998. Assessing the speciation and the biochemical processes affecting the mobility of selenium from a geological repository of radioactive wastes to the biosphere. Analusis 26, 193–198.

Shibutani T, Yui M, Yoshikawa H, 1994. Sorption mechanism of Pu, Am and Se on sodium-bentonite, Mat. Res. Soc. Symp. Proc., 333, 725–730.

Silva R J, Bidoglio G, Rand M H, Robouch P B, Wanner H, Puigdomenech I, 1995. Chemical Thermodynamics of Americium. North-Holland, Amsterdam.

Sylvester E R, Hudson E A, Allen P G, 2000. The structuire of uranium (VI) sorption complexes on silica, alumina, and montmorillonite. Geochim. Cosmochim. Acta 64, 2431–2438.

**SKB, 2002.** Preliminary Safety Evaluation, Based on Initial Site Investigation Data, Planning Document. SKB TR-02-28, Svensk Kärnbränslehantering AB.

SKB, 2004. SR-Can process report, SKB report in progress, Svensk Kärnbränslehantering AB.

**SKI and SSI, 2001.** "SKI's and SSI's Joint Review of SKB's Safety Assessment Report, SR 97", Summary, SKI Report 01:3, Swedish Nuclear Power Inspectorate; SSI Report 2001:02, Swedish Radiation Protection Authority.

Stumm W, Morgan J J, 1996. Aquatic Chemistry: Chemical equilibria and rates in natural waters, 3rd ed., Wiley Interscience, New York, USA.

Tachi Y, Shibutani T, Sato H, Shibata M, 1999b. Sorption amd diffusion behavior of palladium in bentonite, granodiorite and tuff. JNC Technical Report, JNC TN8400 99-088.

Tachi Y, Shibutani T, Sato H, Yui M, 2001. Experimental and modelling studies on sorption and diffusion of radium in bentonite. J. Contam. Hydrol., 47, 171–186.

**Taki H, Hata K, 1991.** Measurement Study on Distribution Coefficient and Effective Diffusion Coefficient for Some Rocks and Bentonite JNC Technical Report, PNC ZJ 1214 91-010

**Torstenfelt B, Kipatsi H, Andersson K, Allard B, Olofsson U, 1982.** Transport of actinides through a bentonite backfill, In: Lutze, W. (eds) Scientific basis for Nuclear Waste Management, Elsevier Science Publishers B.V., Amsterdam.

**Torstenfelt B, Allard B, 1986.** Migration of fission products and actinides in compacted bentonite, SKB TR 86-14, Svensk Kärnbränslehantering AB.

**Torstenfelt B, Allard B, Andersson K, Kipatsi H, Eliasson L, Olofsson U, Persson H, 1983.** Radionuclide diffusion and mobilities in compacted bentonite. KBS Technical Report 83-34, Svensk Kärnbränslehantering AB.

**Tsukamoto M, Ohe T, Fujita T, 1994.** Diffusion of neptunium(V) in loosely compacted sodium bentonite, Radiochim. Acta, 66/67, 397–403.

**Turner G D, Zachara J M, McKinley J P, Smith S C, 1996.** Surface-charge properties and  $UO_2^{2+}$  adsorption of a subsurface smectite. Geochim. Cosmochim. Acta 60, 3399–3414.

**Turner D R, Pabalan R T, Beretetti F P, 1998.** Neptunium(V) sorption on montmorillonite: An experimental and surface complexation modelling study. Clays and Clay Minerals vol., 3, 256–269.

**Ulrich H J, Degueldre C, 1993.** The sorption of <sup>210</sup>Pb, <sup>210</sup>Bi and <sup>210</sup>Po on montmorillonite: a study with emphasis on reversibility aspects and on the radioactive decay of adsorbed nuclides, Radiochim. Acta, 62, 81-90

van Loon L R, Soler J M, Jakob A, Bradbury M H, 2003a. Effect of confining pressure on the diffusion of HTO, <sup>36</sup>Cl- and <sup>125</sup>I- in layered argillaceous rock (Opalinus Clay): Diffusion perpendicular to the fabric. Appl. Geochem., 18, 1653–1662.

**van Loon L R, Soler J M, Bradbury M H, 2003b.** Diffusion of HTO, <sup>36</sup>Cl<sup>-</sup> and <sup>125</sup>l<sup>-</sup> in Opalinus Clay samples from Mont Terri Effect of confining pressure. J. Contam. Hydrology 61, 73–83.

Waite T D, Davis J A, Fenton B R, Payne T E, 2000. Approaches to modelling uranium (VI) adsorption on natural mineral assemblages. Radiochim. Acta 88, 687–693.

Wanner H, Albinsson Y, Karnland O, Wieland E, Wersin P, Charlet L, 1994. The acid/base chemistry of montmorillonite. Radiochim. Acta 66/67, 157–162.

Wanner H, Albinsson Y, Wieland E, 1996. A thermodynamic surface model for caesium sorption on bentonite, Fresenius J. Anal. Chem., 354, 763–769.

**Wersin P, 2003.** Geochemical modelling of bentonite porewater in high-level waste repositories. J. Contam. Hydrology 61, 405–422.

Wilmot R D, Galson D A, 2000. "Expert Judgement in Performance Assessment" SKI report 2000:04, Swedish Nuclear Power Inspectorate.

Wilmot R D, Galson D A, Hora S C, 2000. "Expert Judgements in Performance Assessments. Report of an SKI/SSI Seminar" SKI report 2000:35, Swedish Nuclear Power Inspectorate.

Yu J-W, Neretnieks I, 1997. Diffusion and sorption properties of radionuclides in compacted bentonite. SKB TR 97-12. Svensk Kärnbränslehantering AB.

# Appendix A

# **Groundwater compositions**

r				
		Saline GW (I)	Non-saline GW (II)	Highly saline GW (III)
Na⁺	(mol/l)	7.390E-02	1.455E-02	5.739E-01
K	(mol/l)	3.320E-04	5.120E-05	3.320E-04
Ca <sup>+2</sup>	(mol/l)	4.120E-02	9.650E-04	4.116E-02
Mg <sup>+2</sup>	(mol/l)	4.530E-03	6.990E-04	4.530E-03
CO3 <sup>-2</sup>	(mol/l)	7.700E-04	1.985E-03	7.248E-04
H+	(mol/l)	8.589E-04	2.010E-03	7.700E-04
CI-	(mol/l)	1.557E-01	1.495E-02	6.550E-01
SO <sub>4</sub> -2	(mol/l)	3.850E-03	5.100E-04	3.850E-03
рН	(-)	7.000	7.904	7.175
pCO <sub>2</sub>	(-)	-2.558	-2.867	-2.792
Charge balance	(%)	0.995	-0.005	0.336
Precipitated solids		-	calcite	calcite
lonic strength	(mol/l)	0.2075	0.1979	0.7062

I) pH and charge balance correction

II) pH correction with addition of NaOH

III) Saline GW with the addition of 0.5 M of NaCl

		Saline GW (IV)	Saline GW (IV)
Na⁺	(mol/l)	8.3870E-02	3.7990E-01
K	(mol/l)	3.3200E-04	3.3200E-04
Ca <sup>+2</sup>	(mol/l)	4.0437E-02	2.1644E-03
Mg <sup>+2</sup>	(mol/l)	2.4003E-04	2.5217E-08
CO <sub>3</sub> <sup>-2</sup>	(mol/l)	7.1213E-06	1.0971E-04
H+	(mol/l)	-5.3007E-04	-2.1919E-01
CI-	(mol/l)	1.5570E-01	1.5570E-01
SO4 <sup>-2</sup>	(mol/l)	3.8500E-03	3.8500E-03
рН	(-)	10.499	13.154
pCO <sub>2</sub>	(-)	-9.419	-13.141
Charge balance	(%)	0.995	0.489
Precipitated solids		brucite	0.489
		calcite	brucite
			calcite
			portlandite
Ionic strength	(mol/l)	0.20223	0.23024

IV) Saline GW at high pH (addition of NaOH for pH correction)

# Appendix B

## **Bentonite characteristics**

## **Composition of bentonites**

The composition of MX-80 bentonite, Kunipia-F and Kunigel-V1 as used for the calculation of the Kd and Da values is given in Table A.1 and Table A.2.

#### Table B.1. Composition and other specific parameters of Wyoming bentonite MX-80.

Parameters	MX-80	Reference
Minerals:		
smectite	88.6% <sup>1</sup>	-
quartz	10%	/van Olphen and Fripiat, 1979/
calcite	0.7%	/Müller-Vonmoos and Kahr, 1983/
Soluble impurities:		
NaCl	0.007%	/Wanner et al, 1992/
CaSO4	0.34%	/Wanner et al, 1992/
Surface chemical parame	ters:	
CEC	85 meq/100g	/Wieland et al, 1994/
exchangeable Na⁺	81.7%	/Müller-Vonmoos and Kahr, 1983/
exchangeable K <sup>+</sup>	0.3%	/Müller-Vonmoos and Kahr, 1983/
exchangeable Mg <sup>2+</sup>	3.9%	/Müller-Vonmoos and Kahr, 1983/
exchangeable Ca2+	14.1% <sup>1</sup>	-
SOH site density	2.52×10 <sup>-5</sup> mol/g <sup>2</sup>	/Wieland et al, 1994/

<sup>1</sup> Calculated by difference

 $^{\rm 2}$  Obtained for pretreated MX-80 which had a CEC of 108 meq/100g. Adapted for the smectite content of untreated MX-80.

	Kunigel-V1	Kunipia-F
Minerals:		
smectite	46–49% <sup>1</sup>	99% /2/; >95% <sup>1</sup>
quartz	29–38% <sup>1</sup>	traces 1
feldspar	2.7–5.5% <sup>1</sup>	
calcite	2.1–2.6% <sup>1</sup>	traces 1
dolomite	2.0–3.8% <sup>1</sup>	
zeolite	3.0-3.5% <sup>1</sup>	
pyrite	0.5–0.7% <sup>1</sup>	
Soluble impurities:		
NaCl	0.001% <sup>3</sup>	0.071% <sup>3</sup>
CaSO₄	0.38% <sup>3</sup>	0.694% <sup>3</sup>
KCI	0.004% <sup>3</sup>	0.005% <sup>3</sup>
Surface sites and cation exchange par	ameters:	
SOH sites (edge sites)	1.36×10⁻⁵ mol/g ⁴	2.84×10⁻⁵ mol/g ⁴
Cation exchange capacity, CEC	60.1 meq/100g <sup>1</sup>	108 meq/100g ⁵
exchangeable Na <sup>+</sup>	85.3% <sup>6</sup>	<b>97.1%</b> ⁵
exchangeable K <sup>+</sup>	1.5% <sup>6</sup>	0.9% 5
exchangeable Mg <sup>2+</sup>	2.3% <sup>6</sup>	<b>1.3%</b> ⁵
exchangeable Ca <sup>2+</sup>	10.9% <sup>6</sup>	0.7% 5
Surface area:		
surface area of layer sites, S(layer)	810 m²/g 7	810 m²/g <sup>7</sup>
surface area of edge sites, S(edge)	3 m <sup>2</sup> /g <sup>from 4</sup>	3 m <sup>2</sup> /g <sup>from 4</sup>
<sup>1</sup> /Sasaki et al, 1995/		

#### Table B.2. Kunigel-V1 and Kunipia-F: Composition and other specific parameters relevant to ISD modeling. The references are given in the footnote.

<sup>2</sup> /Ashida et al, 1994/

<sup>3</sup> /Ochs et al, 1999/

<sup>4</sup> /Wieland et al, 1994/ obtained for pretreated MX-80 which had a CEC of 108 meq/100g, adapted for the smectite content of Kunigel-V1

#### <sup>5</sup> /Yajima, 1993/

<sup>6</sup> /Ochs et al, 1999/. Estimated based on the amount of leachable cations reported by /Sasaki et al, 1995/
<sup>7</sup> /Sato et al, 1995/

## **Porewater compositions**

This appendix includes a detailed description of calculated porewater compositions (C.1) as well as a brief discussion on potential effects of extremely saline groundwaters on porewater composition (C.2).

## C.1 Calculation of Porewater composition

- Porewater compositions were calculated using the bentonite model of /Wieland et al, 1994/.
- SOH surface sites densities were scaled by a factor 0.79 (= ratio of CEC values) to account for geochemical characteristics of the substrate MX-80 in comparison to the montmorillonite used by /Wanner et al, 1994/.
- For comparison porewater compositions were also calculated with the bentonite model of /Bradbury and Baeyens, 1997/ (data not shown).
- For systems open toCO<sub>2</sub>, the pH conditions obtained using alternatively the model by /Wieland et al, 1994/ or the model by /Bradbury and Baeyens, 1997/ are similar (ΔpH=0.02 units). In CO<sub>2</sub> open systems, the pH conditions are controlled by the carbonate buffer. In closed systems, a ΔpH≈0.4 units was observed.

Fable C-1. Calculated porewater compositions reflecting variation of groundwater chemistry (salinity) and CO $_2$	conditions.
Table C-1. Calculated porewater compositions reflecting variation of groundwater chemistry (salinity) a	nd CO <sub>2</sub>
Table C-1. Calculated porewater compositions reflecting variation of groundwater chemistry (s	alinity) a
Table C-1. Calculated porewater compositions reflecting variation of groundwater che	mistry (s
Table C-1. Calculated porewater compositions reflecting variation of groundv	vater che
Table C-1. Calculated porewater compositions reflecting variation of	ground
Table C-1. Calculated porewater compositions reflecting varia	ation of
Fable C-1. Calculated porewater compositions reflecti	ng varia
Table C-1. Calculated porewater compositions	s reflecti
<b>Table C-1. Calculated porewater</b>	composition
Table C-1. Calculated	porewater
Table C-1.	Calculated
	Table C-1.

			Variatio	n of groundwate	r salinity		
	5	Saline-GW (SGW	)	High-sa	line GW	Non-sal	ine GW
	RPWC	RPW	RPWA	HSPWC	MdSH	NSPWC	NSPW
	mol/L	mol/L	mol/L	mol/L	mol/L	mol/L	mol/L
Na⁺	2.4725E-01	2.5667E-01	2.6321E-01	5.9590E-01	6.1173E-01	1.8395E-01	1.8754E-01
¥⁺	5.2934E-04	5.5048E-04	5.6468E-04	1.1175E-03	1.1504E-03	3.9292E-04	4.0114E-04
Ca <sup>+2</sup>	1.5339E-02	1.4423E-02	1.7035E-02	4.6500E-02	5.0364E-02	1.0379E-02	1.0506E-02
Mg <sup>+2</sup>	3.9366E-03	4.0783E-03	4.2111E-03	1.2613E-02	1.3048E-02	2.6241E-03	2.6303E-03
C03 <sup>-2</sup>	1.2514E-02	1.4778E-03	5.0248E-04	7.1539E-03	8.9140E-04	1.6700E-02	1.9614E-03
Ŧ	-2.6478E-03	4.5512E-02	-6.0509E-02	-1.8369E-03	4.1198E-02	-1.6294E-03	-4.5456E-02
c	1.6035E-01	1.6035E-01	1.6035E-01	6.5965E-01	6.5965E-01	1.9595E-02	1.9595E-02
\$04 <sup>2</sup>	4.7764E-02	4.3614E-02	4.1297E-02	1.8510E-02	1.7321E-02	7.7756E-02	7.2520E-02
H <sub>2</sub> Si04 <sup>-2</sup>	1.0524E-04	1.0805E-04	1.1395E-04	1.0537E-04	1.0826E-04	1.0525E-04	1.0828E-04
SOH	8.5704E-02	8.5704E-02	8.5704E-02	8.5704E-02	8.5704E-02	8.5704E-02	8.5704E-02
LAX	3.3087E+00	3.3087E+00	3.3087E+00	3.3087E+00	3.3087E+00	3.3087E+00	3.3087E+00
Hq	6.593	7.377	7.81	6.329	7.046	6.72	7.555
pC02	-0.98	-2.6	-3.496	-1.118	-2.6	-0.95	-2.6
	closed	open	open	closed	open	closed	open
solids	quartz	quartz	quartz	quartz	quartz	quartz	quartz
	calcite	calcite	calcite	calcite	calcite	calcite	calcite
	dypsum	dypsum	gypsum	gypsum	gypsum	gypsum	gypsum
lonic strength	0.29311	0.29046	0.29159	0.74481	0.75992	0.23269	0.2211

		Variation of ben	tonite properties	£		
	No Im	No Impurities Ca-bentonite				
	RPW-NI-C	RPW-NI	RPW-Ca-C	RPW-Ca		
	mol/L	mol/L	mol/L	mol/L		
Na <sup>+</sup>	1.6469E-01	1.6461E-01	8.5803E-03	8.9927E-03		
K⁺	3.4396E-04	3.4425E-04	2.0993E-05	2.2062E-05		
Ca <sup>+2</sup>	3.8587E-03	3.8560E-03	1.4324E+00	1.0089E-01		
Mg <sup>+2</sup>	1.1500E-03	1.1475E-03	2.8445E-04	3.1249E-04		
C03 <sup>-2</sup>	-8.6071E-03	2.0846E-04	3.7754E-03	6.5913E-04		
H+	-8.3083E-03	-9.3018E-03	-4.2198E-03	-3.1236E-02		
CĽ	1.5570E-01	1.5570E-01	1.6035E-01	1.6035E-01		
\$04 <sup>-2</sup>	3.8500E-03	3.8499E-03	8.6803E-03	8.3750E-03		
H <sub>2</sub> SiO <sub>4</sub> <sup>-2</sup>	1.0494E-04	1.0494E-04	1.0477E-04	1.0493E-04		
SOH	8.5704E-02	8.5704E-02	8.5704E-02	8.5704E-02		
LAX	3.3087E+00	3.3087E+00	3.3087E+00	3.3087E+00		
pН	6.352	6.358	6.321	6.907		
pCO <sub>2</sub>	-2.029	-2.6	-1.383	-2.6		
	closed	open	closed	open		
solids	quartz	quartz	quartz	quartz		
			calcite	calcite		
2			gypsum	gypsum		
lonic strength	0.1756	0.1754	0.2634	0.28275		

Table C-2, Calculated porewater compositions reflecting variation of MX-80 bentonite properties.

Table C-3. Calculated porewater composition reflecting pH variation of the reference groundwater chemistry (RPW).

			ariaton of GW-pł		
	GW-pH = 10.5			GW-pH = 13.2	
	RPW-pH10.5-C	RPW-pH10.5-C	RPW-pH10.5	RPW-pH13.2-C	RPW-pH13.2
	mol/L	mol/L	mol/L	mol/L	mol/L
Na+	2.4802E-01	2.81E-01	2.5476E-01	3.1050E-01	3.0571E-01
Κ+	5.2928E-04	6.01E-04	5.5014E-04	6.0233E-04	5.9643E-04
Ca <sup>+2</sup>	1.5303E-02	1.75E-02	1.4387E-02	1.6222E-02	1.2998E-02
Mg <sup>+2</sup>	3.6909E-03	7.40E-03	3.8234E-03	5.4939E-06	4.9052E-03
CO <sub>3</sub> <sup>-2</sup>	1.2452E-02	1.88E-05	1.4797E-03	1.4967E-05	1.7705E-03
Ŧ	-4.1256E-03	-8.53E-02	-4.3243E-02	-9.0285E-02	-4.9235E-02
-IC	1.6035E-01	1.60E-01	1.6035E-01	1.6035E-01	1.6035E-01
SO4 <sup>-2</sup>	4.7791E-02	4.19E-02	4.3089E-02	4.5560E-02	6.3647E-02
H <sub>2</sub> SiO <sub>4</sub> <sup>-2</sup>	1.0000E-25	1.00E-25	9.9999E-26	1.0000E-25	9.9999E-26
HOS	8.5704E-02	8.57E-02	8.5704E-02	8.5704E-02	8.5705E-02
LAX	3.3087E+00	3.31E+00	3.3087E+00	3.3087E+00	3.3087E+00
Hq	6.607	9.84	7.378	11.465	7.457
pCO <sub>2</sub>	-1.008	-7.57	-2.6	-10.764	-2.6
	closed	closed	open	closed	open
solids	calcite	calcite	calcite	brucite	calcite
	gypsum	brucite	brucite	gypsum	gypsum
		gypsum		calcite	
lonic strength	0.2932	0.3039	0.2906	0.3151	0.3338
		))))))	))))		) ) ) ) ) ) ) ) ) )

## C.2 Potential influence of very high salinity on porewater composition

For the various reference groundwater conditions illustrated in Figure 4-3, as well as for further groundwater variations that include hyper-alkaline conditions, porewater compositions calculated with a thermodynamic bentonite-water interaction model are given in Appendix C.1. The highest salinity level addressed in Appendix C.1 is represented by the highly saline groundwater that leads to the HSPW porewater with an ionic strength of ca 0.8 M. In addition to these conditions, questions may arise with regard to the potential effect of extremely saline groundwater, such as salt-saturated solutions with an ionic strength of ca 6 M.

As in the case of any groundwater, the effects of extremely saline groundwaters on migration parameters for radionuclides will manifest themselves only through the resulting bentonite porewater composition. Due to a lack of experimental data and adquate calculational tools and concepts for calculating porewater composition in the presence of salt-saturated solutions, only some qualitative assessments can be made:

- When groundwaters of such high ionic strengths have to be considered, it becomes very important whether an electrostatic correction term is used for the water-bentonite edge site interaction. If yes (as in the model by /Wieland et al, 1994/), the high ionic strength will invariably lead to lower pH due to deprotonation of the edge sites, because of the surface-charge vs -potential relationships of electric double layer models (cf /Stumm and Morgan, 1996/. On the other hand, the use of a non-electrostatic model (e.g. the model by /Bradbury and Baeyens, 1997/) will in all likelihood lead to a lesser influence on porewater pH.
- On the other hand, MX-80, as well as other bentonite, contain calcite. Compacted bentonite in combination with a pCO<sub>2</sub> imposed by the host rock formation represents a system that is very well buffered with respect to pH. It is not unlikely that ionic strength effects on pH will only be significant in case of a bentonite buffer closed with respect to CO<sub>2</sub>.
- At the same time, it is not clear that the porespace of compacted bentonite can be accessed by a concentrated NaCl solution, due to the permanent negative charge of smectite. It may be speculated that some of the chloride may not able to enter, due to anion exclusion effects.
- Considering the introduction of a concentrated salt solution into the bentonite pores, it cannot be excluded that K<sub>d</sub> for most radionuclides could be affected:
  - Competition by high Na concentartions would lower the K<sub>d</sub> for Cs, Sr, Ra.
  - Complexation by high chloride concentrations would also lower K<sub>d</sub>. This could be sigificant for transition metals and Pb, but possibly also for actinides and lanthanides, in particular in combination with a lowered pH.
  - A significantly lowered pH would lead to a decrease of K<sub>d</sub> for most radionuclides.

## Thermodynamic sorption models and calculational tools

## **Calculational tools**

## The speciation code MIN\_SURF and underlying TDB

MIN SURF /Berner, 1993/ is a surface and solution chemical speciation code based on MINEQL/PSI and DSURF /cf Dzombak and Morel, 1990/. MIN SURF offers the generalized two-layer/diffuse double layer model to calculate surface complexation reactions in the solid surface/water interface, cf /Dzombak and Morel, 1990; Stumm and Morgan, 1996/. The formalisms used to calculate double layer characteristics are based on /Hunter, 1981/ and /de Levie, 1990/. The BMG-version of MIN SURF includes the possibility to model ion exchange reactions, using either the mole fraction or equivalent fraction approach /cf Sposito, 1981/ to calculate the activity of the exchanger species. The code MINEQL/PSI corresponds to the code MINEQL/EIR (EIR was the former Swiss Federal Institute for Reactor Research, now named Paul Scherrer Institute, PSI). Under the latter name, it was made available for the NEA data bank in 1986 /OECD/NEA, 1986, 1994/. This code is an extension of the well-known speciation code MINEQL /Westall et al, 1976/ and was developed at EIR as documented in two technical notes /Schweingruber, 1982, 1984/ which are reproduced in the NEA user's guide /OECD/NEA, 1986/. The extension essentially consists in the inclusion of activity coefficients and temperature dependency routines. Ionic strength corrections are done by using the well-known Davies equation.

The TDB used together with the MIN\_SURF code includes the contents of the Nagra/PSI TDB, recently updated by /Hummel et al, 2002/. The geochemical core data are from the NEA-TDB project (including the CODATA key values), or from /Nordstrom et al, 1990/.

### Bentonite-water model and sorption models used for Ni, Cs, Sr/Ra and Eu

The relevant surface complexation and ion exchange constants for these models are given in the following tables. Both the underlying bentonite-water interaction model as well as the additional ion exchange constant for Cs were taken from the work of Wanner and colleagues. Sorption modelling of Ra and Sr was implemented in the same framework by setting the respective ion exchange constants equal to that of Ca /cf Fletcher and Sposito, 1989/. In case of Ni and Eu, all constants related to the amphoteric edge sites (including surface protolysis constants) were directly taken from the work of Bradbury and Baeyens. Ion exchange was again addressed by the basic bentonite model, using consistent constants for Ni and Eu. Because the protolysis reactions used in the models of Bradbury and Baeyens would lead to a porewater pH that is slightly different from that calculated by the model of Wanner et al, (see section 5.2.5), the pH values for the reference porewaters were imposed in these calculations.

bentonite-water interaction model			
Species	log K <sup>0</sup>	Reference	
Surface site	master species:	Z (ion exchange), log K <sup>0</sup>	
ZNa	20.00	by definition	
ZH	23.00	Wieland et al., (1994)	
ZK	20.26	Fletcher & Sposito (1989)	
Z <sub>2</sub> Mg	40.46'	Wanner et al. (1992)	
Z <sub>2</sub> Ca	40.53'	Wanner et al. (1992)	
Surface site master species:		SOH (surface complexation), log $K^0$	
SOH <sup>+2</sup>	5.4 <sup>II</sup>	Wanner et al. (1994)	
SO	-6.7 <sup>11</sup>	Wanner et al. (1994)	

I) Corresponding to the equivalent fraction model

II) Corresponding to the diffuse double layer model

Ni		
Species	log K <sup>0</sup>	Reference
Surface site master species:		Z (ion exchange), log K <sup>0</sup>
ZNa ZH ZK Z <sub>2</sub> Mg Z <sub>2</sub> Ca Z <sub>2</sub> Ni	20.00 23.00 20.26 40.46' 40.53' 40.53"	by definition Wieland et al., (1994) Fletcher & Sposito (1989) Wanner et al. (1992) Wanner et al. (1992) Lothenbach & Ochs (1999)
Surface site master species: <sup>III</sup>		SOH (surface complexation), log K <sup>0</sup>
$S^{s}OH^{+2}$ $S^{s}O^{-}$ $S^{w1}OH_{2}^{+}$ $S^{w2}OH_{2}^{+}$ $S^{w2}O^{-}$ $S^{s}ONi^{+}$ $S^{w1}O^{-}u^{+}$	4.5 -7.9 4.5 -7.9 6 -10.5 -0.1	Bradbury & Baeyens (1997) Bradbury & Baeyens (1997)
S <sup>w¹</sup> ONi <sup>⁺</sup>	-3.1	Bradbury & Baeyens (1997)

I) Corresponding to the equivalent fraction model

II) Set equal to the constant for Ca, based on Fletcher & Sposito (1989) and Bradbury & Baeyens (1997)

III) Corresponding to surface complexation without electrostatic correction term

Cs		
Species	log K <sup>0</sup>	Reference
Surface site	master species:	Z (ion exchange), log K <sup>0</sup>
ZNa	20.00	by definition
ZH	23.00	Wieland et al., (1994)
ZK	20.26	Fletcher & Sposito (1989)
Z <sub>2</sub> Mg	40.46'	Wanner et al. (1992)
Z <sub>2</sub> Ca	40.53 <sup>1</sup>	Wanner et al. (1992)
ZCs	21.6 <sup>l</sup>	Wanner et al., (1996)
Surface site master species:		SOH (surface complexation), log $K^0$
SOH <sup>+2</sup>	5.4 <sup>11</sup>	Wanner et al. (1994)
SO	-6.7 <sup>11</sup>	Wanner et al. (1994)

I) Corresponding to the equivalent fraction modelII) Corresponding to the diffuse double layer model

Eu		
Species	log K <sup>0</sup>	Reference
Surface site master species:		Z (ion exchange), log K <sup>0</sup>
ZNa	20.00	by definition
ZH	23.00	Wieland et al., (1994)
ZK	20.26	Fletcher & Sposito (1989)
Z <sub>2</sub> Mg	40.46'	Wanner et al., (1992)
Z₂Ca	40.53'	Wanner et al., (1992)
Z₃Eu	64.6'	Wang et al., (1998)
Surface site master species: <sup>II</sup>		SOH (surface complexation), log $K^0$
S <sup>s</sup> OH <sup>+2</sup>	4.5	Bradbury & Baeyens (1997)
S <sup>s</sup> O <sup>-</sup>	-7.9	Bradbury & Baeyens (1997)
$S^{w1}OH_2^+$	4.5	Bradbury & Baeyens (1997)
$S^{w1}O^{-}$	-7.9	Bradbury & Baeyens (1997)
$S^{w2}OH_{2}^{+}$	6	Bradbury & Baeyens (1997)
S <sup>w2</sup> O <sup>-</sup>	-10.5	Bradbury & Baeyens (1997)
$S^{\circ}OEu^{+2} + H^{+}$	1.8	Bradbury & Baeyens (2002)
$S^{\circ}OEuOH^{+} + 2H^{+}$	-5.4	Bradbury & Baeyens (2002)
$S^{\circ}OEu(OH)_{3}^{-} + 2H^{+}$	-22.1	Bradbury & Baeyens (2002)
$S^{w1}OEu^{+2} + H^{+}$	-0.5	Bradbury & Baevens (2002)

I) Corresponding to the equivalent fraction modelII) Corresponding to surface complexation without electrostatic correction term

# Appendix E

## Data derivation tables for K<sub>d</sub> values

Explanations are provided in the following two tables, see section 5.3.5 for details.

List of evluated radionuclides:

- Americium
- Ceasium
- Carbon: no data derivation table given
- Cerium, holmium, and samarium: K<sub>d</sub> derived using Eu as analogue element
- Chloride: no data derivation table given
- Curium: no data derivation table given
- Iodine: no data derivation table given
- Lead
- Neptumium
- Nickel
- Niobium: no data derivation table given
- Palladium: no data derivation table given
- Plutonium
- Protactinium: no data derivation table given
- Radium and strontium
- Radon: no data derivation table given
- Selenium
- Silver: no data derivation table given
- Technetium
- Thorium
- Tin: no data derivation table given
- Uranium
- Zirconium
|   | CEC of the                          | annlication and       | CEC of the             | source material.   | If for both | systems the | substrates are                 |                         | CECT                    |                    |                         |                                   | if source data       | are sorprion<br>edae data →   | pH CF for each | pH condition is        | CF = 1                         |                               |                         |                         |                               |                               |                                |                               |                         |                          |             |                               |                               |                                |                               |                  |                    |             |                               |                               |                                |
|---|-------------------------------------|-----------------------|------------------------|--------------------|-------------|-------------|--------------------------------|-------------------------|-------------------------|--------------------|-------------------------|-----------------------------------|----------------------|-------------------------------|----------------|------------------------|--------------------------------|-------------------------------|-------------------------|-------------------------|-------------------------------|-------------------------------|--------------------------------|-------------------------------|-------------------------|--------------------------|-------------|-------------------------------|-------------------------------|--------------------------------|-------------------------------|------------------|--------------------|-------------|-------------------------------|-------------------------------|--------------------------------|
| _ | tainty (U)                          |                       | MASH                   | <u> </u>           | 11/B1       |             | F3/C3                          | L1'L3                   |                         |                    | 15/B5                   | 16/B6                             | L4:L5                | L4:L6                         |                |                        |                                | 1.6                           | 2.0                     |                         |                               |                               |                                | 1.6                           | 4.1                     | 2.0                      | L17.L18.L19 |                               |                               |                                | 1.6                           | 1.4              | 2.0                | L24'L25'L26 |                               |                               |                                |
| ¥ | on (C) / Uncer<br>Factor(F)         | ц.                    | RPWC                   | <u> Тих-80</u> -   | H1/B1       |             | E3/C3                          | K1 <sup>.</sup> K3      |                         |                    | H5/B5                   | H6/B6                             | K4:K5                | K4:K6                         | Ę              |                        |                                | 1.6                           | 2.0                     | K11:K12                 |                               |                               |                                | 1.6                           | 4.1                     | 2.0                      | K17'K18'K19 |                               |                               |                                | 1.6                           | 1.4              | 2.0                | K24 K25 K26 |                               |                               |                                |
| 7 | Conversi                            |                       | RPW                    | <u>г – мх-80</u> – | G1/B1       |             | D3/C3                          | J1:J3                   |                         |                    | G5/B5                   | G6/B6                             | J4:J5                | J4:J6                         |                |                        |                                |                               | 2.0                     |                         |                               |                               |                                | 1.6                           | 1.4                     | 2.0                      | J17.J18.J19 |                               |                               |                                | 1.6                           | 4.1.4            | 2.0                | J24:J25:J26 |                               |                               |                                |
| _ |                                     |                       | MSH                    | MX-80              | 85.32       | 7.046       |                                |                         |                         |                    | 0.0040                  | 1.0000                            |                      |                               |                |                        | B9:L4                          |                               | actors                  |                         | G10'L13                       | G10/L13                       | B9:L7                          |                               | actors                  |                          |             | G16:L20                       | G16/L20                       | B9'L8                          |                               | factors          |                    |             | G23'L27                       | G23/L27                       |                                |
| т |                                     | Application           | RPWC                   | MX-80              | 85.32       | 6.59        |                                |                         |                         |                    | 0.0003                  | 0.9996                            |                      |                               |                |                        | B9 <sup>.</sup> K4             |                               | uncertainty f           |                         | G10 <sup>.</sup> K13          | G10/K13                       | B9 <sup>.</sup> K7             |                               | uncertainty f           |                          |             | G16K20                        | G16/K20                       | B9.K8                          |                               | uncertainty 1    |                    |             | G23 <sup>.</sup> K27          | G23/K27                       |                                |
| G |                                     |                       | RPW                    | MX-80              | 85.32       | 7.377       | _                              |                         |                         |                    | 0.0018                  | у 1.0000                          |                      |                               |                |                        | B9:J4                          |                               | specific                |                         | G10:J13                       | G10/J13                       | B9:J7                          |                               | specific                |                          |             | G16:J20                       | G16/J20                       | 86:68                          |                               | specific         |                    |             | G23:J27                       | G23/J27                       |                                |
| Ľ | ions                                |                       |                        | SWy-1              | 87          | 7.046       | 224                            |                         | т                       |                    |                         |                                   |                      | ved RN                        | T              | ption                  |                                |                               |                         |                         |                               |                               |                                |                               |                         |                          |             |                               |                               | L                              |                               |                  |                    |             |                               |                               |                                |
| ш | Conditi                             | formation (b          | NaCIO4                 | SWy-1              | 87          | 6.59        | 224                            | /                       | oplication pl           | d from the         | ion-edge                |                                   |                      | iculateu sor<br>total dissolv | application    | exes as sor            | (sə)                           |                               |                         | alculated so            | cotal dissol<br>application   | mplexes)                      |                                |                               |                         |                          |             |                               |                               |                                |                               |                  |                    |             |                               |                               |                                |
| ۵ |                                     | Additional in         | 0.1 M                  | SWy-1              | 87          | 7.377       | 355                            |                         | ies for the al          | ons obtaine        | nental sorpt            |                                   |                      | ion and the                   | ration in the  | (CO <sub>3</sub> compl | tive compley                   |                               | ;                       | tween the cities of the | tration and the               | s (no CO <sub>3</sub> co      |                                | - Pariat                      |                         |                          |             |                               |                               |                                |                               |                  |                    |             |                               |                               |                                |
| ပ |                                     |                       |                        | SWy-1              | 87          | 7.6         | 355                            |                         | K <sub>d</sub> valu     | conditi            | experir                 |                                   | 101                  |                               | T concent      | systems                | competi                        |                               |                         |                         |                               | system                        |                                |                               | N <sub>d</sub> value op | esponaing                |             |                               |                               |                                |                               |                  |                    |             |                               |                               |                                |
| 8 |                                     | Data source (a)       | SBPW                   | - <u>- MX-80</u>   |             | 9.2         |                                |                         |                         |                    | 0.0087                  | 1.0000                            |                      |                               |                |                        |                                |                               |                         |                         |                               |                               |                                |                               | experimental            | irom me corr<br>isotherm |             |                               |                               |                                |                               |                  |                    |             |                               |                               | 2-18)                          |
| ۷ | e data                              | from                  | ∐⁄<br>I                | 4                  | (meq/100g)  |             |                                | ulated                  | d the                   |                    | urce L                  | s as                              |                      |                               | F              |                        | alculated                      | and the                       |                         | source                  | (eavaidii                     |                               | (F/d)                          |                               |                         |                          |             | (F/d)                         | (F/d)                         | (6/T)                          |                               |                  |                    |             | (lL/g)                        | (L/g)                         | AGRA NRB 02                    |
|   | K <sub>d</sub> value for the source | PH condition obtained | the experimental sorpt | edge               |             |             | CIO4 (pH conversion) (b)       | ratio between the calcu | sorbing RN-fraction and | total dissolved RN | concentration in the so | system (CO <sub>3</sub> complexe: | sorption competitive | complexes)                    |                | P                      | F-tq ratio between the ca      | sorbing RN-fraction           | Ipad total dissolved KN | concentration in the    |                               |                               | F-total 2)                     |                               |                         | npacted                  |             |                               |                               | F-total 3)                     |                               |                  | npacted            |             |                               |                               | n Th(IV) sorption isotherm (N  |
|   | Th(IV)                              |                       | Solution               | Substrate          | 1 CEC       | 2 pH        | 3 K <sub>d</sub> for 0.1 M NaC | 4 CF-total 1            |                         | Speciation         | 5 CF-spec 2             | 6 CF-spec 3                       | 7 CF-total 2         | 8 CF-total 3                  |                | 9 Experimental K       | 10 Predicted K <sub>d</sub> (C | 11 UF-starting K <sub>d</sub> | 12 UF-batch -> com      | 13 UF-total             | 14 K <sub>d</sub> upper limit | 15 K <sub>d</sub> lower limit | 16 Predicted K <sub>d</sub> (C | 17 UF-starting K <sub>d</sub> | 18 UF-speciation        | 19 UF-batch -> com       | 20 UF-total | 21 K <sub>d</sub> upper limit | 22 K <sub>d</sub> lower limit | 23 Predicted K <sub>d</sub> (C | 24 UF-starting K <sub>d</sub> | 25 UF-speciation | 26 UF-batch -> com | 27 UF-total | 28 K <sub>d</sub> upper limit | 29 K <sub>d</sub> lower limit | (a): K <sub>d</sub> value fron |

(b): K<sup>4</sup> values from Th(IV) sorption edge (NAGRA NRB 02-18) SBPW: Synthetic bentonite porewater used in the sorption of Th on MX-80 (Bradbury&Baeyens, 2003) RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) HSPW: High saline porewater (Beberg water)

Control         Control         Control         Control         Control         Control           State         BSP         BSP<		A	В	ပ	D	ш	L	ŋ	н	_	ſ	У	-	Σ	z	0
Subtraction         Data source (a) (b) (b) (b) (b) (b) (b) (b) (b) (b) (b)	Tc(IV) analogue to Th(IV)						Ŭ	onditions						Conversi	on (C) / Uncer Factor(F)	tainty (U)
Statistic         Staty         RFWC         NEW         Col         NEW         RFW         Col         NEW         RFW         RFW <t< th=""><th></th><th></th><th>Data source (a)</th><th>Additi</th><th>onal informati</th><th>on (b)</th><th></th><th>Additional in</th><th>formation (c)</th><th></th><th></th><th>Application</th><th></th><th></th><th>СF</th><th></th></t<>			Data source (a)	Additi	onal informati	on (b)		Additional in	formation (c)			Application			СF	
Statement         Statement <t< th=""><th>Solution</th><th></th><th>SBPW</th><th>RPW</th><th>RPWC</th><th>MASH</th><th></th><th>0.1 M P</th><th>VaCIO4</th><th></th><th>RPW</th><th>RPWC</th><th>HSPW</th><th>RPW</th><th>RPWC</th><th>MASH</th></t<>	Solution		SBPW	RPW	RPWC	MASH		0.1 M P	VaCIO4		RPW	RPWC	HSPW	RPW	RPWC	MASH
1         0	Substrate		MX-80	MX-80	MX-80	MX-80	SWy-1	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
2         PH         7.37         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.377         6.59         7.046         7.047         7.0	1 cec	(meq/100g)	79	85.32	85.32	85.32	87	87	87	87	85.32	85.32	85.32	J1/B1	K1/B1	L1/B1
3 k, for 0.1 MucDC, (F) contraction         63         35.         2.4.         2.4.         2.4.         2.4.         2.4.           6 F-tatal 1         (Thio Downerm the calculated softing anaboure RM-fraction and the out discound fast and the out discound fa	2 pH		7.6	7.377	6.59	7.046	7.6	7.377	6.59	7.046	7.377	6.59	7.046			
4 CF start 1     With Demonstration and the orbit of assolved the total discoved text.     Million       5 Face at 0     the total discoved text.     0.0003     0	3 K <sub>d</sub> for 0.1 M NaClO <sub>4</sub> (pH conversion) (c		63				355	355	224	224				G3/F3	H3/F3	13/H3
Sected         Story of an angle RM, mode R	4 CF-total 1 ratio betwee	n the calculated												M1.M3	N1 <sup>.</sup> N3	01:03
Stream     Interaction     Interaction     Interaction     Interaction     Interaction     Interaction     Interaction     Interaction       6 Ferera 2 (b)     system (C0, complexes as system (C0, complexes as reprior compared     10000     00040     100000	sorbing anal	ogue RN-fractio	on and				ratio be	tween the c	alculated so	rbing						
5         Cr-sterz 2(b)         concentration in the source         10000	Speciation the total diss	solved analogue	e RN				RN-frac	tion and the	total disso	ved RN						
6         Created 2         system (C0, complexes as sorption         100000         10000	5 CF-spec 2 (b) concentratio	n in the source		0.0018	0.0003	0.0040	concen	tration in th	e applicatio		1.0000	0.9998	1.0000	J5/C5	K5/D5	L5/E5
7         CF-total 2         Sorption competitive complexes)         M446           6 F-total 3         CF-total 3         CF-total 3         M446           8 CF-total 3         CF-total 3         CF-total 4         M446           9 Experimental K <sub>1</sub> (a)         (Ug)         600         Factorian and the calculated         Relation and the calculated achieven the calculated         M446           9 Experimental K <sub>1</sub> (a)         (Ug)         600         Factorian and the calculated         Relation and the calculated achieven the calculated         Relation and the calculated achieven the calculated         M446           10 F-tatal + competition         (Ug)         concentration in the splitication         J104413         L10013         L1           11 F-stating K <sub>1</sub> (Ug)         concentration in the splitication         J104413         L10013         L1           15 K <sub>1</sub> cover limit         (Ug)         concentration in the splitication         J104413         L10013         L1           16 Factorial 2         (Ug)         Complexes         Seconplexes         Seconplexes         J104413         L10013         L1           16 Factorial 2         (Ug)         Complexes         Seconplexes         J104413         L10013         L1         L1           16 Factorial 2         (Ug)	6 CF-spec 3 (b) system (CO <sub>3</sub>	complexes as		1.0000	0.9996	1.0000	system	s (CO <sub>3</sub> com	olexes as so	rption	1.0000	- 1.0000	1.0000	I J6/C6	K6/D6	16/E6
B Creard 3         Creard 3         Manue           9 Experiments/(a)         (Lg)         softing analogue RN-fraction and ratio between the calculated of reaction softing analogue RN-fraction and 10 PredicierK (creatar 1)         (Lg)         Ratio between the calculated contentiation in the total disstored analogue RN-fraction and RN concentration in the total disstored system (to CO, complexes)         Manue         BOVA         BOVA<	7 CF-total 2 sorption con	npetitive compl	exes)				compet	itive comple	(səxe	7				M4:M5	N4'N5	04:05
9         Experimental K, (a)         (ug)         63.00         ratio between the calculated sorbing         RM. faction and the total dissolved analogue RM. faction and the total dissolved         B9.44         B9.04         16           10         Frading K_G         (ug)         (ug)         (for total dissolved analogue RM. faction         RN. concentration         B9.44         B9.04         16           11         U-stanting K_G         (ug)         (for total dissolved analogue RM. faction         RN. concentration         B9.44         B9.04         16           12         U-batch - scompacted         (ug)         (for total dissolved analogue RM. faction and the total dissolved         B9.44         B9.04         16           13         (Fradat)         (ug)         (ug)         (for contant dissolved analogue RM. faction and the total dissolved         B9.44         B9.04         16           16         Fradation         (ug)         (uo)         (for contant dissolved analogue RM. faction and the total dissolved         B9.44         B9.04         16           16         Fradation         (uo)         (for contant dissolved analogue RM. faction and the total dissolved         B9.44         B9.04         16           17         Ur-stanting K_G         (for contant dissolved analogue RM. faction and the total dissolved         B9.44	8 CF-total 3									$\mathbb{N}$				M4·M6	N4 <sup>.</sup> N6	04:06
Between the calculated soching         (U)         63300         sorting analogue RNTraction and classified         retio between the calculated soching         B9144         B9104         116           10         Fradetion (T)         (U)         system (no C0, complexes)				<b>Zratio betwee</b>	en the calculation	ated									!	
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	9 Experimental K <sub>d</sub> (a)	(6/1)	63.00	sorbing <u>ana</u>	logue RN-fra	iction and	ratio be	tween the c	alculated sc	rbing					Ч	
11         UF-starting K <sub>4</sub> concentration in the source         RN concentration in the application         16           12         UF-batch > competited         system (no CO <sub>3</sub> complexes)         system (no CO <sub>3</sub> complexes)         100013         K10N13         L10013         20           14         K <sub>1</sub> over limit         (Ug)         10         J10M13         K10N13         L10013         20           15         K <sub>1</sub> over limit         (Ug)         10         J10M13         K10N13         L10013         114           16         Predicted K <sub>1</sub> (F-total 2)         (Ug)         1         17         114         20         114           16         Federation         (Ug)         1         10         10         10         114         20         114           16         Federation         (Ug)         1         10         114         20         114           16         Federation         (Ug)         1         1         20         20         20           17         Federation         (Ug)         1         1         20         20         20           16         Federation         (Ug)         1         1         20         20         20	10 Predicted K <sub>d</sub> (CF-total 1)	(F/d)		the total dis	solved analc	gue RN	RN-frac	tion and the	total disso	ved	B9 M4	B9·N4	B9:04			
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	11 UF-starting K <sub>d</sub>			concentratio	on in the sou	LCe	RN con	centration i	n the applic:	ation				1.6	1.6	1.6
	12 UF-batch -> compacted			system (no	co <sub>3</sub> complex	(se)	system	s (ດດ <sup>3</sup> com	DIEXES AS SO	rption				2.0	2.0	2.0
14 $k_1$ upper limit $(10)$ $(10)$ $(10)$ $(10013)$ <	13 UF-total			J			compet		(sex)					M11 <sup>.</sup> M12	N11 <sup>.</sup> N12	011.012
15 $(10)$	14 K <sub>d</sub> upper limit	(6/1)									J10.M13	K10 <sup>.</sup> N13	L10 <sup>.013</sup>			
Interface         <	15 K <sub>d</sub> lower limit	(F/d)									J10/M13	K10/N13	L10/013			
17 $U^{-3}$ stating $K_{3}$ 1 $U^{-3}$ stating $K_{3}$ $U^{-3}$	16 Predicted K <sub>d</sub> (CF-total 2)	(F/d)									7M.68	B9.N7	B9.07			
I8         UF-speciation         I	17 UF-starting K <sub>d</sub>													1.6	1.6	1.6
13         UF-batch-3 compacted         1         1         2         0         2         0         1	18 UF-speciation													1.4	1.4	1.4
20         UF-lotat         (L-0tat)	19 UF-batch -> compacted													2.0	2.0	2.0
21 $(u_{0}per limit)$ $(u_{0})$ <th>20 UF-total</th> <th></th> <th>J17<sup>.</sup>J18<sup>.</sup>J19</th> <th>K17<sup>.</sup>K18<sup>.</sup>K19</th> <th>L17L18L19</th>	20 UF-total													J17 <sup>.</sup> J18 <sup>.</sup> J19	K17 <sup>.</sup> K18 <sup>.</sup> K19	L17L18L19
	<b>21</b> K <sub>d</sub> upper limit	(F/d)									J16:M20	K16 <sup>.</sup> N20	L16 <sup>.</sup> 020			
33         Predicted K <sub>4</sub> (CF total 3)         (Lg)         (Lg)         B908	22 K <sub>d</sub> lower limit	(F/d)									J16'M20	K16'N20	L16 <sup>(</sup> 020			
24 UF-starting K <sub>4</sub> 24 UF-starting K <sub>4</sub> 16       16         25 UF-starting K <sub>4</sub> 25 UF-starting K <sub>4</sub> 16       14         26 UF-batch->compacted       1       16       14         26 UF-batch->compacted       1       16       14         26 UF-batch->compacted       1       16       16         27 UF-total       1       16       16       16         27 UF-total       10       1       123027       123027       123027         28 Kupper limit       10       1       123027       123027       123027       123027	23 Predicted K <sub>d</sub> (CF-total 3)	(F/d)									8W.68	B9 <sup>.</sup> N8	80.6 <b>8</b>			
25         UF-speciation         1         1.4           26         UF-batch->compacted         1         1         1.4           26         UF-batch->compacted         1         1         2.0           27         UF-total         1         1         2.0         2.0           27         UF-total         1         1         2.0         2.0         2.0           28         Varper limit         1         1         2.0         2.0         2.0         2.0         2.0           28         Varper limit         1         1         1         1         2.0 <td>24 UF-starting K<sub>d</sub></td> <td></td> <td>1.6</td> <td>1.6</td> <td>1.6</td>	24 UF-starting K <sub>d</sub>													1.6	1.6	1.6
26         UF-batch -> compacted         Image: Compacted	25 UF-speciation													1.4	1.4	1.4
27         UF-total         27         UF-total           28         K4 upper limit         U_0)         U_0         U_24.025.02t           28         K4 upper limit         U_10)         U_23.027         L23.027           29         K4 lower limit         U_10)         U_23.027         L23.027	26 UF-batch -> compacted													2.0	2.0	2.0
28 K <sub>d</sub> upper limit (Ug) (Ug) 28 K <sub>d</sub> upper limit (Ug) 29 K <sub>d</sub> lower limit (Ug) 29 K <sub>d</sub> lower limit (Lg) 29 K <sub>d</sub> lower li	27 UF-total	_												J24 <sup>.</sup> J25 <sup>.</sup> J26	K24'K25'K26	L24:L25:L26
29 (K, lower limit 1/g) 23/M27 (23/M27 1/23/027 1/23/027 1/23/027 1/23/027 1/23/027 1/23/027 1/23/027 1/23/027	28 K <sub>d</sub> upper limit	(6/T)									J23:M27	K23'N27	L23 <sup>.</sup> 027			
	29 K <sub>d</sub> lower limit	(F/d)									J23/M27	K23/N27	L23/027			

(a): Data from Nagra NTB 02-18 for Th
 (b): Th speciation for MX-80 application
 (b): Th speciation from Th (Bradbury & Bayens, unpublished data in Nagra NRB 02-18)
 (c): pH conversion from Th (Bradbury & Beberg water)
 RPWC: Reference porewater (saline Beberg water)
 RPWC: High saline porewater (Beberg water)

Predicted K<sub>d</sub> values for Am(III). Data source: Experimental K<sub>d</sub> values for Am(III) sorption on smectite (speciation calculated with TDB of Silva et al., 1995)

Am(III)				Cond	litions				Factor(F)	ומווויא (ה)
			Data source (	a)		Application			ц.	
Solution			0.1 M NaCIO	4	RPW	RPWC	MSPW	RPW	RPWC	MASH
Substrate		Smectite	Smectite	Smectite	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	62	62	62	85.32	85.32	85.32	1.08	1.08	1.08
Hd		7.377	6.593	7.046	7.377	6.59	7.046	-	-	-
pCO2		-3.5	-3.5	-3.5	-2.6	-0.98	-2.6			
CF-total 1								1.080	1.080	1.080
Speciation										
CF-spec. 2 (b)		0.792	0.979	0.906	0.505	0.0863	0.334	0.637	0.088	0.368
CF-spec. 3 (b)		1.000	1.000	1.000	0.893	0.5060	0.567	0.893	0.506	0.567
CF-total 2 (b)								0.688	0.095	0.398
CF-total 3 (b)								0.964	0.546	0.613
									UF	
Experimental K <sub>d</sub>	(F/d)	63.10	19.95	39.81						
Predicted K <sub>d</sub> (CF-total 1)	(F/d)				68.14	21.55	43.00			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								4.4	4.4	4.4
K <sub>d</sub> upper limit	(IC/d)				302.4	95.6	190.8			
K <sub>d</sub> lower limit	(L/g)				15.36	4.86	9.69			
Predicted K <sub>d</sub> (CF-total 2) (b)	(F/d)				43.427	1.899	15.834			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	4.1	1. 4
UF-batch -> compacted								2:0	2.0	2.0
UF-total					200 000	202 77	00 275	6.2	9.7	6.2
					CU0.802	11.191	30.373			
K <sub>d</sub> lower limit	(F/d)				6.990	0.306	2.549			
Predicted K <sub>d</sub> (CF-total 3) (b)	(F/d)				60.827	10.904	24.387			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	4.1	4.
UF-batch -> compacted								2.0	2.0	2.0
UF-total								6.2	6.2	6.2
Kd upper limit	(F/d)				377.902	67.743	151.514			
K. Iower limit	(F/d)				9.791	1.755	3.925			

(a): Na-smectite at 0.1 M NaClO4 (Gorgeon, 1994). Speciation calculated with TDB of Silva et al., (1995)
(b): Speciation calculated with TDB of Silva et al., (1995)
(c): Speciation calculated with TDB of Guillaumont et al., (2003)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system
HSPW: High saline porewater (Beberg water)

edicted K <sub>d</sub> values for Ar illaumont et al., 2003)	m(III). D	ata sourc	e: Exper	imental I	X <sub>d</sub> value:	s for Am	(III) sorp	tion on s	mectite (	speciatic	n ca
				Cond	itions			Conversi	on (C) / Uncer Factor(F)	tainty (U)	
			Data source (a			Application			Ъ		
ution			0.1 M NaCIO		RPW	RPWC	MdSH	RPW	RPWC	HSPW	
strate		Smectite	Smectite	Smectite	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	
~	(meq/100g)	62	62	62	85.32	85.32	85.32	1.08	1.08	1.08	
		7.377	6.593	7.046	7.377	6.59	7.046	-	-	-	
02		-3.5	-3.5	-3.5	-2.6	-0.98	-2.6				
total 1								1.080	1.080	1.080	

0.360 0.861

0.072 0.892

0.242 0.883

0.267 0.861

0.0684 0.8915

0.107 0.883

0.742 1.000

0.949 1.000

0.441 1.000

Speciation CF-spec. 2 (c) CF-spec. 3 (c)

CF-total 2 (c) CF-total 3 (c)

0.389 0.930

0.078 0.963

0.262 0.954

Ь

1.6 2.0 4.4 4.4

1.6 1.4 2.0 4.4

1.6 2.0 4.4

43.00

21.55

68.14

39.81

19.95

63.10

(l/g)

(IL/g)

Predicted K<sub>d</sub> (CF-total 1)

UF-starting K<sub>d</sub>

Experimental K<sub>d</sub>

190.8 9.69

95.6

302.4

4.86

15.36

(L/g)

(L/g)

Predicted K<sub>d</sub> (CF-total 2) (c)

K<sub>d</sub> lower limit

UF-starting  $K_d$ 

UF-CEC UF-speciation UF-batch -> compacted UF-total

K<sub>d</sub> upper limit

K<sub>d</sub> lower limit

15.491

1.553

16.511

1.6 1.4 6.2 6.2

1.6 1.4 1.4 6.2 6.2

1.6 1.4 6.2 6.2

96.245

2.493

9.647 0.250

102.576

2.658

(b/l) (b/l) (l/g)

Predicted K<sub>d</sub> (CF-total 3) (c)

UF-starting  $K_d$ 

UF-CEC

UF-speciation UF-batch -> compacted

K<sub>d</sub> upper limit

UF-total

K<sub>d</sub> lower limit

37.022

19.211

60.202

1.6 1.4 6.2 6.2

1.6 1.4 6.2 6.2

1.6 1.4 6.2 6.2

230.011

119.357

374.022

5.959

3.092

9.690

lculated with TDB of Pre Gu Solu Solu PPH CEC CEC

(a): Na-smectite at 0.1 M NaClO4 (Gorgeon, 1994). Speciation calculated with TDB of Guillaumont et al., (2003)
(b): Speciation calculated with TDB of Silva et al., (1995)
(c): Speciation calculated with TDB of Guillaumont et al., (2003)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system
HSPW: High saline porewater (Beberg water)

(F/g)

UF-CEC UF-batch -> compacted UF-total K<sub>d</sub> upper limit

FIGUICIEU Nd Values 101 US	(I). V alues cal	iculated Dase		er uy wanner	cr al. (13	(06	
Cs(I)					Co Uncerta	inversion inities (U)	(C) / Factor(F)
Solution			Application			СF	
Substrate		RPW	RPWC	HSPW	RPW	RPWC	MdSH
Material		MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	85.32	85.32	85.32			
Hd		7.377	6.593	7.046			
	(W)	0.32	0.29	0.76			
pCO <sub>2</sub>	(atm)	-2.6	-0.98	-2.6			
						ЧF	
Calculated Kd (b)	(F/d)	0.1050	0.1030	0.0348			
UF-model					3.0	3.0	3.0
UF-batch -> compacted					2.0	2.0	2.0
UF-total					6.0	6.0	6.0
Kd upper limit	(L/g)	0.6300	0.6180	0.2088			
Kd lower limit	(L/g)	0.0175	0.0172	0.0058			
positive error bar		0.5250	0.5150	0.1740			
negative error bar		0.0875	0.0858	0.0290			
(a): Data-Source for Sr from (	NAGRA NRTB	02-18 and rei	ference therein				

Predicted K<sup>a</sup> values for Cs(D Values calculated based on the model by Wanner et al. (1996)

(b): For starting condition Kd value calculated for MX-80 and reference porewater (NAGRA NRTB 02-18) RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg/sea water)

Predicted K <sub>d</sub> values f	or Cs(I).	Data source: E	Experimen	tal K <sub>d</sub> val	ues for C	s(I) sor	ption o	n MX-8
Cs(II)			Conditio	su		Col	inty (U) F	C) / actor(F)
		Data source (a)		Application			СF	
Solution		SBPW	RPW	RPWC	MdSH	RPW	RPWC	MdSH
Substrate		MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	79	85.32	85.32	85.32	-	-	-
Н		7.6	7.377	6.59	7.046			
pCO2		-3.5	-2.6	-0.98	-2.6			
Na dissolved			0.2518	0.2518	0.6118			
Sum of cations	(mol/L)	6.77E-01	0.2942	0.2863	0.7397	0.43	0.42	1.09
CF-total 1						0.43	0.42	1.09
Sneciation								
CsCI/Cs tot		0.21869	0.06492	0.06488	0.23301			
CF-spec						e	ო	-
CF-total 2						1.46	1.43	1.03
							1	
Experimental Kd	(IT/a)	0.050					5	
Brodictod Kd (CE total 1)			0 033	1000	0 DEE			
Fredicted Ad (CF-total 1)	(LUU)		770.0	170.0	ccn.n	•		1
UF-starting Kd							1.6	1.0
UF-cations conversion						2:2	Q.7	C.2
UF-batch -> compacted						2.0	2.0	2.0
UF-total						8.0	8.0	8.0
Kd upper limit	(b/l)		0.173	0.169	0.436			
Kd lower limit	(L/g)		0.003	0.003	0.007			
positive error bar			0.152	0.148	0.381			
negative error bar			0.019	0.019	0.048			
Predicted Kd (CF-total 2)	(L/g)		0.073	0.071	0.051			
UF-starting Kd						1.6	1.6	1.6
UF-cations conversion						2.5	2.5	2.5
UF-speciation						1.4	1.4	1.4
UF-batch -> compacted						2.0	2.0	2.0
UF-total						11.1	11.1	11.1
Kd upper limit	(L/g)		0.818	0.796	0.573			
Kd lower limit	(L/g)		0.007	0.006	0.005			
positive error bar			0.744	0.725	0.522			
negative error bar			0.067	0.065	0.047			
Calculated Kd	(L/a)		0.105	0.103	0.0348			

(a): Kd values from NAGRA NRB 02-18 SBPW: Synthetic bentonite porewater used in the sorption of Cs on MX-80 (Bradbury&Baeyens, 2003) RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

125

$\widetilde{\mathbf{x}}$
Ū.
$\mathbf{x}$
$\geq$
Ξ
9
E
.Э
5
E E
0
01
$\overline{\mathbf{I}}$
Π
r Ti
OI
£
Ś
e
1
'n
$\geq$
73
$\mathbf{X}$
Ц
5
E.
B
Ξ.
Ξ.
ğ
5
б
<u> </u>
e.
~ `
5
urc
ourc
sourc
ta sourc
ata sourc
Data sourc
. Data sourc
I). Data sourc
III). Data sourc
i(III). Data sourc
Eu(III). Data sourc
Eu(III). Data sourc
or Eu(III). Data sourc
for Eu(III). Data sourc
s for Eu(III). Data sourc
tes for Eu(III). Data source
lues for Eu(III). Data sourc
alues for Eu(III). Data sourc
values for Eu(III). Data sourc
d values for Eu(III). Data sourc
Kd values for Eu(III). Data sourc
d K <sub>d</sub> values for Eu(III). Data sourc
ed K <sub>d</sub> values for Eu(III). Data sourc
cted K <sub>d</sub> values for Eu(III). Data sourc
licted K <sub>d</sub> values for Eu(III). Data sourc
edicted K <sub>d</sub> values for Eu(III). Data sourc
redicted K <sub>d</sub> values for Eu(III). Data sourc

Eu(III)					Conditio	IS				Conversi	on (C) / Uncer Factor(F)	tainty (U)
		Source data (a)	1	Additional in	formation (t	(		Application			CF	
Solution		SBPW		0.1 M I	VaCIO₄		RPW	RPWC	MASH	Mda	RPWC	MdSH
Substrate		MX-80	SWy-1 (c)	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	29	87	87	87	87	85.32	85.32	85.32	1	r.	-
Hď		7.6	7.6	7.377	6.59	7.046	7.377	6.59	7.046			
K <sub>d</sub> for 0.1 M NaClO <sub>4</sub>	(I/J)		354.81	223.87	35.48	158.49						
CF-pH										0.631	0.100	0.447
CF-total 1										0.631	0.100	0.447
Speciation												
CF-spec 2		0.120					0.052	0.041	0.125	0.437	0.341	1.044
CF-spec 3		0.539					0.57	0.555	0.49	1.054	1.031	0.904
CF-total 2										0.276	0.034	0.466
CF-total 3										0.665	0.103	0.404
											٩	
Experimental K <sub>d</sub>	(D/J)	12.6										
Predicted K <sub>d</sub> (CF-total 1)	(F/J)						7.94	1.26	5.62			
UF-starting K <sub>d</sub>										1.6	1.6	1.6
UF-pH-conversion										2.5	2.5	2.5
UF-batch -> compacted										2.0	2.0	2.0
UF-total										7.9	7.9	7.9
K <sub>d</sub> upper limit	(L/g)						62.9	10.0	44.6			
K <sub>d</sub> lower limit	(L/g)						1.00	0.16	0.71			
Predicted K <sub>d</sub> (CF-total 2)	(L/g)						3.473	0.430	5.872			
UF-starting K <sub>d</sub>										1.6	1.6	1.6
UF-pH-conversion										2.5	2.5	2.5
UF-speciation										1.4	1.4	1.4
UF-batch -> compacted										2.0	2.0	2.0
UF-total Ker limit							20 622	1 701	EE JED	11.1	11.1	11.1
	(6))						000.00	161.4	00.400			
	(LUG)						0.313	0.039	170.0			
Predicted K <sub>d</sub> (CF-total 3)	(L/g)						8.376	۲	5.084			
UF-starting K <sub>d</sub>										1.6	1.6	1.6
UF-pH-conversion										2.5	2.5	2.5
UF-speciation										1.4	1.4	1.4
UF-batch -> compacted										2.0	2.0	2.0
UF-total										11.1	11.1	11.1
K <sub>d</sub> upper limit	(L/g)						93.369	14	56.670			
K <sub>d</sub> lower limit	(L/g)						0.751	0	0.456			

(a) SBPW: Synthetic bentonite porewater used in the sorption of Eu on MX-80 (Bradbury&Baeyens, 2003)
(b) Eu sorption on Swy-1 in 0.1 NaClO4 solution for pH conversion
(c) Starting condition for pH conversion factor calculation
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system
HSPW: High saline porewater (Beberg water)

Predicted and calculated K<sub>d</sub> values for Eu(III). Data source: experimental K<sub>d</sub> values for Eu(III) sorption on SWy-1 montmorillonite.

	Eu(III)				Conditic	su			Conversio	יטווט / עווט Factor(F)	ertainty (U)
Observation         OIM MECIO         NE-YW         RFWV         RFWV <th></th> <th></th> <th>Dati</th> <th>a Source (a)</th> <th></th> <th></th> <th>Application</th> <th></th> <th></th> <th>СF</th> <th></th>			Dati	a Source (a)			Application			СF	
Swyci         Swyci <th< th=""><th>Solution</th><th></th><th>0.1</th><th>M NaCIO4</th><th></th><th>RPW</th><th>RPWC</th><th>MSPW</th><th>RPW</th><th>RPWC</th><th>MASH</th></th<>	Solution		0.1	M NaCIO4		RPW	RPWC	MSPW	RPW	RPWC	MASH
CEC         (mont/100)         87         87         87         85.22         85.32         85.32         65.93         70.46         1	Substrate		Swy-1	Swy-1	Swy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
pH $7.377$ 6.59 $7.046$ $7.37$ 6.59 $7.046$ $1$ $1$ $1$ Special         Special         100         1000         1000         0.622         0.041         0.125         0.631         0.141         0.141           Special         1000         1000         1000         0.656         0.641         0.125         0.641         0.125           CF-spec 3         CF-spec 3         0.041         0.052         0.041         0.125         0.647         0.647           CF-spec 3         CF-spec 3         0.041         1.000         1.000         0.568         0.641         0.611         0.125           CF-spec 3         CF-spec 3         0.641         1.000         1.000         0.568         0.647         0.47           CF-spec 3         CF-spec 3         0.641         1.4         1.4         1.4         1.4           Predicted Kd CF-stal 1         U         U         1.4         1.4         1.4         1.4         1.4           Version         U         U         1.4         1.4         1.4         1.4         1.4           Version         U         U	CEC	(meq/100g)	87	87	87	85.32	85.32	85.32	0.98	0.98	0.98
CF-teal 1         0.961 <th0.961< th="">         0.961         0.961</th0.961<>	Hd		7.377	6.59	7.046	7.377	6.59	7.046	-	-	-
Speciation         1000         1000         1000         1000         0.053         0.041         0.125         0.041         0.125           Gr-spec 3         Cr-spec 3         1.000         1.000         0.066         0.066         0.665         0.641         0.125           Gr-spec 3         Cr-spec 3         Cr-spec 3         0.01         0.001         0.063         0.665         0.415           Gr-spec 3         Cr-spec 3         0.01         0.01         0.051         0.041         0.123           Gr-spec 3         Cr-spec 3         0.01         0.05         0.051         0.041         0.123           Gr-spec 3         Cr-spec 3         3.54         155.43         156.4         0.69         0.655         0.47           F-stell         U         U         0         210.55         3.48         156.4         16         17           F-stell         U         U         0         210.55         3.48         156.4         16         16         16           U-Stell         U         U         0.91         17.43         19.47         16         16         16         16           U-Stell         U         U         U	CF-total 1								0.981	0.981	0.981
CF-spec 2         1000         1000         1000         1000         0.556         0.41         0.125         0.431         0.226           CF-spec 3         CF-spec 3         0.555         0.437         0.565         0.437         0.565         0.437           CF-spec 3         CF-spec 3         0.555         0.437         0.567         0.567         0.647         0.725           CF-spec 3         CF-spec 3         0.55         0.437         0.557         0.437         0.557         0.437           CF-spec 3         0.55         0.437         0.553         0.437         0.557         0.437           CF-spec 3         0.55         0.43         1.5         1.43         1.5         0.557         0.437           CF-spec 3         0.55         0.437         2.04         2.04         2.0         2.0           UF-section         UF-section         0.56         0.437         2.4         4.4         4.4           UF-section         UF-section         UF-section         1.4         1.4         1.4         1.4           UF-section         UF-section         UF-section         1.43         1.4.4         1.4         1.4         1.4         1.4	Speciation										
	CF-spec 2		1.000	1.000	1.000	0.052	0.041	0.125	0.052	0.041	0.125
CF-total 2         CF-total 2         0.061         0.040         0.123           CF-total 3         CF-total 3         0.061         0.040         0.123           Experimental Kd         (U)         23.87         35.48         158.49         7         9         7           Experimental Kd         (U)         23.87         35.48         158.49         158.43         16         16         16           U-batch         (U)         23.87         35.48         158.49         158.43         16         16         16         16           U-batch         (U)         U         23.87         34.80         158.43         16         16         16         16           U-batch         (U)         U         97.23         7.84         26.82         34.80         16         16         16           U-batch         (U)         U         97.23         17.82         16.82         16         16         16           U-batch         (U)         U         97.23         17.84         17.84         14         14         14           U-batch         (U)         U         17.55         14.3         16.85         16.85         16.82	CF-spec 3		1.000	1.000	1.000	0.568	0.555	0.487	0.568	0.555	0.487
CF-total 3         CG54         0.477           Freetimental Kd         (Lg)         223.87         35.48         156.43         1.6         1.6         1.6           Freetimental Kd         (Lg)         223.87         35.48         156.43         1.6         1.6         1.6           UF-estiming Kd         (Lg)         223.87         35.48         156.43         1.6         1.6         1.6           UF-estiming Kd         (Lg)         223.87         35.48         155.43         1.6         1.6         1.6           UF-estiming Kd         (Lg)         219.55         34.80         155.43         1.6         1.4         1.4           UF-estiming Kd         (Lg)         219.55         34.80         155.43         1.6         1.6         1.6           UF-estiming Kd         (Ug)         11.55         34.47         7.84         35.02         2.0	CF-total 2								0.051	0.040	0.123
Experimental Kd $(Ug)$ 23.387         35.48         158.49         158.49         158.49         158.43         158.49         158.43         16	CF-total 3								0.557	0.545	0.477
Experimental Kd $(Ug)$ $22.3$ KJ $35.43$ $155.43$ $16$										IJ	
Predicted Kd ( $CF$ -total 1)         ( $Ug$ )         ( $Ug$ )         ( $Ug$ )         ( $Ug$ )         ( $10$ )         ( $16$ ) <td>Experimental Kd</td> <td>(l/d)</td> <td>223.87</td> <td>35.48</td> <td>158.49</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>	Experimental Kd	(l/d)	223.87	35.48	158.49						
UF-starting kd         UF         Inf	Predicted Kd (CF-total 1)	(IT/a)				219.55	34.80	155.43			
UF-CEC         UF-CEC         14	UF-starting Kd	5							1.6	1.6	1.6
UF-batch - compacted         U	UF-CEC								1.4	1.4	1.4
UF-total         UF-total $(1-0)$	UF-batch -> compacted								2.0	2.0	2.0
Kd upper limit         (Lg) $974.29$ $154.42$ $689.75$ $r$ $r$ Kd upper limit         (Ug) $49.47$ $7.84$ $35.02$ $r$ $r$ VE clocket K(F-total 2)         (Ug) $r$	UF-total								4.4	4.4	4.4
Kd lower limit         ( $Ug$ )         ( $1.6$ <t< td=""><td>Kd upper limit</td><td>(F/d)</td><td></td><td></td><td></td><td>974.29</td><td>154.42</td><td>689.75</td><td></td><td></td><td></td></t<>	Kd upper limit	(F/d)				974.29	154.42	689.75			
Predicted Kd ( $\Gamma$ -total 2) $(Ug)$ $(Ug)$ $(Ug)$ $(1.52)$ $1.43$ $19.47$ $1.6$	Kd lower limit	(F/d)				49.47	7.84	35.02			
UF-starting kd         I	Predicted Kd (CF-total 2)	(F/d)				11.52	1.43	19.47			
UF-CEC         UF-CEC         1.4	UF-starting Kd								1.6	1.6	1.6
UF-speciation         I	UF-CEC								1.4	1.4	1.4
UF-batch $\sim$ compacted         U         U         L <thl< th="">         L         L         L</thl<>	UF-speciation								1.4	1.4	1.4
UF-total       UF-total $(Ug)$ $(Ug)$ $(Ug)$ $(C)$	UF-batch -> compacted								2.0	2.0	2.0
Kd upper limit         (Ug)         (Ug)         (157         8.856         120.987         (19           Kd lower limit         (Lg)         1.854         0.229         3.134         (19         (16           Predicted Kd (CF-total 3)         (Lg)         1.854         0.229         3.134         (16         1.6           UF-starting Kd         (L-starting Kd         (19,3)         75.68         1.6         1.6         1.6           UF-starting Kd         (17,10)         (18,10)         (18,10)         1.4         1.4         1.4           UF-speciation         (17,10)         (17,10)         (17,10)         (17,10)         (17,10)         (17,10)           UF-total         (17,10)         (17,10)         (17,10)         (17,10)         (17,10)         (17,10)           UF-total         (17,10)         (17,10)         (17,10)         (17,10)         (17,10)         (17,10)	UF-total								6.2	6.2	6.2
Kd lower limit         (L/g)         (L/g)         1.854         0.229         3.134         P         P           Predicted Kd (CF-total 3)         Pedicted Kd (CF-total 3)         1 <td>Kd upper limit</td> <td>(L/g)</td> <td></td> <td></td> <td></td> <td>71.557</td> <td>8.856</td> <td>120.987</td> <td></td> <td></td> <td></td>	Kd upper limit	(L/g)				71.557	8.856	120.987			
Predicted Kd (CF-total 3)         Predicted Kd (CF-total 3)         P13.32         75.68         1.6	Kd lower limit	(F/d)				1.854	0.229	3.134			
UF-starting Kd       1       1.6	Predicted Kd (CF-total 3)					124.68	19.32	75.68			
UF-CEC         II-CEC         II-4	UF-starting Kd								1.6	1.6	1.6
UF-speciation         UF-speciation         1.4	UF-CEC								1.4	1.4	1.4
UF-batch -> compacted         Image: Definition of the state of	UF-speciation								1.4	1.4	1.4
UF-total         UF-total         6.2         <	UF-batch -> compacted								2.0	2.0	2.0
Kd upper limit         774.618         120.037         470.158         P           Kd lower limit         20.069         3.110         12.181         P           Calculated Kd (b)         30.969         6.344         39.458         P	UF-total								6.2	6.2	6.2
Kd lower limit         20.069         3.110         12.181           Calculated Kd (b)         30.969         6.344         39.458	Kd upper limit					774.618	120.037	470.158			
Calculated Kd (b) 6.344 39.458 6.344 39.458	Kd lower limit					20.069	3.110	12.181			
	Calculated Kd (b)					30.969	6.344	39.458			

(a) Eu sorption on Na-SWy-1 in 0.1 NaCIO4 solution (Bradbury and Baeyens, 2003b) (b) Calculated using the thermodynamic sorption model by Bradbury and Baeyens (2002a) RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

Predicted K <sub>d</sub> values for	r Pb(II). I	Data sou	rce: Exp	erimenta	al K <sub>d</sub> val	lues for ]	Pb(II) sc	rption o	n montn	norillonite
Pb(II)				Cond	itions			Conversio	n (C) / Unce Factor(F)	rtainty (U)
			ata Source (	a)		Application			СF	
Solution			0.1 M NaCIO	1	RPW	RPWC	MSH	RPW	RPWC	MASH
Substrate		E	ontmorilloni	te	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	76	76	76	85.32	85.32	85.32	1.12	1.12	1.12
ц		-3.5	-3.5	7.040 -3.5	-2.6	0.98 0.0-	7.040 -2.6	-	-	-
CF-total 1								1.12	1.12	1.12
Speciation										
CF-spec. 2 CF-spec. 3		0.9999 1.0000	1.0000 1.0000	1.0000 1.0000	0.9908 0.9921	0.9907 0.9922	0.7305 0.7305	0.9909 0.9921	0.9907 0.9922	0.7305 0.7305
CF-total 2								1.112	1.112	0.820
CF-total 3								1.114	1.114	0.820
									ЧF	
Experimental K <sub>d</sub>	(L/g)	690.99	31.623	56.234						
Predicted Kd (CF-total 1)	(F/d)				74.17	35.50	63.13			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								4.4	4.4	4.4
K <sub>d</sub> upper limit	(L/g)				329.15	157.54	280.15			
K <sub>d</sub> lower limit	(F/d)				16.71	8.00	14.23			
Predicted K <sub>d</sub> (CF-total 2)	(F/d)				73.50	35.17	46.12			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								4.1	1.4	4.1
UF-batch -> compacted								2.0	2.0	2.0
UF-total					00 01 7		01 000	6.2	6.2	6.2
	(L/g)				450.03	LC.812	20.082			
K <sub>d</sub> lower limit	(F/d)				11.83	5.66	7.42			
Predicted K <sub>d</sub> (CF-total 3)	(F/d)				73.58	35.22	46.12			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								6.2	6.2	6.2
K <sub>d</sub> upper limit	(L/g)				457.16	218.84	286.53			
K <sub>d</sub> lower limit	(F/d)				11.84	5.67	7.42			
		1-1-1								

(a): Pb(II) sorption on montmorillonite (Ulrich & Degueldre, 1993) RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

													•		
Np(IV) analogue to Th(IV)							Condi	tions					Conversio	n (C) / Uncer Factor(F)	tainty (U)
		Data source (a)	Additio	nal informat	ion (b)	∢	dditional inf	ormation (c			Application			сF	
Solution		SBPW	RPW	RPWC	MSPW		0.1 M N	aCIO4		RPW	RPWC	HSPW	RPW	RPWC	MASH
Substrate		MX-80	MX-80	MX-80	MX-80	SWy-1	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	62	85.32	85.32	85.32	87	87	87	87	85.32	85.32	85.32	1	1	1
Ha		7.6	7.377	6.593	7.046	7.6	7.377	6.59	7.046	7.377	6.59	7.046			
Kd for 0.1 M NaClO4 (pH conversion) (a)						355	355	224	224	-2.6	-0.98	-2.6	1.0000	0.6310	0.6310
CF-total 1													1.0000	0.6310	0.6310
Speciation															
CF-spec 2 (b)			0.0018	0.0003	0.0040					0.99988	0.97249	1.00000	551.5263	3.66E+03	251.7173
CF-spec 3 (b)			1.0000	0.9996	1.0000					0.99989	1.00000	1.00000	0.9999	1.0004	1.0000
CF-total 2													551.53	2309.42	158.823
CF-total 3													0.9999	0.6312	0.6310
														Ŀ	
Experimental Kd (a)	(F/d)	63.00													
Predicted Kd (CF-total 1)	(T/a)									63.0	39.8	39.8			
UF-starting Kd													2.5	2.5	2.5
UF-pH-conversion													2.5	2.5	2.5
UF-batch -> compacted													2.0	2.0	2.0
UF-total													12.6	12.6	12.6
Kd upper limit	(F/d)									795.0	501.6	501.6			
Kd lower limit	(F/d)									5.0	3.2	3.2			
Predicted Kd (CF-total 2)	(F/d)									34746.2	145493.5	10005.8			
UF-starting Kd	) 												2.5	2.5	2.5
UF-pH-conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													17.7	17.7	17.7
Kd upper limit	(F/d)									613853.6	2570405.1	176771.2			
Kd lower limit	(F/d)									1966.7	8235.4	566.4			
Predicted Kd (CF-total 3)	(F/d)									63.0	39.8	39.8			
UF-starting Kd													2.5	2.5	2.5
UF-pH-conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													17.7	17.7	17.7
Kd upper limit	(L/g)									1112.9	702.5	702.3			
Kd lower limit	(F/d)									3.6	2.3	2.3			

Predicted K<sub>d</sub> values for Np(IV). Data source: Experimental K<sub>d</sub> values for Th(IV) sorption on MX-80.

Kd lower limit

(a): Data from Nagra NTB 02-18 for Th
 (b): Th speciation for MX-80 application
 (c): pH conversion from Th (Bradbury & Baeyens, unpublished data in Nagra NRB 02-18)
 SEPW: Synthetic benchoite porewater used in the sorption of Th on MX-80 (Bradbury&Baeyens, 2003)
 RPW: Reference porewater (saline Beberg water)
 RPWC: Reference porewater (Bailne Beberg water)
 RPWC: High saline porewater (Beberg water)

Np(V)				Condi	tions			Cor Uncerta	nversion ( iinty (U) F	C) / actor(F)
			ata source (a)			Application			Ŀ	
Solution			0.1 M NaNO3		Mda	RPWC	MdSH	Mda	RPWC	MSPW
Substrate		SAz-1	SAz-1	SAz-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	120	120	120	85.32	85.32	85.32	0.711	0.711	0.711
Hd		7.377	6.593	7.046	7.377	6.593	7.046	-	-	-
CF-total 1								0.711	0.711	0.711
Speciation										
CF-spec 2		0.94795	0.94794	0.94794	0.897244	0.885968	0.975747	0.947	0.935	1.029
CF-spec 3		0.94795	0.94794	0.94794	0.980163	0.977987	0.995317	1.034	1.032	1.050
CF-total 2								0.673	0.665	0.732
CF-total 3								0.735	0.734	0.747
									Ŀ	
Experimental Kd	(L/g)	0.034	0.020	0.031						
Predicted Kd (CF-total 1)	(L/g)				0.024	0.014	0.022			
UF-starting Kd	ò							1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								4.4	4.4	4.4
Kd upper limit	(L/g)				0.107	0.063	0.098			
Kd lower limit	(L/g)				0.005	0.003	0.005			
Predicted Kd (CF-total 2)	(L/g)				0.023	0.013	0.023			
UF-starting Kd								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total					0110	120000	111	6.2	6.2	6.2
Kd Iower limit					0.004	0.0023/4	0.004			
	(L/g)				0.004	0.002134	0.004			
Predicted Kd (CF-total 3)	(L/g)				0.025	0.015	0.023			
UF-starting Kd								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								6.2	6.2	6.2
Kd upper limit	(L/g)				0.155	0.090930	0.143			
Kd lower limit	(L/g)				0.004	0.002356	0.004			

Predicted K<sub>d</sub> values for Np(V). Data source: Experimental K<sub>d</sub> values for Np(V) sorption on SAz-1.

(a): Np(V) sorption on SAz-1 Na-montmorillonite in 0.1 M NaNO3 (Turner et al., 1998) RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

Predicted K <sub>d</sub> values for Ni. Da	ata sourc	e: experimen	tal K <sub>d</sub> v	alues for	r Ni(II) s	sorption	on MX-	80				
Ni(II)					Condi	tions				Conversi	on (C) / Uncei Factor(F)	tainty (U)
		Data source (a)	Ā	dditional inf	ormation (b			Application			СF	
Solution		SBPW		0.1 M N	aCI04		RPW	RPWC	MSPW	RPW	RPWC	HSPW
Substrate		MX-80	Swy-1	Swy-1	Swy-1	Swy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	29	87	87	87	87	85.32	85.32	85.32	1	1	۲
pH		7.6	7.6	7.377	6.59	7.046	7.377	6.59	7.046			
Kd for 0.1 M NaCIO4 (pH conversion) (a)	(L/g)		3.162	2.818	0.631	0.891				0.891	0.200	0.282
CF-total 1										0.891	0.200	0.282
Speciation												
CF-spec 2		0.4268					0.572	0.544	0.409	1.340	1.275	0.958
CF-spec 3		0.4273					0.574	0.546	0.409	1.342	1.278	0.958
CF-total 2										1.194	0.254	0.270
CF-total 3										1.196	0.255	0.270
											Ŀ	
Experimental Kd	(L/g)	0.2512										
Predicted Kd (CF-total 1)	(F/d)						0.224	0.050	0.071			
UF-starting Kd										1.6	1.6	1.6
UF-pH conversion										2.5	2.5	2.5
UF-batch -> compacted										2.0	2.0	2.0
UF-total										7.9	7.9	7.9
Kd upper limit	(L/g)						1.8	0.4	9.0			
Kd lower limit	(L/g)						0.03	0.01	0.01			
Predicted Kd (CF-total 2)	(F/d)						0.300	0.064	0.068			
UF-starting Kd										1.6	1.6	1.6
UF-pH conversion										2.5	2.5	2.5
UF-speciation										1.4	1.4	1.4
UF-batch -> compacted										2.0	2.0	2.0
UF-total										11.1	11.1	11.1
Kd upper limit	(F/d)						3.328	0.709	0.753			
Kd lower limit	(L/g)						0.027	0.006	0.006			
Predicted Kd (CF-total 3)	(L/g)						0.300	0.064	0.068			
UF-starting Kd										1.6	1.6	1.6
UF-pH conversion										2.5	2.5	2.5
UF-speciation										1.4	1.4	1.4
UF-batch -> compacted										2.0	2.0	2.0
UF-total										11.1	11.1	11.1
Kd upper limit	(F/d)						3.334	0.711	0.752			
Kd lower limit	(F/d)						0.027	0.006	0.006			

NAV 90 . NICTIN. ġ Ξ 401 IV . + ç Ξ Jintod IV

(a): Nagra NTB 02-18
 (b): Ni scrption data for Na-SWy1 (Baeyens and Bradbury, 1997)
 Spruct Synthetic bentonite porewater used in the scrption of Ni+2 on MX-80 (Table 4 in Nagra NTB 02-18)
 RPW: Reference porewater (saline Beberg water)
 RPWC: Reference porewater (saline Beberg water) for a closed system
 High saline porewater (Beberg water)

Ţ
V V
S
Ц
10
101
pt
201
Õ
Ē
Ż
or
sf
ue
/al
∕ p
$\mathbf{X}$
tal
en
В
en.
xp
ð.
ce
ur
SC
uta
ñ
: <u>-</u> :
Z
foi
S
lué
va
, T
Ţ
te
лlа
lc
ca
Jd
aī
ed
ict
ed
$\mathbf{P}_{\mathbf{r}}$

		•								
Ni+2				Condition	SL			Conversi	on (C) / Uncert Factor(F)	ainty (U)
		S	arting (a)			Application	-		СF	
Solution		0.1	M NaCIO4		Mda	RPWC	MdSH	RPW	RPWC	MSPW
Substrate		Swy-1	Swy-1	Swy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	87	87	87	85.32	85.32	85.32	0.98	0.98	0.98
Н		7.377	6.59	7.046	7.377	6.59	7.046	1.000	1.000	1.000
CF-total 1								0.981	0.981	0.981
Speciation										
CF-spec. 2		1.000	1.000	1.000	0.5719	0.5442	0.4091	0.572	0.544	0.409
CF-total 2								0.561	0.534	0.401
									UF	
Experimental Kd	(L/g)	2.8184	0.6310	0.8913						
Predicted Kd (CF-total 1)	(F/d)				2.764	0.619	0.874			
JF-starting Kd								1.6	1.6	1.6
JF-CEC								1.4	1.4	1.4
JF-batch -> compacted								2.0	2.0	2.0
JF-total								4.5	4.5	4.5
Kd upper limit	(L/g)							12.4	2.8	3.9
4d lower limit	(L/g)				0.62	0.14	0.20			
Predicted Kd (CF-total 2)	(L/g)				1.581	0.337	0.358			
JF-starting Kd								1.6	1.6	1.6
JF-CEC								1.4	1.4	1.4
JF-speciation								1.4	1.4	1.4
JF-batch -> compacted								2.0	2.0	2.0
JF-total								6.3	6.3	6.3
Kd upper limit	(L/g)				9.913	2.112	2.243			
Ad lower limit	(F/d)				0.252	0.054	0.057			
						000	007			
Calculated Kg (b)	([[]]				4.964	070 L	1.436			

(a): Ni sorption data for Na-SWy1 (Baeyens and Bradbury, 1997)
(b): Calculated using the thermodynamic sorption model of Bradbury and Baeyens (1997) RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water)
HSPW: High saline porewater (Beberg water)

Pu(III) analogue to Am(III)						Conditions					Conversio	1 (C) / Uncer Factor(F)	tainities (U)
		ă	ata Source (	(a)	Additic	onal information	tion (b)		Application			Ъ	
Solution		0	1 M NaCIC	4	RPW	RPWC	MASH	RPW	RPWC	MASH	RPW	RPWC	MdSH
Substrate		Smectite	Smectite	Smectite	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	79	62	62	85.32	85.32	85.32	85.32	85.32	85.32	1.08	1.08	1.08
РН		7.377	6.593	7.046	7.377	6.59	7.046	7.377	6.59	7.046	-	-	-
CF-total 1											1.080	1.080	1.080
Speciation													
CF-spec 2 (95Sil/Bid-TDB for Am) (b)	(mol/L)				0.505	0.086	0.334	0.1568	0.0967	0.2659	0.311	1.121	0.797
CF-spec 3 (03Gul/Fan-IDB for Am) (b)	(mol/L)				0.107	0.068	0.267	0.1568	0.0967	0.2659	1.466	1.413	0.994
CF-total 2 CF-total 3											0.335 1.583	1.210 1.526	0.861 1.074
												Ŀ	
Experimental K <sub>d</sub> (b)	(F/d)	63.10	19.95	39.81									
Predicted K <sub>d</sub> (CF-total 1)	(L/a)							68.14	21.55	43.00			
UF-starting K <sub>d</sub>											2.5	2.5	2.5
UF-CEC											1.4	4.1	1.4
UF-batch -> compacted											2.0	2.0	2.0
UF-total											7.0	7.0	7.0
K <sub>d</sub> upper limit	(F/d)							479.3	151.6	302.4			
K <sub>d</sub> lower limit	(F/d)							9.69	3.06	6.11			
Predicted K <sub>d</sub> (CF-total 2)	(L/a)							21.162	24.146	34.263			
UF-starting K <sub>d</sub>	) -										2.5	2.5	2.5
UF-CEC											1.4	1.4	1.4
UF-speciation											1.4	1.4	1.4
UF-batch -> compacted			_								2.0	2.0	2.0
UF-total											9.8	9.8	9.8
K <sub>d</sub> upper limit	(F/d)					_		208.369	237.757	337.370			
K <sub>d</sub> lower limit	(F/d)							2.149	2.452	3.480			
Predicted K <sub>d</sub> (CF-total 3)	(L/a)							99.901	30.448	42.749			
UF-starting K <sub>d</sub>											2.5	2.5	2.5
UF-CEC											1.4	1.4	1.4
UF-speciation	_		_								1.4	1.4	1.4
UF-batch -> compacted											2.0	2.0	2.0
UF-total											9.8	9.8	9.8
K <sub>d</sub> upper limit	(F/d)							983.681	299.810	420.929			
K <sub>d</sub> lower limit	(F/d)							10.146	3.092	4.341			

Predicted K<sub>d</sub> values for Pu(III). Data source: Experimental K<sub>d</sub> values for Am(III) sorption on montmorillonite.

(a): Am(III) used as analogue for Pu(III) (Gergeon, 1994)
 (b): Am speciation for MX-80 application calculated with different TDB RPW: Reference porewater (saline Beberg water)
 RPWC: Reference porewater (saline Beberg water)
 High saline porewater (Beberg water)

Ö
$\tilde{\mathbf{x}}$
Ť
$\mathbf{\mathbf{N}}$
$\sim$
$\geq$
U
5
$\mathbf{U}$
n
Ö
·Ξ
Ħ
- 24
Ξ
2
01
>
$\sim$
Ч
Ē
۲. '
F
,0
f
$\mathbf{v}$
Ö
Ĕ
E.
а
>
·
1
$\mathbf{X}$
F
43
Ц
Ö
ā
P
Ľ.
Ы
Der
tper
xper
Exper
: Exper
e: Exper
ce: Exper
rce: Exper
urce: Exper
ource: Exper
source: Exper
1 source: Exper
ta source: Exper
ata source: Exper
Data source: Exper-
Data source: Exper-
. Data source: Exper
). Data source: Exper
V). Data source: Exper
IV). Data source: Exper-
(IV). Data source: Exper
u(IV). Data source: Exper
Pu(IV). Data source: Exper
· Pu(IV). Data source: Exper
or Pu(IV). Data source: Exper
for Pu(IV). Data source: Exper
for Pu(IV). Data source: Exper
s for Pu(IV). Data source: Exper
es for Pu(IV). Data source: Exper
ues for Pu(IV). Data source: Exper
lues for Pu(IV). Data source: Exper
alues for Pu(IV). Data source: Exper-
values for Pu(IV). Data source: Exper
<sup>1</sup> values for Pu(IV). Data source: Exper
<sup>d</sup> values for Pu(IV). Data source: Exper
K <sub>d</sub> values for Pu(IV). Data source: Exper
I K <sub>d</sub> values for Pu(IV). Data source: Exper
ed K <sub>d</sub> values for Pu(IV). Data source: Exper
ted K <sub>d</sub> values for Pu(IV). Data source: Exper
cted K <sub>d</sub> values for Pu(IV). Data source: Exper
icted K <sub>d</sub> values for Pu(IV). Data source: Exper
dicted K <sub>d</sub> values for Pu(IV). Data source: Exper
edicted K <sub>d</sub> values for Pu(IV). Data source: Exper
redicted K <sub>d</sub> values for Pu(IV). Data source: Exper

Pu(IV), analogy with Th (IV)							Cond	itions					Conversio	n (C) / Uncel Factor(F)	rtainty (U)
		Data source (a)	Additio	nal informat	tion (b)	٩	dditional in	formation (c			Application			СF	
Solution		SBPW	RPW	RPWC	MdSH		0.1 M N	laCI04		RPW	RPWC	MdSH	RPW	RPWC	MSH
Substrate		MX-80	MX-80	MX-80	MX-80	SWy-1	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	62 (	85.32	85.32	85.32	28	87	87	87	85.32	85.32	85.32	1	1	1
pH		7.6	7.377	6.593	7.046	7.6	7.377	6.59	7.046	7.377	6.59	7.046			
Kd for 0.1 M NaClO4 (pH conversion) (b)						355	355	224	224				1.0000	0.6310	0.6310
CF-total 1													1.000	0.631	0.631
Speciation															
CF-spec 2			0.0018	0.0003	0.0040					0.0015020	0.0000002	0.0035631	8.28E-01	6.93E-04	8.97E-01
CF-spec 3			1.0000	0.9996	1.0000					0.9985	1.0000	0.9965	0.999	1.000	0.997
CF-total 2													0.82849	0.00044	0.56591
CF-total 3													0.99851	0.63119	0.62876
														Ŀ	
Experimental Kd for Th (a)	(F/d)	63.00													
Predicted Kd (CF-total 1)	(IT/a)									63.00	39.75	39.75			
UF-starting Kd													2.5	2.5	2.5
UF-pH conversion													2.5	2.5	2.5
UF-batch -> compacted													2.0	2.0	2.0
UF-total													12.6	12.6	12.6
Kd upper limit	(F/d)									795.0	501.6	501.6			
Kd lower limit	(F/d)									4.99	3.15	3.15			
Predicted Kd (CF-total 2)	(F/a)									52.19	0.03	35.65			
UF-starting Kd	5												2.5	2.5	2.5
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													17.67	17.67	17.67
Kd upper limit	(F/d)									922.1	0.5	629.9			
Kd lower limit	(L/g)									2.95	0.00	2.02			
Predicted Kd (CF-total 3)	(F/d)									62.91	39.76	39.61			
UF-starting Kd													2.5	2.5	2.5
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													17.67	17.67	17.67
Kd upper limit	(L/g)									1111.3	702.5	699.8			
Kd lower limit	(F/d)									3.56	2.25	2.24			

(a): Data from Nagra NTB 02-18 (Bradbury and Baeyens, 2003a) for Th
(b): Th speciation for MX-80 application
(c): pH conversion from Th (Bradbury & Baeyens 2003a)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water)
HSPW: High saline porewater (Beberg water)

Pu(V) analogue to Np(V)						Conditions					Cor Uncerta	iversion ( inty (U) Fi	C) / actor(F)
			Data Source (a)		Additi	onal informat	ion (b)		Application			Ъ	
Solution			0.1 M NaNO3		RPW	RPWC	MSPW	RPW	RPWC	MASH	RPW	RPWC	MSPW
Substrate		SAz-1	SAz-1	SAz-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	120	120	120	85.32	85.32	85.32	85.32	85.32	85.32	0.711	0.711	0.711
ΡΗ		7.377	6.593	7.046	7.377	6.593	7.046	7.377	6.593	7.054	-	-	-
CF-total 1											0.711	0.711	0.711
Speciation													
CF-spec 2 (analogue)					0.8972	0.8860	0.9757	0.8825	0.8696	0.9719	0.984	0.982	0.996
CF-spec 3 (analogue)					0.9802	0.9780	0.9953	1.0000	1.0000	1.0000	1.020	1.023	1.005
CF-total 2 CF-total 3											0.699	0.698 0.727	0.708 0.714
												Ŀ	
Experimental K <sub>d</sub>	(b/J)	0.034	0.020	0.031									
Predicted K <sub>d</sub> (CF-total 1)	(b/J)							0.024	0.014	0.022			
UF-starting K <sub>d</sub>											2.5	2.5	2.5
UF-CEC											4.1	1.4	1.4
UF-batch -> compacted											2.0	2.0	2.0
UF-total											7.0	7.0	7.0
K <sub>d</sub> upper limit	(F/d)							0.169	0.100	0.155			
K <sub>d</sub> lower limit	(F/J)							0.003	0.002	0.003			
Predicted K <sub>d</sub> (CF-total 2)	(6/T)							0.024	0.014	0.022			
UF-starting K <sub>d</sub>											2.5	2.5	2.5
UF-CEC											1.4	1.4	1.4
UF-speciation											1.4	1.4	1.4
UF-batch -> compacted											2.0	2.0	2.0
UF-total											9.8	9.8	9.8
K <sub>d</sub> upper limit	(L/g)							0.233	0.137103	0.215			
K <sub>d</sub> lower limit	(L/g)							0.002	0.001414	0.002			
Predicted K, (CF-total 3)	(Г/а)							0.025	0.015	0.022			
UF-starting K <sub>d</sub>	) -										2.5	2.5	2.5
UF-CEC											1.4	1.4	1.4
UF-speciation											1.4	1.4	1.4
UF-batch -> compacted											2.0	2.0	2.0
UF-total											9.8	9.8	9.8
K <sub>d</sub> upper limit	(F/d)							0.242	0.142831	0.217			
K <sub>d</sub> lower limit	(L/g)							0.002	0.001473	0.002			

Predicted K<sub>d</sub> values for Pu(V). Data source: Experimental K<sub>d</sub> values for Np(V) sorption on Saz-1.

(a): Np(V) sorption on SAz-1 Na-montmorillonite in 0.1 M NaNO<sub>3</sub> (Turner et al., 1998)
(b): Np(V) speciation for MX-80 application RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

Predicted K<sub>d</sub> values for Pu(VI). Data source: Experimental K<sub>d</sub> values for U(VI) sorption on Saz-1.

			1										
u(VI), analogy with U (VI)						Conditio	IS				Cor Uncerta	iversion ( inty (U) Fa	C) / Ictor(F)
		Ö	ata source (	a)	Additio	nal informati	on (b)		Application			Ъ	
Solution		0	.1 M NaNO3		RPW	RPWC	MASH	RPW	RPWC	MSPW	RPW	RPWC	MSPW
Substrate		Saz-1	Saz-1	Saz-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	120	120	120	85.32	85.32	85.32	85.32	85.32	85.32	0.711	0.711	0.711
H		7.377	6.593	7.046	7.377	6.593	7.046	7.377	6.593	7.046	-	-	-
CF-total 1											0.711	0.711	0.711
Speciation													
CF-spec 2 (analogue)					0.03560	0.00455	0.28489	0.59629	0.55225	0.88289	16.750	121.361	3.099
CF-spec 3 (analogue)					1.00000	1.00000	1.00000	0.99297	0.99294	0.98615	0.993	0.993	0.986
CF-total 2											11.9095	86.2877	2.2034
CF-total 3											0.7060	0.7060	0.7012
												Ъ	
Experimental Kd	(F/d)	4	20	4									
Predicted Kd (CF-total 1)	(L/g)							2.844	14.22	2.844			
JF-starting Kd	ĵ										2.5	2.5	2.5
JF-CEC											1.4	1.4	1.4
JF-batch -> compacted											2	2	2
JF-total											7.0	7.0	7.0
State of the st	(L/g)							20.003	100.013	20.003			
sd lower limit	(F/d)							0.404	2.022	0.404			
Predicted Kd (CF-total 2)	(b/J)							47.638	1725.754	8.814			
JF-starting Kd											2.5	2.5	2.5
JF-CEC											1.4	1.4	1.4
JF-speciation											1.4	1.4	1.4
JF-batch -> compacted											2.0	2.0	2.0
JF-total											9.8	9.8	9.8
<pre>Kd upper limit</pre>	(L/g)							469.074	16992.796	86.784			
<d limit<="" lower="" td=""><td>(L/g)</td><td></td><td></td><td></td><td></td><td></td><td></td><td>4.838</td><td>175.264</td><td>0.895</td><td></td><td></td><td></td></d>	(L/g)							4.838	175.264	0.895			
Predicted Kd (CF-total 3)	(b/J)							2.824	14.120	2.805			
JF-starting Kd											2.5	2.5	2.5
JF-CEC											1.4	1.4	1.4
JF-speciation											1.4	1.4	1.4
JF-batch -> compacted											2.0	2.0	2.0
JF-total											9.8	9.8	9.8
<pre>Kd upper limit</pre>	(L/g)							27.807	139.029	27.616			
Kd lower limit	(L/g)							0.287	1.434	0.285			
		- -											

(a): U(VI) Saz-1 montmorillonite in 0.1 NaNO3 (Pabalan & Turner, 1997)
(b): U(VI) speciation for MX-80 application RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

Calculated  $K_d$  values for Sr(II) and Ra(II): Thermodynamic sorption model based on the ion exchange constant given for  $Ca^{2+}$  in Fletcher and Sposito (1989).

Sr(II) or Ra(II)					Co Uncertai	nversion inities (U)	(C) / Factor(F)
Solution			Application			СF	
Substrate		RPW	RPWC	HSPW	RPW	RPWC	MSPM
Material		MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	85.32	85.32	85.32			
Hd		7.377	6.593	7.046			
						υF	
Calculated K <sub>d</sub>	(L/g)	0.0051	0.0053	0.0013			
UF-model					3.0	3.0	3.0
UF-batch -> compacted					2.0	2.0	2.0
UF-total					6.0	6.0	6.0
$K_d$ upper limit	(F/d)	0.0308	0.0319	0.0080			
$K_d$ lower limit	(F/d)	0.0009	0.0009	0.0002			

RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

		-	1	~	-		I					
Se(IV)					Conditions					Conversic (U	n (C) / Und ) Factor(F)	certainty
		Data Source (a)	A	Aditional info	rmation (b)		4	<b>Application</b>			СF	
Solution		SBPW		0.1 M Nã	ICIO4		RPW	RPWC	MdSH	RPW	RPWC	MdSH
Substrate		MX-80	SWy-1 (c)	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	79	87	87	87	87	85.32	85.32	85.32	t	Ļ	1
pH		7.8	7.8	7.377	6.59	7.046	7.377	6.59	7.046			
K <sub>d</sub> for 0.1 M NaClO <sub>4</sub>	(L/g)		0.079	0.1259	0.2512	0.1585				1.58	3.16	2.00
CF-pH conversion												
CF-total 1										1.585	3.162	1.995
Speciation												
CF-spec 2		1.000					0.809	0.963	0.909	0.809	0.963	0.909
CF-total 2										1.283	3.044	1.813
											ЦГ	
Experimental K <sub>d</sub>	(L/g)	0.03										
Predicted Kd (CF-total 1)	(F/d)						0.04	0.09	0.06			
UF-starting Kd	Ď									1.6	1.6	1.6
UF-pH-conversion										2.5	2.5	2.5
UF-batch -> compacted										2.0	2.0	2.0
UF-total										8.0	8.0	8.0
K <sub>d</sub> upper limit	(L/g)						0.4	0.7	0.4			
K <sub>d</sub> lower limit	(L/g)						0.01	0.01	0.01			
Predicted K. (CF-total 2)	(1/a)						0.036	0.086	0.051			
UF-starting Kd										1.6	1.6	1.6
UF-pH-conversion										2.5	2.5	2.5
UF-speciation										1.4	4.1	1.4
UF-batch -> compacted										2.0	2.0	2.0
UF-total										11.1	11.1	11.1
K <sub>d</sub> upper limit	(F/d)						0.403	0.956	0.570			
K <sub>d</sub> lower limit	(L/g)						0.003	0.008	0.005			

Predicted K<sub>d</sub> values for Se (IV) Data source: Experimental K<sub>d</sub> values for Se(IV) sorption on MX-80. K<sub>d</sub> values for Se(VI) are zero.

(a) SBPW: Synthetic bentonite porewater used in the sorption of Se(IV) on MX-80 (Nagra NTB 02-18)
(b) SeIV sorption on Swy-1 in 0.1 NaCIO4 solution for pH conversion
(c) Starting condition for pH conversion factor calculation
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system
HSPW: High saline porewater (Beberg water)

		•				~	•								
Tc(IV), analogy with Th(IV)						ö	nditions						Conversio	on (C) / Uncei Factor(F)	tainty (U)
		Data source (a)	Addition	nal informat	ion (b)		dditional inf	ormation (c)			Application			СF	
Solution		SBPW	RPW	RPWC	MSPW		0.1 M N	aCIO4		RPW	RPWC	MSPW	RPW	RPWC	HSPW
Substrate		MX-80	MX-80	MX-80	MX-80	SWy-1	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	62	85.32	85.32	85.32	87	87	87	87	85.32	85.32	85.32	٢	1	٢
Hd		7.6	7.377	6.59	7.046	7.6	7.377	6.59	7.046	7.377	6.59	7.046	-	-	-
Kd for 0.1 M NaCIO4 (pH conversion) (c)		63				355	355	224	224				-	0.6310	0.6310
CF-total 1													-	0.6310	0.6310
Speciation															
CF-spec 2 (b)			0.0018	0.0003	0.0040					1.0000	0.9998	1.0000	551.59	3763.03	251.72
CF-spec 3 (b)			1.0000	0.9996	1.0000					1.0000	1.0000	1.0000	1.00	1.00	1.00
CF-total 2													551.59	2374.31	158.82
CF-total 3													1.000	0.631	0.631
Experimental Kd (a)	(L/d)	63.00												Ŀ	
Predicted Kd (CF-total 1)	(1/0)									63.000	39.750	39.750			
UF-starting Kd													4.0	4.0	4.0
UF-pH conversion													2.5	2.5	2.5
UF-batch -> compacted													2.0	2.0	2.0
UF-total													20.0	20.0	20.0
Kd upper limit	(F/d)									1260.000	795.006	795.006			
Kd lower limit	(L/g)									3.150	1.988	1.988			
Predicted Kd (CF-total 2)	(F/d)									34750	149582	10006			
UF-starting Kd	ĵ												4.0	4.0	4.0
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													28.0	28.0	28.0
Kd upper limit	(L/g)									973011	4188285	280163			
Kd lower limit	(L/g)									1241	5342	357			
Predicted Kd (CF-total 3)	(F/d)									63.00	39.76	39.75			
UF-starting Kd													4.0	4.0	4.0
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													28.0	28.0	28.0
Kd upper limit	(L/g)									1764.01	1113.42	1112.99			
Kd lower limit	(L/g)									2.25	1.42	1.42			

Predicted K<sub>d</sub> values for Tc(IV). Data source: Experimental K<sub>d</sub> values for Th(IV) sorption on MX-80.

(a) Data from Bradbury and Baeyens (2003a) for Th
(b) Th speciation for MX-80 application
(c) pH conversion based on data for Th reported by Bradbury and Baeyens (2003a)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system
HSPW: High saline porewater (Beberg water)

		Ţ	,		~ ~	-						
Th(IV)					Condition	IS				Conversi	on (C) / Uncert Factor(F)	ainty (U)
		Data source (a)		Additional inf	ormation (b)			Application			СF	
Solution		SBPW		0.1 M N	aCIO4		RPW	RPWC	MSPW	RPW	RPWC	MASH
Substrate		MX-80	SWy-1	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	62 (	87	87	87	87	85.32	85.32	85.32	1	-	1
Hd		7.6	7.6	7.377	6.59	7.046	7.377	6.59	7.046			
Kd for 0.1 M NaClO4 (pH conversion) (b)		63	355	355	224	224				-	0.6310	0.6310
CF-total 1										-	0.6310	0.6310
Speciation												
CF-spec 2		0.0087					0.0018	0.0003	0.0040	0.209	0.031	0.457
CF-spec 3		1.0000					1.0000	0.9996	1.0000	1.000	1.000	1.000
CF-total 2										0.209	0.019	0.289
CF-total 3										1.000	0.631	0.631
											UF	
Experimental Kd	(b/J)	63										
Predicted Kd (CF-total 1)	(IT/a)						63.096	39.811	39.811			
UF-starting Kd	ĥ									1.6	1.6	1.6
UF-pH conversion										2.5	2.5	2.5
UF-batch -> compacted										2.0	2.0	2.0
UF-total										7.9	7.9	7.9
Kd upper limit	(F/J)						500.000	315.479	315.479			
Kd lower limit	(F/J)						7.962	5.024	5.024			
Predicted Kd (CF-total 2)	(F/J)						13.168	1.218	18.206			
UF-starting Kd										1.6	1.6	1.6
UF-pH conversion										2.5	2.5	2.5
UF-speciation										1.4	1.4	1.4
UF-batch -> compacted										2.0	2.0	2.0
UF-total										11.1	11.1	11.1
Kd upper limit	(L/g)	(					146.084	13.509	201.981			
Kd lower limit	(b/l)						1.187	0.110	1.641			
Predicted Kd (CF-total 3)	(b/J)						63.096	39.796	39.811			
UF-starting Kd										1.6	1.6	1.6
UF-pH conversion										2.5	2.5	2.5
UF-speciation										1.4	1.4	1.4
UF-batch -> compacted										2.0	2.0	2.0
UF-total										11.1	11.1	11.1
Kd upper limit	(F/d)						699.999	441.507	441.670			
Kd lower limit	(L/g)	(					5.687	3.587	3.588			

Predicted K<sub>d</sub> values for Th(IV). Data source: Experimental K<sub>d</sub> values for Th(IV) sorption on MX-80

(a) Kd value from Th(IV) sorption isotherm (Bradbury and Baeyens, 2003a)
(b) Kd values from Th(IV) sorption edge (Bradbury and Baeyens, 2003a)
SBPW: Synthetic bentonite porewater used in the sorption of Th on MX-80 (Bradbury and Baeyens, 2003a)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water)
High saline porewater (Beberg water)

Th(IV)				Condition	s			Conversi	on (C) / Uncel Factor(F)	tainty (U)
		ŏ	ata source (a)			Application			CF	
Solution			.1 M NaCIO₄		RPW	RPWC	MdSH	RPW	RPWC	MdSH
Substrate		SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	87	87	87	85.32	85.32	85.32	0.9807	0.9807	0.9807
Hd		7.377	6.59	7.046	7.377	6.59	7.046	-	-	-
CF-total 1								0.9807	0.9807	0.9807
Speciation										
CF-spec 2		1.0000	1.0000	1.0000	0.0018	0.0003	0.0040	0.0018	0.0003	0.0040
CF-spec 3		1.0000	1.0000	1.0000	1.00000	0.99963	1.00000	1.0000	0.9996	1.0000
CF-total 2 CF-total 3								0.0018 0.9807	0.0003 0.9803	0.0039 0.9807
									UF	
Experimental K <sub>d</sub>	(b/J)	355	224	224						
Predicted K <sub>d</sub> (CF-total 1)	(6/J)				347.962	219.549	219.549			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								4.4	4.4	4.4
K <sub>d</sub> upper limit	(6/T)				1544.150	974.293	974.293			
K <sub>d</sub> lower limit	(I/J)				78.410	49.474	49.474			
Predicted K . (CF-total 2)	(1 /0)				0.631	0.058	0.872			
UF-starting K <sub>d</sub>	6							1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								6.2	6.2	6.2
K <sub>d</sub> upper limit	(b/l)				3.919	0.362	5.419			
K <sub>d</sub> lower limit	(IL/g)				0.102	0.009	0.140			
Predicted K <sub>d</sub> (CF-total 3) (b)	(6/J)				347.961	219.468	219.549			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								6.2	6.2	6.2
K <sub>d</sub> upper limit	(b/l)				2161.808	1363.508	1364.010			
K <sub>d</sub> lower limit	(6/1)				56.007	35.325	35.338			

Predicted K<sub>d</sub> values for Th(IV). Data source: Experimental K<sub>d</sub> values for Th(IV) sorption on SWy-1.

(a): K<sub>4</sub> values from NAGRA NRB 02-18
(b): Predicted K<sub>4</sub> values are similar to Kd values predicted with CF-total 1
SBPW: Synthetic bentonite porewater used in the sorption of Th on MX-80 (Bradbury&Baeyens, 2003)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water) for a closed system
HSPW: High saline porewater (Beberg water)

Predicted K<sub>d</sub> values for U(IV). Data source: Experimental K<sub>d</sub> values for Th(IV) sorption on MX-80

U(IV), analogy with Th(IV)						Cor	ditions						CONVERSIO	n (ບ) / Uncer Factor(F)	tainty (u)
		Data source (a)	Additio	nal informat	ion (b)	◄	dditional inf	ormation (c)	_		Application			СF	
Solution		SBPW	RPW	RPWC	MSPW		0.1 M N	aCIO4		RPW	RPWC	MSPW	RPW	RPWC	MSPM
Substrate		MX-80	MX-80	MX-80	MX-80	SWy-1	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	56	85.32	85.32	85.32	87	87	87	87	85.32	85.32	85.32	Ł	÷	<del>.</del>
PH		7.6	7.377	6.593	7.046	7.6	7.377	6.59	7.046	7.377	6.593	7.046	-	-	~
Kd for 0.1 M NaClO4 (pH conversion) (c)						355	355	224	224				-	0.6310	0.6310
CF-total 1													-	0.6310	0.6310
Speciation															
CF-spec. 2 analogue (b)			0.001813	0.000266	0.003973					1.000000	1.000000	1.000000	551.5941	3763.7124	251.7174
CF-spec. 3 analogue (b)			0.999999	0.999631	0.999999					1.000000	1.000000	1.000000	1.0000	1.0004	1.0000
CF-total 2													551.5941	2374.7420	158.8229
CF-total 3													1.0000	0.6312	0.6310
														5	
Experimental Kd (a)	(F/d)	63.00													
Predicted Kd (CF-total 1)	(F/d)									63.00	39.75	39.75			
UF-starting Kd													2.5	2.5	2.5
UF-pH conversion													2.5	2.5	2.5
UF-batch -> compacted													2.0	2.0	2.0
UF-total													12.6	12.6	12.6
Kd upper limit	(F/d)									795.006	501.615	501.615			
Kd lower limit	(F/d)									4.992	3.150	3.150			
Predicted Kd (CF-total 2)	(F/d)									34750	149609	10006			
UF-starting Kd													2.5	2.5	2.5
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total	:												17.7	17.7	17.7
Kd upper limit	(F/d)									613929	2643109	176771			
Kd lower limit	(F/d)									1967	8468	566			
Predicted Kd (CF-total 2)	(F/d)				Π					63.00	39.76	39.75			
UF-starting Kd													2.5	2.5	2.5
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	4.1
UF-batch -> compacted													2.0	2.0	2.0
UF-total													17.7	17.7	17.7
Kd upper limit	(F/d)									1113	703	702			
Kd lower limit	(L/g)									4	2	2		_	

(a) Kd values from for Th (Bradbury and Baeyens, 2003a)
(b) Th speciation for MX-80 application
(c) pH conversion based on data for Th reported by Bradbury and Baeyens (2003a)
SBPW: Synthetic bentonite porewater used in the sorption of Th on MX-80 (Bradbury and Baeyens, 2003)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water)
HSPW: High saline porewater (Beberg water)

								Č	/ uniareion /	1
U(VI)				ö	nditions			Uncerta	inty (U) F	actor(F)
		Da	ita source (	a)		Application			ĥ	
Solution		0	.1 M NaNO	_	RPW	RPWC	MSPW	RPW	RPWC	MSPW
Substrate		Saz-1	Saz-1	Saz-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	120	120	120	85.32	85.32	85.32	0.711	0.711	0.711
Hd		7.377	6.593	7.046	7.377	6.593	7.046	-	-	-
CF-total 1								0.711	0.711	0.711
Speciation										
CF-spec 2		0.8595	0.9954	0.9669	0.0356	0.0046	0.2849	0.041	0.005	0.295
CF-spec 3		1.0000	0.9998	1.0000	1.0000	1.0000	1.0000	1.000	1.000	1.000
CF-total 2								0.029	0.003	0.210
CF-total 3								0.711	0.711	0.711
									Ъ	
Experimental K <sub>d</sub>	(F/d)	4	20	4						
Predicted K <sub>d</sub> (CF-total 1)	(F/d)				2.844	14.22	2.844			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								4.4	4.4	4.4
K <sub>d</sub> upper limit	(F/d)				12.621	63.104	12.621			
K <sub>d</sub> lower limit	(F/d)				0.641	3.204	0.641			
Predicted K <sub>d</sub> (CF-total 2)	(F/d)				0.118	0.065	0.838			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total								6.2	6.2	6.2
K <sub>d</sub> upper limit	(F/d)				0.732	0.404	5.206			
K <sub>d</sub> lower limit	(F/d)				0.019	0.010	0.135			
Predicted K <sub>d</sub> (CF-total 3)	(F/d)				2.844	14.224	2.844			
UF-starting K <sub>d</sub>								1.6	1.6	1.6
UF-CEC								1.4	1.4	1.4
UF-speciation								1.4	1.4	1.4
UF-batch -> compacted								2.0	2.0	2.0
UF-total					000	00000		6.2	6.2	6.2
Kd upper limit	(F/d)				17.669	88.368	17.670			
	(r/g)	_			0.458	2.289	0.458			

Predicted K<sub>4</sub> values for U(VI). Data source: Experimental K<sub>4</sub> values for U(VI) sorption on Saz-1

(a): U(VI) Saz-1 montmorillonite in 0.1 NaNO3 (Pabalan&Turner, 1997) RPW: Reference porewater (saline Beberg water) RPWC: Reference porewater (saline Beberg water) for a closed system HSPW: High saline porewater (Beberg water)

30
n MX-8
rption o
(IV) sol
s for Th
d value
nental K
Experin
source:
V). Data
or Zr(IV
values f
${\it sted} \; K_d$
Predi

Zr(IV), analogy with Th (IV)							Cond	tions					Conversio	on (C) / Unce Factor(F)	rtainty (U)
		Data source (a)	Additio	nal informat	ion (b)	•	dditional inf	ormation (c			Application			Ŀ	
Solution		SBPW	RPW	RPWC	MSH		0.1 M N	aClO4		RPW	RPWC	MSPW	RPW	RPWC	HSPW
Substrate		MX-80	MX-80	MX-80	MX-80	SWy-1	SWy-1	SWy-1	SWy-1	MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
CEC	(meq/100g)	79	85.32	85.32	85.32	87	87	87	87	85.32	85.32	85.32	-	-	٢
pH		7.6	7.377	6.593	7.046	7.6	7.377	6.59	7.046	7.377	6.59	7.046	-	-	-
Kd for 0.1 M NaClO4 (pH conversion) (b)						355	355	224	224				1.0000	0.6310	0.6310
CF-total 1													1.000	0.631	0.631
Speciation												l			
CF-spec 2			0.0018	0.0003	0.0040					0.0582	0.2735	0.1202	32.1	1029.4	30.3
CF-spec 3			1.0000	0.9996	1.0000					0.0582	0.2735	0.1202	0.058	0.274	0.120
CF-total 2													32.12004	649.52566	19.08855
CF-total 3													0.05823	0.17264	0.07583
														Ъ	
Experimental Kd for Th (a)	(b/J)	63.00													
Predicted Kd (CF-total 1)	(b/J)									63.00	39.75	39.75			
UF-starting Kd													4.0	4.0	4.0
UF-pH conversion													2.5	2.5	2.5
UF-batch -> compacted													2.0	2.0	2.0
UF-total													20.0	20.0	20.0
Kd upper limit	(b/T)									1260.0	795.0	795.0			
Kd lower limit	(I/J)									3.15	1.99	1.99			
Predicted Kd (CF-total 2)	(I/J)									2023.56	40920.12	1202.58			
UF-starting Kd													4.0	4.0	4.0
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													28.00	28.00	28.00
Kd upper limit	(6/I)									56659.8	1145763.3	33672.2			
Kd lower limit	(6/T)									72.27	1461.43	42.95			
Predicted Kd (CF-total 3)	(D/)									3.67	10.88	4.78			
UF-starting Kd													4.0	4.0	4.0
UF-pH conversion													2.5	2.5	2.5
UF-speciation													1.4	1.4	1.4
UF-batch -> compacted													2.0	2.0	2.0
UF-total													28.00	28.00	28.00
Kd upper limit	(6/T)									102.7	304.5	133.8			
Kd lower limit	(6/1)									0.13	0.39	0.17			

(a) Data from Bradbury and Baeyens (2003a) for Th
(b) Th speciation for MX-80 application
(c) pH conversion based on data for Th reported by Bradbury and Baeyens (2003a)
RPW: Reference porewater (saline Beberg water)
RPWC: Reference porewater (saline Beberg water)
High saline porewater (Beberg water)

144

## Appendix F

## **Radionuclide speciation tables**

The speciation of each radionuclide was calculated for the main application systems (RPW, RPWC and HSPW) and for the experimental  $K_d$  data source system.

The speciation data for each RN are summarized in the following tables.

The conversion factor (CF) for speciation was calculated following equations 5.3a and 5.3b (see text).

The CFs for speciation were calculated in two different ways:

- 1) Including the RN CO<sub>3</sub> complexes in the sum of the competitive complexes (orange).
- 2) Excluding the RN CO<sub>3</sub> complexes from the sum of the competitive complexes (blue).

## Abbreviation used in the tables:

Dissolved RN complexes which are considered competitive with respect to sorption = RN cmp

Dissolved RN complexes which are considered non competitive with respect to sorption = RN non cmp

Total dissolved RN concentration = RN tot

List of evaluated radionuclides:

- Americium
- Ceasium: no speciation table given
- Carbon: no speciation table given
- · Cerium, holmium, and samarium: derived using Eu as analogue element
- Chloride: no speciation table given
- Curium: no speciation table given
- Iodine: no speciation table given
- Lead
- Neptumium
- Nickel
- Niobium: no speciation table given
- Palladium: no speciation table given

- Plutonium
- Protactinium: no speciation table given
- Radium and strontium
- Radon: no speciation table given
- Selenium
- Silver: no speciation table given
- Technetium
- Thorium
- Tin: no speciation table given
- Uranium
- Zirconium

System			Application			Data source	
Substrate			MX-80			Na-smectite	
Solution		RPW	RPWC	HSPW		0.1 M NaCIO4	
Am <sup>+3</sup> added							
Hq		7.377	6.593	7.046	7.377	6.628	7.136
pCO2		-2.6	-0.98	-2.6	-3.5	-3.5	-3.5
Am <sup>+3</sup> dissolved	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-10	1.00E-10	1.00E-10
Am <sup>+3</sup>	(mol/L)	6.21E-10	6.38E-10	1.49E-09	1.77E-11	6.61E-11	3.68E-11
AmCl <sup>+2</sup>	(mol/L)	1.82E-10	1.86E-10	2.14E-09			
AmCO <sub>3</sub> <sup>+</sup>	(mol/L)	3.52E-09	4.05E-09	2.32E-09	2.07E-11	2.09E-12	9.37E-12
Am(CO <sub>3</sub> ) <sub>2</sub>	(mol/L)	1.13E-10	1.46E-10	1.61E-11	8.41E-14	2.29E-16	8.30E-15
Am(CO <sub>3</sub> ) <sup>3-3</sup>	(mol/L)	2.46E-10	1.49E-12	2.52E-14	5.96E-17	4.39E-21	1.28E-18
Am(SO <sub>4</sub> ) <sup>+</sup>	(mol/L)	1.02E-12	3.60E-09	2.06E-09			
Am(SO4)2 <sup>-</sup>	(mol/L)	8.91E-10	1.15E-09	1.27E-10			
$\Sigma$ competitive Am complexes							
(Am-cmp)	(mol/L)	4.95E-09	9.14E-09	6.66E-09	2.08E-11	2.09E-12	9.38E-12
Am tot-Am-cmp	(mol/L)	5.049E-09	8.627E-10	3.34E-09	7.92E-11	9.79E-11	9.06E-11
(Am tot- (Am-cmp)) / Am tot	(mol/L)	0.505	0.086	0.334	0.79	0.98	0.91
Am <sup>+3</sup> dissolved	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-10	1.00E-10	1.00E-10
Am <sup>+3</sup>	(mol/L)	6.21E-10	6.38E-10	1.49E-09	1.77E-11	6.61E-11	3.68E-11
AmCl <sup>+2</sup>	(mol/L)	1.82E-10	1.86E-10	2.14E-09			
AmCO <sub>3</sub> <sup>+</sup>	(mol/L)	3.52E-09	4.05E-09	2.32E-09	2.07E-11	2.09E-12	9.37E-12
Am(CO <sub>3</sub> ) <sub>2</sub>	(mol/L)	1.13E-10	1.46E-10	1.61E-11	8.41E-14	2.29E-16	8.30E-15
Am(CO <sub>3</sub> ) <sup>3-3</sup>	(mol/L)	2.46E-10	1.49E-12	2.52E-14	5.96E-17	4.39E-21	1.28E-18
Am(SO <sub>4</sub> ) <sup>+</sup>	(mol/L)	1.02E-12	3.60E-09	2.06E-09			
Am(SO4) <sub>2</sub>	(mol/L)	8.91E-10	1.15E-09	1.27E-10			
∑ competitive Am complexes (Am-cmp)	(mol/L)	1.07E-09	4.94E-09	4.33E-09	0.00E+00	0.00E+00	0.00E+00
Am tot-Am-cmp	(mol/L)	8.926E-09	5.060E-09	5.67E-09	1.00E-10	1.00E-10	1.00E-10
(Am tot- (Am-cmp)) / Am tot	(mol/L)	0.893	0.506	0.567	1.00	1.00	1.00

Am(III) speciation (TDB Silva et al., 1995)

System			Application			Data source	
Substrate			MX-80			Na-smectite	
Solution		RPW	RPWC	HSPW		0.1 M NaCIO4	
Am <sup>+3</sup> added		1.00E-08	1.00E-08	1.00E-08	1.00E-10	1.00E-10	1.00E-10
Hq		7.377	6.593	7.046	7.377	6.628	7.136
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5	-3.5	-3.5
Am <sup>+3</sup> dissolved	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-10	1.00E-10	1.00E-10
Am⁺ <sup>3</sup>	(mol/L)	7.80E-10	6.48E-10	2.25E-09	2.92E-11	8.83E-11	6.07E-11
AmCl <sup>+2</sup>	(mol/L)	3.53E-11	2.93E-11	4.99E-10			
AmCl <sub>2</sub> <sup>+1</sup>	(mol/L)	1.77E-13	1.46E-13	1.15E-11			
AmcO <sub>3</sub> <sup>+</sup>	(mol/L)	7.00E-09	6.53E-09	5.53E-09	5.40E-11	4.42E-12	2.45E-11
Am(CO <sub>3</sub> ) <sup>2</sup>	(mol/L)	5.64E-10	5.90E-10	9.67E-11	5.52E-13	1.22E-15	5.44E-14
Am(CO <sub>3</sub> ) <sup>3</sup>	(mol/L)	8.11E-13	9.56E-13	2.39E-14	6.20E-17	3.71E-21	1.33E-18
AmHCO <sub>3</sub> <sup>+2</sup>	(mol/L)	1.96E-10	1.11E-09	3.05E-10	1.26E-12	6.28E-13	1.23E-12
Am(SO₄) <sup>+</sup>	(mol/L)	1.11E-09	1.03E-09	8.75E-10			
Am(SO4)2 <sup>-</sup>	(mol/L)	2.23E-11	2.34E-11	3.83E-12			
AmOSi(OH) <sub>3</sub> <sup>+2</sup>	(mol/L)	1.71E-15	1.42E-15	4.92E-15			
∑ competitive Am complexes (Am-cmb)	( I/lom)	8.93E-09	9.32F-09	7.33E-09	5.59E-11	5.05E-12	2.58E-11
	()						
Am tot-Am-cmp	(mol/L)	1.070E-09	6.842E-10	2.67E-09	4.41E-11	9.49E-11	7.42E-11
(Am tot- (Am-cmp)) / Am tot	(mol/L)	0.11	0.07	0.27	0.44	0.95	0.74
AmCl <sup>+2</sup>	(mol/L)	3.53E-11	2.93E-11	4.99E-10			
AmCl <sub>2</sub> <sup>+1</sup>	(mol/L)	1.77E-13	1.46E-13	1.15E-11			
AmCO <sub>3</sub> <sup>+</sup>	(mol/L)	7.00E-09	6.53E-09	5.53E-09	5.40E-11	4.42E-12	2.45E-11
Am(CO <sub>3</sub> ) <sup>2</sup>	(mol/L)	5.64E-10	5.90E-10	9.67E-11	5.52E-13	1.22E-15	5.44E-14
Am(CO <sub>3</sub> ) <sup>3</sup>	(mol/L)	8.11E-13	9.56E-13	2.39E-14	6.20E-17	3.71E-21	1.33E-18
AmHCO <sub>3</sub> <sup>+2</sup>	(mol/L)	1.96E-10	1.11E-09	3.05E-10	1.26E-12	6.28E-13	1.23E-12
Am(SO₄) <sup>+</sup>	(mol/L)	1.11E-09	1.03E-09	8.75E-10			
Am(SO4)2 <sup>-</sup>	(mol/L)	2.23E-11	2.34E-11	3.83E-12			
AmOSi(OH) <sub>3</sub> * <sup>2</sup>	(mol/L)	1.71E-15	1.42E-15	4.92E-15			
$\Sigma$ competitive Am complexes							
(Am-cmp)	(mol/L)	1.17E-09	1.08E-09	1.39E-09	0.00E+00	0.00E+00	0.00E+00
Am tot-Am-cmp	(mol/L)	8.835E-09	8.915E-09	8.61E-09	1.00E-10	1.00E-10	1.00E-10
(Am tot- (Am-cmp)) / Am tot	(mol/L)	0.88	0.89	0.86	1.00	1.00	1.00

Am(III) speciation (TDB Guillaumont et al., 2003)

System		Data source		Application		Data source		Application	
Substrate		MX-80		MX-80		08-XM		MX-80	
Solution		SBPW	RPW	RPWC	HSPW	SBPW	RPW	RPWC	HSPW
Eu <sup>+3</sup> added	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10
Ha		7.6	7.377	6.593	7.046	7.6	7.377	6.593	7.046
pCO <sub>2</sub>		-3.5	-2.6	-0.98	-2.6	-3.5	-2.6	-0.98	-2.6
Eu <sup>+3</sup> dissolved	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10	1.00E-10
Eu⁺3	(mol/L)	8.79E-12	4.51E-12	4.01E-12	1.16E-11	8.79E-12	4.51E-12	4.01E-12	1.16E-11
Eu(OH) <sup>4</sup>	(mol/L)	1.46E-18	8.15E-20	5.25E-23	1.25E-20	1.46E-18	8.15E-20	5.25E-23	1.25E-20
EuCO <sub>3</sub> <sup>+</sup>	(mol/L)	4.17E-11	5.10E-11	5.09E-11	3.61E-11	4.17E-11	5.10E-11	5.09E-11	3.61E-11
Eu(CO <sub>3</sub> ) <sup>2</sup>	(mol/L)	1.50E-13	5.17E-13	5.79E-13	7.93E-14	1.50E-13	5.17E-13	5.79E-13	7.93E-14
EuSO4 <sup>+</sup>	(mol/L)	2.34E-11	2.86E-11	2.85E-11	2.02E-11	2.34E-11	2.86E-11	2.85E-11	2.02E-11
Eu(SO <sub>4</sub> ) <sup>2</sup>	(mol/L)	3.74E-12	1.29E-11	1.45E-11	1.98E-12	3.74E-12	1.29E-11	1.45E-11	1.98E-12
EuCl <sup>+2</sup>	(mol/L)	1.26E-11	1.48E-12	1.31E-12	1.87E-11	1.26E-11	1.48E-12	1.31E-12	1.87E-11
EuCI <sub>2</sub> <sup>+</sup>	(mol/L)	6.37E-12	1.78E-13	1.57E-13	1.04E-11	6.37E-12	1.78E-13	1.57E-13	1.04E-11
EuH <sub>3</sub> SiO <sub>4</sub> <sup>+2</sup>	(mol/L)	4.09E-19	1.13E-19	1.65E-20	1.58E-19	4.09E-19	1.13E-19	1.65E-20	1.58E-19
∑ competitive Eu complexes (Eu- cmp)	( 1/lom)	8.80F-11	9.48F-11	9.59E-11	8.75E-11	4.61E-11	4.32E-11	4.45E-11	5.13E-11
	()								
Eu tot- (Eu-cmp)		1.20E-11	5.25E-12	4.10E-12	1.25E-11	5.39E-11	5.68E-11	5.55E-11	4.87E-11
(Eu tot- (Eu-cmp)) / Eu tot		0.120	0.052	0.041	0.125	0.539	0.568	0.555	0.487

Eu(III) speciation (TDB from Hummel et al., 2002)

System			Application			Data Source	
Substrate			MX-80		Montmorillon	ite (Ulrich & Degu	ueldre, 1993)
Solution		RPW	RPWC	HSPW		0.1 M NaCIO <sub>4</sub>	
Pb <sup>+2</sup> added	(mol/L)	1.00E-11	1.00E-11	1.00E-11	1.00E-11	1.00E-11	1.00E-11
Hq		7.377	6.593	7.046	7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5	-3.5	-3.5
Pb <sup>+2</sup> dissolved	(mol/L)	1.00E-11	1.00E-11	1.00E-11	1.00E-11	1.00E-11	1.00E-11
Pb <sup>+2</sup>	(mol/L)	9.00E-12	8.48E-13	3.79E-13	4.29E-12	9.22E-12	7.25E-12
PbOH	(mol/L)	2.67E-13	4.13E-14	5.73E-14	1.53E-12	5.40E-13	1.21E-12
Pb(OH)2	(mol/L)	1.71E-15	4.33E-17	1.76E-16	1.04E-14	6.03E-16	3.82E-15
PbCO <sub>3</sub> <sup>+</sup>	(mol/L)	5.42E-12	5.73E-12	5.57E-13	4.17E-12	2.43E-13	1.54E-12
PbSO4 <sup>+</sup>	(mol/L)	1.42E-12	1.50E-12	1.46E-13			
PbCI <sup>+</sup>	(mol/L)	1.52E-12	1.43E-12	2.97E-12			
PbC12	(mol/L)	3.76E-13	3.54E-13	3.19E-12			
Pb(OH) <sup>3</sup>	(mol/L)	4.69E-19	1.95E-21	2.18E-20	2.68E-18	2.56E-20	4.60E-19
Pb(CO <sub>3</sub> )2 <sup>-4</sup>	(mol/L)	1.25E-14	1.48E-14	2.48E-16	9.55E-16	1.50E-18	7.67E-17
Pb(SO <sub>4</sub> ) <sup>2<sup>-2</sup></sup>	(mol/L)	1.36E-14	1.62E-14	2.70E-16			
PbCI <sub>3</sub>	(mol/L)	6.17E-14	5.80E-14	2.16E-12			
PbCI4 <sup>-2</sup>	(mol/L)	3.97E-15	3.73E-15	5.38E-13			
7 compatitive Dh complexes							
(Pb-cmp)	(mol/L)	9.17E-14	9.27E-14	2.69E-12	9.55E-16	1.50E-18	7.67E-17
Pb tot-(Pb-cmp)	(mol/L)	9.908E-12	9.907E-12	7.305E-12	9.998E-12	1.000E-11	1.000E-11
(Pb tot- (Pb-cmp)) / Pb tot		0.9908	0.9907	0.7305	0.9999	1.0000	1.0000

1999)
t al.,
Yui et
TDB
speciation (
Pb(II) \$

Svstem			Application			Data Source	
Substrate			MX-80		Montmorillon	ite (Ulrich & Degu	ueldre, 1993)
Solution		RPW	RPWC	HSPW		0.1 M NaCIO4	
Pb <sup>+2</sup> added	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-11	1.00E-11	1.00E-11
Hq		7.377	6.593	7.046	7.377	6.593	7.046
pCO2		-2.6	-0.98	-2.6	-3.5	-3.5	-3.5
Pb <sup>+2</sup> dissolved	(mol/L)	1.00E-11	1.00E-11	1.00E-11	1.00E-11	1.00E-11	1.00E-11
Pb <sup>+2</sup>	(mol/L)	9.00E-12	8.48E-13	3.79E-13	4.29E-12	9.22E-12	7.25E-12
PbOH <sup>+</sup>	(mol/L)	2.67E-13	4.13E-14	5.73E-14	1.53E-12	5.40E-13	1.21E-12
Pb(OH)2	(mol/L)	1.71E-15	4.33E-17	1.76E-16	1.04E-14	6.03E-16	3.82E-15
PbC0 <sub>3</sub> <sup>+</sup>	(mol/L)	5.42E-12	5.73E-12	5.57E-13	4.17E-12	2.43E-13	1.54E-12
PbSO4 <sup>+</sup>	(mol/L)	1.42E-12	1.50E-12	1.46E-13			
PbCl <sup>+</sup>	(mol/L)	1.52E-12	1.43E-12	2.97E-12			
PbC12	(mol/L)	3.76E-13	3.54E-13	3.19E-12			
Pb(OH) <sup>3</sup>	(mol/L)	4.69E-19	1.95E-21	2.18E-20	2.68E-18	2.56E-20	4.60E-19
Pb(CO <sub>3</sub> ), <sup>2</sup>	(mol/L)	1.25E-14	1.48E-14	2.48E-16	9.55E-16	1.50E-18	7.67E-17
Pb(S0 <sub>4</sub> )2 <sup>-2</sup>	(mol/L)	1.36E-14	1.62E-14	2.70E-16			
PbCl <sub>3</sub>	(mol/L)	6.17E-14	5.80E-14	2.16E-12			
PbCI4 <sup>-2</sup>	(mol/L)	3.97E-15	3.73E-15	5.38E-13			
C competitive Dh complexes							
Z competitive ru complexes	(mol/L)	7.93E-14	7.79E-14	2.69E-12	0.00E+00	0.00E+00	0.00E+00
Pb tot-(Pb-cmp)	(mol/L)	9.921E-12	9.922E-12	7.305E-12	9.999E-12	1.000E-11	1.000E-11
(Pb tot- (Pb-cmp)) / Pb tot		0.9921	0.9922	0.7305	1.0000	1.0000	1.0000

Pb(II) speciation (TDB Yui et al., 1999)

C. indexe			Amilantian		Date Course
			Application		
Substrate			MX-80		MX-80
Solution		RPW	RPWC	HSPW	SBPW
Np <sup>+4</sup> added	(mol/L)	1.00E-09	1.00E-09	1.00E-09	1.00E-09
Hq		7.377	6.593	7.046	7.6
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5
Np <sup>+4</sup> dissolved	(mol/L)	1.00E-09	1.00E-09	1.00E-09	1.00E-09
Np <sup>+4</sup>	(mol/L)	2.49E-27	3.34E-24	3.31E-26	2.28E-28
Np(OH) <sup>+3</sup>	(mol/L)	3.65E-21	8.04E-19	2.78E-20	6.47E-22
Np(OH) <sub>4</sub>	(mol/L)	1.00E-09	9.72E-10	1.00E-09	1.00E-09
Np(CO <sub>3</sub> ) <sub>4</sub> <sup>4</sup>	(mol/L)	1.28E-14	2.75E-11	3.79E-16	1.85E-17
Np(CO <sub>3</sub> )5 <sup>4</sup>	(mol/L)	4.73E-19	1.15E-15	1.69E-21	1.58E-22
Np(SO <sub>4</sub> ) <sub>2</sub>	(mol/L)	1.41E-22	2.39E-19	1.41E-22	2.28E-24
NpSO4 <sup>+2</sup>	(mol/L)	3.73E-24	5.62E-21	1.53E-23	1.56E-25
NpCI	(mol/L)	1.12E-27	1.50E-24	7.73E-26	4.68E-28
∑ competitive Np(IV) complexes (Np(IV) cmp)	(mol/L)	1.28E-14	2.75E-11	3.79E-16	1.85E-17
Np(IV) -tot- Np(IV) cmp		1.00E-09	9.72E-10	1.00E-09	1.00E-09
(Np(IV) tot- Np(IV) cmp / Np(IV) tot		0.9999	0.9725	1.0000	1.0000
Nn <sup>+4</sup> dissolved	( 1/lom)	1 00F-09	1 00F-09	1 00F-09	1 00F-09
N <sup>44</sup>	(mol/L)	2.49E-27	3.34E-24	3.31E-26	2.28E-28
Np(OH) <sup>+3</sup>	(mol/L)	3.65E-21	8.04E-19	2.78E-20	6.47E-22
Np(OH) <sub>4</sub>	(mol/L)	1.00E-09	9.72E-10	1.00E-09	1.00E-09
Np(CO <sub>3</sub> ) <sub>4</sub> <sup>4</sup>	(mol/L)	1.28E-14	2.75E-11	3.79E-16	1.85E-17
Np(CO <sub>3</sub> )5 <sup>4</sup>	(mol/L)	4.73E-19	1.15E-15	1.69E-21	1.58E-22
Np(SO <sub>4</sub> ) <sub>2</sub>	(mol/L)	1.41E-22	2.39E-19	1.41E-22	2.28E-24
NpSO4 <sup>+2</sup>	(mol/L)	3.73E-24	5.62E-21	1.53E-23	1.56E-25
NpCI	(mol/L)	1.12E-27	1.50E-24	7.73E-26	4.68E-28
∑ competitive Np(IV) complexes		1 45E-22	2 44F-19	1 57E-22	2 44F-24
(d		77-JOF-1	01-J++-3	1.01 1-44	17-71-73
Np(IV) -tot- Np(IV) cmp		1.00E-09	1.00E-09	1.00E-09	1.00E-09
(Np(IV) tot- Np(IV) cmp / Np(IV) tot		0.9999	1.0000	1.0000	1.0000

NpIV speciation (TDB Hummel et al., 2002)

Svstem			Application			Data Source	
Substrate			MX-80		Na-n	nontmorillonite	(Saz-1)
Solution		RPW	RPWC	HSPW		0.1 NaNO3	
NpO <sub>2</sub> <sup>+</sup> added	(mol/L)	1.00E-07	1.00E-07	1.00E-07	1.00E-07	1.00E-07	1.00E-07
Hd		7.377	6.593	7.136	7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5	-3.5	-3.5
NpO <sub>2</sub> <sup>+</sup> dissoved	(mol/L)	1.00E-07	1.00E-07	1.00E-07	1.00E-07	1.00E-07	1.00E-07
NpO <sub>2</sub> <sup>+</sup>	(mol/L)	8.97E-08	8.86E-08	9.76E-08	9.48E-08	9.48E-08	9.48E-08
NpO <sub>2</sub> (OH)	(mol/L)	7.92E-12	1.28E-12	4.13E-12	8.88E-12	1.46E-12	4.15E-12
NpO <sub>2</sub> (OH) <sub>2</sub>	(mol/L)	1.28E-16	3.41E-18	3.02E-17	1.35E-16	3.65E-18	2.94E-17
NpO <sub>2</sub> SO <sub>4</sub>	(mol/L)	1.98E-09	2.20E-09	4.68E-10			
NpO <sub>2</sub> NO <sub>3</sub>	(mol/L)				5.21E-09	5.21E-09	5.21E-09
NpO <sub>2</sub> (OH)(CO <sub>2</sub> ) <sub>2</sub> <sup>4</sup>	(mol/L)	1.03E-15	2.13E-16	1.60E-17	2.65E-60	6.61E-63	2.34E-61
NpO <sub>2</sub> (CO <sub>2</sub> )	(mol/L)	8.29E-09	9.20E-09	1.96E-09	6.81E-31	8.38E-32	2.95E-31
NpO <sub>2</sub> (CO <sub>2</sub> ) <sub>2</sub> <sup>-3</sup>	(mol/L)	3.52E-12	4.40E-12	1.43E-13	1.38E-56	2.10E-58	2.61E-57
NpO <sub>2</sub> (CO2) <sub>3</sub> <sup>-5</sup>	(mol/L)	4.24E-17	5.98E-17	2.34E-19	4.89E-84	9.14E-87	4.01E-85
$\Sigma$ competitive NpO <sub>2</sub> <sup>+</sup> complexes							
(NpO2+ cmp)	(mol/L)	1.03E-08	1.14E-08	2.43E-09	5.21E-09	5.21E-09	5.21E-09
$(NpO_2^{+} tot) - (NpO_2^{+} cmp)$	(mol/L)	8.97E-08	8.86E-08	9.76E-08	9.48E-08	9.48E-08	9.48E-08
$(NpO_2^+ \text{ tot- } (NpO_2^+ \text{ cmp})) / NpO_2^+ \text{ tot}$		0.897244	0.885968	0.975747	0.947946	0.947942	0.947944
NpO <sub>2</sub> SO <sub>4</sub>	(mol/L)	1.98E-09	2.20E-09	4.68E-10			
NpO <sub>2</sub> NO <sub>3</sub>	(mol/L)				5.21E-09	5.21E-09	5.21E-09
NpO <sub>2</sub> (OH)(CO <sub>2</sub> ) <sub>2</sub> <sup>-4</sup>	(mol/L)	1.03E-15	2.13E-16	1.60E-17	2.65E-60	6.61E-63	2.34E-61
NPO <sub>2</sub> (CO <sub>2</sub> )	(mol/L)	8.29E-09	9.20E-09	1.96E-09	6.81E-31	8.38E-32	2.95E-31
NpO <sub>2</sub> (CO <sub>2</sub> ) <sub>2</sub> <sup>-3</sup>	(mol/L)	3.52E-12	4.40E-12	1.43E-13	1.38E-56	2.10E-58	2.61E-57
NpO <sub>2</sub> (CO2) <sub>3</sub> -5	(mol/L)	4.24E-17	5.98E-17	2.34E-19	4.89E-84	9.14E-87	4.01E-85
$\sum$ competitive NpO <sub>2</sub> <sup>+</sup> complexes							
(NpO2+ cmp)	(mol/L)	1.98E-09	2.20E-09	4.68E-10	5.21E-09	5.21E-09	5.21E-09
$(NpO_2^{+} tot) - (NpO_2^{+} cmp)$	(mol/L)	9.80E-08	9.78E-08	9.95E-08	9.48E-08	9.48E-08	9.48E-08
$(NpO_2^+ \text{ tot- } (NpO_2^+ \text{ cmp})) / NpO_2^+ \text{ tot}$		0.980163	0.977987	0.995317	0.947946	0.947942	0.947944

Np(V) speciation (TDB Hummel et al., 2002)

al., 2002)
et i
Hummel
(TDB
speciation
Ni(II)

Substrate			Application		Data source
Substrate			MX-80		MX-80
Solution		RPW	RPWC	HSPW	SBPW
Ni <sup>+2</sup> added	(mol/L)	1.00E-07	1.00E-07	1.00E-07	1.00E-07
Hd		7.451	6.628	7.136	7.6
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5
Ni <sup>+2</sup> dissolved	(mol/L)	1.000E-07	1.00E-07	1.000E-07	1.00E-07
Ni <sup>+2</sup>	(mol/L)	5.70E-08	5.44E-08	4.08E-08	4.24E-08
Ni(OH) <sub>3</sub>	(mol/L)	6.21E-16	2.62E-18	4.91E-17	2.29E-15
Ni(OH) <sub>4</sub> <sup>-2</sup>	(mol/L)	2.32E-23	1.61E-26	7.83E-25	1.34E-22
NICO3	(mol/L)	1.72E-10	1.84E-10	3.00E-11	4.94E-11
NI(CO <sub>3</sub> )2 <sup>-2</sup>	(mol/L)	5.85E-14	7.05E-14	1.98E-15	5.47E-15
NISO4	(mol/L)	2.78E-08	2.98E-08	4.86E-09	7.99E-09
Ni(SO <sub>4</sub> )2 <sup>-2</sup>	(mol/L)	5.82E-09	7.02E-09	1.97E-10	5.44E-10
NICI+	(mol/L)	6.83E-09	6.52E-09	2.26E-08	2.14E-08
NICI2	(mol/L)	2.17E-09	2.07E-09	3.14E-08	2.74E-08
$\Sigma$ competitive Ni complexes					
(Ni-cmp)	(mol/L)	4.28E-08	4.56E-08	5.9089E-08	5.73E-08
Ni tot-Ni cmp with Ni-CO <sub>3</sub>	(mol/L)	5.72E-08	5.44E-08	4.09E-08	4.27E-08
(Ni tot- (Ni-cmp)) / Ni tot		0.572	0.5442	0.4091	0.427
Ni <sup>+2</sup> dissolved	(mol/L)	1.000E-07	1.00E-07	1.000E-07	1.00E-07
Ni <sup>+2</sup>	(mol/L)	5.70E-08	5.44E-08	4.08E-08	4.24E-08
Ni(OH) <sub>3</sub>	(mol/L)	6.21E-16	2.62E-18	4.91E-17	2.29E-15
Ni(OH) <sub>4</sub> <sup>-2</sup>	(mol/L)	2.32E-23	1.61E-26	7.83E-25	1.34E-22
NICO3	(mol/L)	1.72E-10	1.84E-10	3.00E-11	4.94E-11
Ni(CO <sub>3</sub> )2 <sup>-2</sup>	(mol/L)	5.85E-14	7.05E-14	1.98E-15	5.47E-15
NISO4	(mol/L)	2.78E-08	2.98E-08	4.86E-09	7.99E-09
Ni(SO <sub>4</sub> )2 <sup>-2</sup>	(mol/L)	5.82E-09	7.02E-09	1.97E-10	5.44E-10
NICH	(mol/L)	6.83E-09	6.52E-09	2.26E-08	2.14E-08
NICI2	(mol/L)	2.17E-09	2.07E-09	3.14E-08	2.74E-08
(Ni-cmp)	(mol/L)	4.26E-08	4.54E-08	5.91E-08	5.73E-08
Ni tot-Ni cmp without Ni-CO <sub>3</sub>	(mol/L)	5.736E-08	5.461E-08	4.094E-08	4.273E-08
(Ni tot- (Ni-cmp)) / Ni tot		0.574	0.546	0.409	0.427

System			Application			Data Source	
Substrate			MX-80			Na-smectite	
Solution		RPW	RPWC	HSPW		0.1 M NaCIO4	
Pu <sup>+3</sup> added	(mol/L)	1.00E-09	1.00E-09	1.00E-09	1.00E-07	1.00E-07	1.00E-07
Ha		7.377	6.628	7.046	7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5	-3.5	-3.5
precipitated solid							
Pu <sup>+3</sup> total	(mol/L)	1.00E-09	1.00E-09	1.00E-09	1.00E-07	1.00E-07	1.00E-07
Pu⁺³		9.45E-11	8.72E-11	1.96E-10	5.28E-08	8.72E-08	7.06E-08
Pu(OH) <sup>+2</sup>	(mol/L)	6.23E-11	9.44E-12	6.98E-11	4.72E-08	1.28E-08	2.94E-08
PuSO <sub>4</sub> <sup>+</sup>	(mol/L)	2.70E-10	3.15E-10	3.34E-11			
Pu(SO <sub>4</sub> )2 <sup>-</sup>		5.34E-10	5.53E-10	3.04E-10			
PuCl <sup>+2</sup>	(mol/L)	3.90E-11	3.59E-11	3.97E-10			
∑ competitive Pu(III) complexes (Pu(III) cmp)	(mol/L)	8.43E-10	9.03E-10	7.34E-10	0.00E+00	0.00E+00	0.00E+00
Pu(III) tot - Pu(III) cmp	(mol/L)	1.57E-10	9.67E-11	2.66E-10	1.00E-07	1.00E-07	1.00E-07
(Pu(III) tot - Pu(III) cmp) / Pu(III) tot		0.1568	0.0967	0.2659	1.0000	1.0000	1.0000

et al., 2002)
<b>DB</b> Hummel
speciation (T
Pu(III)
motor.
---
ayatelli
Substrate
Solution
Pu <sup>+4</sup> added
HO
pCO <sub>2</sub>
precipitated solid
Pu <sup>+4</sup> dissolved
Pu <sup>+4</sup>
Pu(OH) <sup>+3</sup>
Pu(OH)2 <sup>+2</sup>
Pu(OH) <sub>3</sub> <sup>+</sup>
Pu(OH)4
- "(OH)"
Pu(OH),CO3.
PuSO4 <sup>+2</sup>
Pu(SO <sub>4</sub> ) <sup>2</sup>
Pu(SO <sub>4</sub> ) <sub>3</sub> <sup>-2</sup>
PuCl <sup>+3</sup>
PuCI <sub>2</sub> *2
Pu(CO <sub>3</sub> )* <sup>2</sup>
Pu(CO <sub>3</sub> ) <sub>2</sub>
Pu(CO <sub>3</sub> ) <sub>3</sub> <sup>-2</sup>
Pu(CO <sub>3</sub> ) <sup>4</sup>
Pu(CO <sub>3</sub> ) <sub>5</sub> <sup>-6</sup>
∑ competitive Pu(IV) complexes
(Pu(IV) cmp)
Pu(IV) tot - Pu(IV) cmp
(Pu(IV) tot- Pu(IV) cmp) / Pu(IV) tot

Pu(IV) Speciation (TDB Hummel et al., 2002)

System			Application	
Substrate			MX-80	
Solution		RPW	RPWC	HSPW
PuO <sub>2</sub> <sup>+</sup> added	(mol/L)	1.00E-08	1.00E-08	1.00E-08
Hď		7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6
PuO <sub>2</sub> <sup>+</sup> dissolved	(mol/L)	1.00E-08	1.00E-08	1.00E-08
PuO <sub>2</sub> <sup>+</sup>		8.80E-09	8.69E-09	9.70E-09
PuO <sub>2</sub> (OH)	(mol/L)	2.88E-11	4.68E-12	1.53E-11
PuO <sub>2</sub> (CO <sub>3</sub> ) <sub>3</sub> <sup>-5</sup>	(mol/L)	1.31E-18	1.85E-18	7.36E-21
PuO <sub>2</sub> (CO <sub>3</sub> )	(mol/L)	1.17E-09	1.30E-09	2.81E-10
$\Sigma$ competitive PuO <sub>2</sub> <sup>+</sup> complexes				
(PuO <sub>2</sub> <sup>+</sup> cmp)	(mol/L)	1.17E-09	1.30E-09	2.81E-10
(PuO <sub>2</sub> <sup>+</sup> tot)-(PuO <sub>2</sub> <sup>+</sup> cmp)	(mol/L)	8.83E-09	8.70E-09	9.72E-09
(PuO $_2^+$ tot- (PuO $_2^+$ cmp)) / PuO $_2^+$ tot		0.8825	0.8696	0.9719
PuO <sub>2</sub> (CO <sub>3</sub> ) <sub>3</sub> -5	(mol/L)	1.31E-18	1.85E-18	7.36E-21
PuO <sub>2</sub> (CO <sub>3</sub> ) <sup>-</sup>	(mol/L)	1.17E-09	1.30E-09	2.81E-10
$\Sigma$ competitive PuO <sub>2</sub> <sup>+</sup> complexes				
(PuO <sub>2</sub> <sup>+</sup> cmp)	(mol/L)	0.00E+00	0.00E+00	0.00E+00
(PuO <sub>2</sub> <sup>+</sup> tot)-(PuO <sub>2</sub> <sup>+</sup> cmp)	(mol/L)	1.00E-08	1.00E-08	1.00E-08
(PuO <sub>2</sub> <sup>+</sup> tot- (PuO <sub>2</sub> <sup>+</sup> cmp)) / PuO <sub>2</sub> <sup>+</sup> tot		1.0000	1.0000	1.0000

Pu(V) speciation (Hummel et al., 2002)

System			Application	
Substrate			MX-80	
Solution		RPW	RPWC	HSPW
PuO <sub>2</sub> <sup>+2</sup> added	(mol/L)	7.50E-09	7.50E-09	7.50E-09
Hq		7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6
PuO <sub>2</sub> <sup>+2</sup> dissolved	(mol/L)	7.50E-09	7.50E-09	7.50E-09
PuO <sub>2</sub> *2	(mol/L)	6.95E-12	6.08E-12	3.99E-11
PuO <sub>2</sub> OH <sup>+</sup>	(mol/L)	2.11E-10	3.03E-11	6.17E-10
PuO <sub>2</sub> (OH) <sub>2</sub>	(mol/L)	7.42E-11	1.75E-12	1.04E-10
PuO2CO3	(mol/L)	4.18E-09	4.10E-09	5.86E-09
PuO <sub>2</sub> SO <sub>4</sub>	(mol/L)	3.98E-11	3.91E-11	5.58E-11
PuO <sub>2</sub> (SO <sub>4</sub> ) <sup>-2</sup>	(mol/L)	1.12E-11	1.24E-11	3.05E-12
PuO <sub>2</sub> (CO <sub>3</sub> ) <sup>22</sup>	(mol/L)	2.84E-09	3.13E-09	7.69E-10
PuO <sub>2</sub> (CO <sub>3</sub> ) <sub>3</sub> <sup>4</sup>	(mol/L)	1.37E-10	1.71E-10	5.69E-12
PuO <sub>2</sub> CI <sup>+</sup>	(mol/L)	1.66E-12	1.45E-12	4.42E-11
PuO <sub>2</sub> Cl <sub>2</sub>	(mol/L)	7.28E-15	6.36E-15	8.44E-13
$\Sigma$ competitive PuO <sub>2</sub> <sup>+2</sup> complexes				
(PuO <sub>2</sub> <sup>+2</sup> cmp)	(mol/L)	3.03E-09	3.36E-09	8.78E-10
(PuO <sub>2</sub> <sup>+2</sup> tot)-(PuO <sub>2</sub> <sup>+2</sup> cmp)	(mol/L)	4.47E-09	4.14E-09	6.62E-09
$(PuO_2^{+2} tot)-(PuO_2^{+2} cmp) / PuO_2^{+2} tot$		0.596290	0.55224956	0.88289
PuO <sub>2</sub> SO <sub>4</sub>	(mol/L)	3.98E-11	3.91E-11	5.58E-11
PuO <sub>2</sub> (SO <sub>4</sub> ) <sup>2<sup>2</sup></sup>	(mol/L)	1.12E-11	1.24E-11	3.05E-12
PuO <sub>2</sub> (CO <sub>3</sub> ) <sup>22</sup>	(mol/L)	2.84E-09	3.13E-09	7.69E-10
PuO <sub>2</sub> (CO <sub>3</sub> ) <sub>3</sub> <sup>-4</sup>	(mol/L)	1.37E-10	1.71E-10	5.69E-12
PuO <sub>2</sub> Cl <sup>+</sup>	(mol/L)	1.66E-12	1.45E-12	4.42E-11
PuO <sub>2</sub> Cl <sub>2</sub>	(mol/L)	7.28E-15	6.36E-15	8.44E-13
$\Sigma$ competitive PuO <sub>2</sub> <sup>+2</sup> complexes				
(PuO <sub>2</sub> <sup>+2</sup> cmp)	(mol/L)	5.27E-11	5.30E-11	1.04E-10
(PuO <sub>2</sub> <sup>+2</sup> tot)-(PuO <sub>2</sub> <sup>+2</sup> cmp)	(mol/L)	7.45E-09	7.45E-09	7.40E-09
$(PuO_2^{+2} \text{ tot})-(PuO_2^{+2} \text{ cmp}) / PuO_2^{+2} \text{ tot}$		0.992968	0.992936	0.986148

Pu(VI) speciation (Hummel et al., 2002)

Se(IV) speciation (TDB Grenthe et	t al., 199(	0 and Pearson	ı et al., 1991)				
System			Data source			Application	
Substrate			NaSWy-1		MX-80	MX-80	MX-80
Solution			0.1 M NaCIO4		RPW	RPWC	HSPW
Se <sup>+4</sup> added	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-07	1.00E-07	1.00E-07
Hq		7.377	6.593	7.046	7.377	6.593	7.046
pCO <sub>2</sub>		-3.5	-3.5	-3.5	-2.6	-0.98	-2.6
SeO <sub>3</sub> <sup>-2</sup> dissolved	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-07	1.00E-07	1.00E-07
H <sub>2</sub> SeO <sub>3</sub>	(mol/L)	1.74E-13	1.22E-12	4.08E-13	1.58E-12	1.15E-11	3.93E-12
SeO <sub>3</sub> -2		1.64E-09	3.12E-10	8.38E-10	1.91E-08	3.72E-09	9.14E-09
HSeO <sub>3</sub> <sup>-</sup>	(mol/L)	8.36E-09	9.69E-09	9.16E-09	8.09E-08	9.63E-08	9.09E-08
$\sum competitive se complexes$							
(Se-cmp)	(mol/L)	8.36E-09	9.69E-09	9.16E-09	8.09E-08	9.63E-08	9.09E-08
(Se(IV) tot - Se(IV)-cmp / Se(IV) tot		0.8360	0.9686	0.9161	0.8094	0.9626	0.9086

_
Ξ
66
Ť
<u> </u>
a
et
Ę
So
a
ě
<u> </u>
2
a
90
ő
ς.
١Ľ
ٽ.
Ð
ĥ
Ľ
Ð
Ū
ш
<u>0</u>
E
Z
Ë
a
S
ă
S
$\geq$
e(I
_

, 2003
et al.
Guillaumont
(TDB
speciation
Tc(IV)

System			Application		Data Source
Substrate			MX-80		MX-80
Solution		RPW	RPWC	HSPW	SBPW
TcO <sup>+2</sup> added	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10
Ha		7.377	6 593	7 046	7.6
			000	30.0	2 7
pcO2		0.7-	-0.98	Q.2-	0.2-
TcO <sup>+2</sup> dissolved	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10
TcO <sup>+2</sup>	(mal/L)	1.45E-25	3.29E-23	1.32E-24	2.93E-26
TcO(OH) <sup>+</sup>	(mol/L)	4.42E-20	1.64E-18	2.04E-19	1.58E-20
TcO(OH) <sub>2</sub>	(mol/L)	2.46E-15	1.50E-14	5.44E-15	1.51E-15
TcO(OH) <sub>3</sub>	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10
TcOCO3	(mol/L)	6.95E-17	1.76E-14	1.54E-16	5.40E-18
TcOCO <sub>3</sub> (OH) <sub>3</sub>	(mol/L)	1.18E-17	4.91E-16	1.18E-17	1.50E-18
∑ competitive Tc(IV) complexes					
(Tc(IV)-cmp)	(mol/L)	8.13E-17	1.81E-14	1.65E-16	6.90E-18
Tc(IV) tot-(Tc(IV)-cmp)	(mol/L)	1.000E-10	9.998E-11	1.000E-10	1.000E-10
(Tc(IV) tot- (Tc(IV)-cmp)) / Tc(IV) tot		0.9999992	0.9998186	0.9999983	0.9999999
TcO <sup>+2</sup> dissolved	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10
TcO <sup>+2</sup>		1.45E-25	3.29E-23	1.32E-24	2.93E-26
TcO(OH)⁺	(mol/L)	4.42E-20	1.64E-18	2.04E-19	1.58E-20
TcO(OH) <sub>2</sub>	(mol/L)	2.46E-15	1.50E-14	5.44E-15	1.51E-15
TcO(OH) <sub>3</sub>	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10
TcOCO3	(mol/L)	6.95E-17	1.76E-14	1.54E-16	5.40E-18
TcOCO <sub>3</sub> (OH) <sub>3</sub>	(mol/L)	1.18E-17	4.91E-16	1.18E-17	1.50E-18
$\Sigma$ non competitive TcO <sup>+2</sup> complexes	(mol/L)	1.00E-10	1.00E-10	1.00E-10	1.00E-10
Σ Tc(IV)-cmp	(mol/L)	-3.51E-16	3.83E-18	2.06E-15	5.40E-15
Tc(IV) tot-(Tc(IV)-cmp)	(mol/L)	1.000E-10	1.000E-10	1.000E-10	9.999E-11
(Tc(IV) tot- (Tc(IV)-cmp)) / Tc(IV) tot without Tc(IV)-CO <sub>3</sub>		1.000	1.000	1.000	1.000

Svstem			Application		Data Source
Substrate			MX-80		MX-80
Solution		RPW	RPWC	HSPW	SBPW
Th <sup>+4</sup> added	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-10
Hq		7.377	6.593	7.046	7.6
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5
Th <sup>+4</sup> dissolved	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-10
Th <sup>+4</sup>	(mol/L)	3.59E-19	7.26E-17	1.05E-17	1.57E-21
Th(OH) <sub>4</sub>	(mol/L)	1.82E-11	2.66E-12	4.01E-11	8.69E-13
ThCI <sup>+3</sup>	(mol/L)	6.42E-20	1.30E-17	9.79E-18	1.29E-21
ThCI <sub>2</sub> <sup>+2</sup>	(mol/L)	8.38E-22	1.69E-19	6.26E-19	7.36E-23
ThCI3 <sup>+</sup>	(mol/L)	2.83E-22	5.70E-20	9.78E-19	1.05E-22
ThCl4	(mol/L)	1.01E-23	2.03E-21	1.52E-19	1.50E-23
Th(OH) <sub>3</sub> CO <sub>3</sub>	(mol/L)	9.98E-09	9.99E-09	9.96E-09	9.91E-11
Th(CO <sub>3</sub> ) <sub>5</sub> <sup>-6</sup>	(mol/L)	2.06E-19	7.51E-17	1.63E-21	3.30E-24
ThCO <sub>3</sub>	(mol/L)	9.59E-16	2.17E-13	8.67E-15	1.92E-18
Th(SO₄) <sub>3</sub> - <sup>2</sup>	(mol/L)	1.29E-14	3.69E-12	5.49E-15	3.24E-18
$\Sigma$ competitive Th complexes (Th-cmp) with Th-CO <sub>3</sub>	(mol/L)	9.98E-09	1.00E-08	9.96E-09	9.91E-11
Th tot-(Th-cmp)	(mol/L)	1.813E-11	2.657E-12	3.973E-11	8.687E-13
(Th tot- (Th-cmp)) / Th tot		0.0018129	0.0002657	0.0039727	0.0086871
ThCI <sup>+3</sup>	(mol/L)	6.42E-20	1.30E-17	9.79E-18	1.29E-21
ThCI <sub>2</sub> <sup>+2</sup>	(mol/L)	8.38E-22	1.69E-19	6.26E-19	7.36E-23
ThCI <sub>3</sub> <sup>+</sup>	(mol/L)	2.83E-22	5.70E-20	9.78E-19	1.05E-22
ThCl <sub>4</sub>	(mol/L)	1.01E-23	2.03E-21	1.52E-19	1.50E-23
Th(OH) <sub>3</sub> CO <sub>3</sub>	(mol/L)	9.98E-09	9.99E-09	9.96E-09	9.91E-11
Th(CO <sub>3</sub> ) <sub>5</sub> <sup>-6</sup>	(mol/L)	2.06E-19	7.51E-17	1.63E-21	3.30E-24
ThCO <sub>3</sub>	(mol/L)	9.59E-16	2.17E-13	8.67E-15	1.92E-18
Th(SO <sub>4</sub> ) <sub>3</sub> <sup>-2</sup>	(mol/L)	1.29E-14	3.69E-12	5.49E-15	3.24E-18
∑ competitive Th complexes (Th-cmb) without Th-CO <sub>3</sub>	(mol/L)	1.29E-14	3.69E-12	5.50E-15	3.24E-18
Th tot-(Th-cmp) without Th-CO4	(mol/L)	9.9999E-09	9.9962E-09	1.0000E-08	1.0000E-10
(Th tot- (Th-cmp)) / Th tot		0.9999987	0.9996314	0.9999994	76666666.0

Th(IV) speciation (TDB Yui et al., 1999)

Svstem			Application			Data Source	
Substrate			MX-80		Na-SM	y1 montmorill	onite
Solution		RPW	RPWC	HSPW		0.1 NaCIO4	
habbe <sup>4+</sup> dT	(  / 0m)	1 00E-08	1 00F-08	1 00E-08	1 00E-10	1 00E-10	1 00F-10
		00		00 100-			
Hq		7.377	6.593	7.046	7.377	6.593	7.046
pCO2		-2.6	-0.98	-2.6	ı	I	ı
Th <sup>+4</sup> dissolved	(mol/L)	1.00E-08	1.00E-08	1.00E-08	1.00E-10	1.00E-10	1.00E-10
Th <sup>+4</sup>	(mol/L)	3.59E-19	7.26E-17	1.05E-17	7.49E-19	1.02E-15	1.58E-17
Th(OH)4	(mol/L)	1.82E-11	2.66E-12	4.01E-11			
ThCl <sup>+3</sup>	(mol/L)	6.42E-20	1.30E-17	9.79E-18			
ThCI <sub>2</sub> <sup>+2</sup>	(mol/L)	8.38E-22	1.69E-19	6.26E-19			
ThCl <sub>3</sub> <sup>+</sup>	(mol/L)	2.83E-22	5.70E-20	9.78E-19			
ThCl4	(mol/L)	1.01E-23	2.03E-21	1.52E-19			
Th(OH) <sub>3</sub> CO <sub>3</sub>	(mol/L)	9.98E-09	9.99E-09	9.96E-09	4.02E-30	3.01E-30	3.75E-30
Th(CO <sub>3</sub> )5 <sup>-6</sup>	(mol/L)	2.06E-19	7.51E-17	1.63E-21			
ThCO <sub>3</sub>	(mol/L)	9.59E-16	2.17E-13	8.67E-15	3.23E-37	5.44E-35	2.96E-36
Th(SO <sub>4</sub> ) <sub>3</sub> <sup>-2</sup>	(mol/L)	1.29E-14	3.69E-12	5.49E-15			
$\Sigma$ competitive Th complexes							
(Th-cmp)	(mol/L)	1.29E-14	3.69E-12	5.50E-15	0.00E+00	0.00E+00	0.00E+00
Th tot-(Th-cmp)	(mol/L)	1.000E-08	9.996E-09	1.000E-08	1.000E-10	1.000E-10	1.000E-10
(Th tot- (Th-cmp)) / Th tot		0.9999987	0.9996314	0.9999994	1.0000	1.0000	1.0000

1999)
et al.,
B Yui
n (TD
speciatio
Th(IV)

System			Application			Application	
Substrate		MX-80	MX-80	MX-80	MX-80	MX-80	MX-80
Solution		RPW	RPWC	HSPW	RPW	RPWC	HSPW
U(IV) added	(mol/L)	1.00E-07	1.00E-07	1.00E-07	1.00E-07	1.00E-07	1.00E-07
Hq		7.377	6.593	7.046	7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-2.6	-0.98	-2.6
precipitated solids		$UO_2$		$UO_2$	$UO_2$	$UO_2$	UO2
U(IV) total dissolved	(mol/L)	3.39E-10	3.39E-10	3.39E-10	3.39E-10	3.39E-10	3.39E-10
n(IV)	(mol/L)	4.53E-33	4.58E-30	6.02E-32	4.53E-33	4.58E-30	6.02E-32
U(OH) <sup>+3</sup>	(mol/L)	3.82E-27	6.83E-25	2.91E-26	3.82E-27	6.83E-25	2.91E-26
U(OH)	(mol/L)	3.39E-10	3.39E-10	3.39E-10	3.39E-10	3.39E-10	3.39E-10
U(OH)5 <sup>-</sup>	(mol/L)	1.07E-14	1.91E-15	4.84E-15	1.07E-14	1.91E-15	4.84E-15
U(CO3)4 <sup>4</sup>	(mol/L)	1.86E-23	1.39E-18	1.86E-23	1.86E-23	1.39E-18	1.86E-23
U(CO3)5 <sup>-6</sup>	(mol/L)	7.39E-29	5.14E-23	7.40E-29	7.39E-29	5.14E-23	7.40E-29
USO4 <sup>+2</sup>	(mol/L)	1.49E-29	5.65E-27	1.49E-29	1.49E-29	5.65E-27	1.49E-29
U(SO4)2	(mol/L)	7.41E-29	1.29E-25	7.41E-29	7.41E-29	1.29E-25	7.41E-29
UCI <sup>+3</sup>	(mol/L)	2.33E-31	4.66E-30	2.33E-31	2.33E-31	4.66E-30	2.33E-31
∑ competitive U(IV) complexes							
(U(IV) cmp)	(mol/L)	1.86E-23	1.39E-18	1.86E-23	8.93E-29	1.35E-25	8.93E-29
U(IV) tot - U(IV) cmp		3.39E-10	3.39E-10	3.39E-10	3.39E-10	3.39E-10	3.39E-10
U(IV) tot - U(IV) cmp / U(IV) tot		1.000	1.000	1.000	1.000	1.000	1.000

U(IV) speciation (TDB Grenthe et al., 1992)

92)	Application	U8"XW
t al., 19		
U(VI) speciation (TDB Grenthe e	System	Cubetrato

Svstem			Application			Data source	
Substrate			MX-80		Na-mo	intmorillonite \$	SAz-1
Solution		RPW	RPWC	MSPW		0.1 M NaNO3	
UO <sub>2</sub> <sup>+2</sup> added	(mol/L)	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09
Hq		7.377	6.593	7.046	7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6	-3.5	-3.5	-3.5
UO2 <sup>+2</sup> dissolved	(mol/L)	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09
UO2 <sup>+2</sup>	(mol/L)	2.73E-14	2.07E-14	9.07E-13	5.85E-13	2.50E-11	3.06E-12
UO2CO3	(mol/L)	3.44E-11	2.92E-11	2.78E-10	1.19E-10	1.37E-10	1.36E-10
UO <sub>2</sub> (OH) <sub>3</sub> <sup>-2</sup>	(mol/L)	9.21E-12	3.08E-14	3.37E-11	2.36E-10	4.48E-11	1.25E-10
UO <sub>2</sub> (OH) <sub>4</sub> *	(mol/L)	8.63E-18	7.74E-21	1.35E-17	1.84E-16	5.75E-18	4.57E-17
(UO <sub>2</sub> ) <sub>3</sub> (OH) <sub>7</sub>	(mol/L)	2.98E-22	4.15E-28	7.14E-20	5.68E-18	1.43E-18	3.92E-18
UO <sub>2</sub> (CO <sub>3</sub> ) <sup>-2</sup>	(mol/L)	2.74E-09	2.62E-09	4.28E-09	9.45E-10	2.95E-11	2.34E-10
(UO <sub>2</sub> )(CO <sub>3</sub> ) <sup>34</sup>	(mol/L)	4.49E-09	4.84E-09	1.07E-09	9.47E-11	7.98E-14	5.12E-12
(UO <sub>2</sub> ) <sub>2</sub> (CO <sub>3</sub> )(OH) <sub>3</sub> <sup>-3</sup>	(mol/L)	1.62E-13	4.62E-16	4.80E-12	1.44E-11	3.15E-12	8.73E-12
UO <sub>2</sub> NO <sub>3</sub> ⁺	(mol/L)				4.33E-14	1.85E-12	2.27E-13
(UO <sub>2</sub> ) <sub>11</sub> (CO <sub>3</sub> ) <sub>6</sub> (OH) <sub>12</sub> <sup>-2</sup>	(mol/L)	1.00E-56	1.00E-77	4.97E-57	5.30E-56	7.00E-57	4.90E-56
$\Sigma$ competitive U(VI) complexes							
(U(VI) cmp)	(mol/L)	7.23E-09	7.47E-09	5.36E-09	1.05E-09	3.46E-11	2.48E-10
U(VI) tot - U(VI) cmp	(mol/L)	2.67E-10	3.41E-11	2.14E-09	6.45E-09	7.47E-09	7.25E-09
U(VI) tot - U(VI) cmp / U(VI) tot		0.035599	0.004550	0.284893	0.859482	0.995393	0.966866
UO, <sup>+2</sup> dissolved	(mol/L)	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09
- U0,*2	(mol/L)	2.73E-14	2.07E-14	9.07E-13	5.85E-13	2.50E-11	3.06E-12
UO2CO3	(mol/L)	3.44E-11	2.92E-11	2.78E-10	1.19E-10	1.37E-10	1.36E-10
UO <sub>2</sub> (OH) <sub>3</sub> <sup>-2</sup>	(mol/L)	9.21E-12	3.08E-14	3.37E-11	2.36E-10	4.48E-11	1.25E-10
UO <sub>2</sub> (OH) <sub>4</sub> *	(mol/L)	8.63E-18	7.74E-21	1.35E-17	1.84E-16	5.75E-18	4.57E-17
(UO <sub>2</sub> ) <sub>3</sub> (OH) <sub>7</sub>	(mol/L)	2.98E-22	4.15E-28	7.14E-20	5.68E-18	1.43E-18	3.92E-18
UO <sub>2</sub> (CO <sub>3</sub> ) <sup>2-2</sup>	(mol/L)	2.74E-09	2.62E-09	4.28E-09	9.45E-10	2.95E-11	2.34E-10
(UO <sub>2</sub> )(CO <sub>3</sub> ) <sup>3-4</sup>	(mol/L)	4.49E-09	4.84E-09	1.07E-09	9.47E-11	7.98E-14	5.12E-12
(UO <sub>2</sub> ) <sub>2</sub> (CO <sub>3</sub> )(OH) <sub>3</sub> <sup>-3</sup>	(mol/L)	1.62E-13	4.62E-16	4.80E-12	1.44E-11	3.15E-12	8.73E-12
	(mol/L)				4.33E-14	1.85E-12	2.27E-13
(UO <sub>2</sub> ),1(CO <sub>3</sub> ) <sub>6</sub> (OH) <sub>12</sub> <sup>-2</sup>	(mol/L)	1.00E-56	1.00E-77	4.97E-57	5.30E-56	7.00E-57	4.90E-56
∑ competitive U(VI) complexes (U(VI) cmp)	(mol/L)	0.00E+00	0.00E+00	0.00E+00	4.33E-14	1.85E-12	2.27E-13
U(VI) tot - U(VI) cmp	(mol/L)	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09	7.50E-09
U(VI) tot - U(VI) cmp / U(VI) tot		1.00000	1.000000	1.00000	0.999994	0.999754	0.999970

System			Application	
Substrate			MX-80	
Solution		RPW	RPWC	HSPW
Zr added	(moi/l)	1.UUE-US	80-300.T	1.00E-US
Hd		7.377	6.593	7.046
pCO <sub>2</sub>		-2.6	-0.98	-2.6
solids		ZrO <sub>2</sub>	ZrO <sub>2</sub>	$ZrO_2$
Zr <sup>+4</sup> dissolved	(  / om)	4 31E-11	9 18F-12	2 09E-11
Zr <sup>+4</sup>	(mol/L)	4.96E-30	6.86E-27	6.61E-29
Zr(OH) <sup>+3</sup>	(mol/L)	2.83E-23	6.42E-21	2.15E-22
Zr(OH) <sub>4</sub>	(mol/L)	2.51E-12	2.51E-12	2.51E-12
Zr(OH)5 <sup>-</sup>	(mol/L)	4.06E-11	6.67E-12	1.84E-11
ZrCl <sup>+3</sup>	(mol/L)	2.23E-30	3.08E-27	1.54E-28
Zr(SO <sub>4</sub> ) <sup>+2</sup>	(mol/L)	1.05E-26	1.63E-23	4.31E-26
∑ competitive Zr complexes				
(Zr-cmp)	(mol/L)	4.06E-11	6.67E-12	1.84E-11
Zr tot-(Zr-cmp)	(mol/L)	2.512E-12	2.512E-12	2.512E-12
(Zr tot- (Zr-cmp)) / Zr tot		0.0582313	0.2735142	0.1201876

Zr(IV) speciation (TDB Hummel et al., 2002)

ISSN 1404-0344 CM Digitaltryck AB, Bromma, 2004